



Ecole Doctorale des Sciences Economiques, Juridiques, Politiques et de Gestion

LEO-UCA, Université Clermont Auvergne, Clermont-Ferrand, France

# PAYMENTS FOR ENVIRONMENTAL SERVICES UNDER IMPERFECT COMPETITION

Thèse Nouveau Régime

Présentée et soutenue publiquement le **vendredi 15 décembre 2023** Pour l'obtention du titre de Docteur *ès* Sciences Économiques de l'Université Clermont Auvergne

Par

# Anneliese KRAUTKRAEMER

sous la direction de:

# Pr. Pascale Motel Combes et Pr. Sonia Schwartz

# Composition du jury:

Dorothée BRECARD	Professeur, Université de Toulon	Rapporteure
Marie-Hélène HUBERT	Professeur, Université de Rennes 1	Rapporteure, Présidente
Dominique BUREAU	Professeur, Ecole Polytechnique,	Suffragant
	Président de la Commission de l'Economie	
	du Développement Durable	
	(Ministère de la Transition Ecologique)	
Pascale MOTEL-COMBES	Professeur, Université Clermont Auvergne	Directrice
Sonia SCHWARTZ	Professeur, Université Clermont Auvergne	Directrice

L'Université Clermont Auvergne n'entend donner aucune approbation ni improbation aux opinions émises dans cette thèse. Ces opinions doivent être considérées comme propres à leur auteur. Cette thèse a été financée par le CPER Auvergne-Rhône-Alpes ATTRIHUM

In loving memory of my dad, Jeff Krautkraemer

# Remerciements

Mes remerciements s'adressent tout d'abord à mes directrices de thèse, les Professeurs Pascale Motel Combes et Sonia Schwartz, pour leurs conseils avisés, leurs encouragements et surtout leur confiance dans la réalisation de cette thèse. Je tiens également à remercier les membres de mon jury de thèse, les Professeurs Marie-Hélène Hubert, Dorothée Brécard et Dominique Bureau, pour avoir pris le temps d'évaluer ce travail. Je me réjouis de pouvoir bénéficier de votre expertise et de partager mes réflexions avec vous lors de ma soutenance. Je remercie également la région Auvergne-Rhône-Alpes qui a financé la thèse.

Je tiens également à remercier tous mes amis et collègues pour le soutien qu'ils m'ont apporté tout au long de ces années. Ensuite, j'exprime ma gratitude à tous mes anciens professeurs qui m'ont encouragée, avec une mention spéciale pour mes professeurs de français au lycée et à l'université, sans lesquels je n'aurais jamais eu la confiance nécessaire pour venir vivre en France. Enfin, je ne serais pas là sans ma famille. Merci pour votre confiance, vos conseils, votre soutien et vos encouragements.

Je vous souhaite une très bonne lecture.

## Résumé

La thèse étudie la mise en œuvre optimale des paiements pour services environnementaux (PSE) dans un cadre de concurrence imparfaite. Son objectif est de tirer des recommandations en termes d'économie publique pour le régulateur, afin de protéger la biodiversité. Les résultats de la thèse permettent d'identifier des scénarios où les PSE sont inefficaces. La thèse se compose de quatre chapitres. Le chapitre 2 est une revue de la littérature sur les PSE. Les chapitres 3, 4 et 5 examinent chacun un aspect particulier de la concurrence imparfaite.

Le chapitre 3 définit le PSE et la taxe pigouvienne de second rang. Nous supposons un modèle où les agriculteurs répartissent leur terre entre une agriculture conventionnelle, biologique et des bandes enherbées. Le régulateur fixe une taxe environnementale sur l'agriculture conventionnelle, génératrice de dommages environnementaux et un PSE sur les bandes enherbées favorisant la biodiversité. Le secteur conventionnel est concurrentiel et le secteur de l'agriculture biologique oligopolistique. Il s'avère que la taxe environnementale de second rang est supérieure au dommage marginal, tandis que le PSE de second rang est inférieur au bénéfice marginal. Le coût marginal social des fonds publics (MCF) est ensuite introduit. Si la taxe environnementale augmente avec le MCF, le PSE diminue avec le MCF si la demande pour le bien conventionnel est inélastique. Ces résultats mettent en évidence une composante contributive de la taxe incitative. Ce chapitre expose des cas où le PSE est inefficace.

Le chapitre 4 étudie l'efficacité des PSE basés sur l'additionnalité. Ces PSE permettent de prendre en compte des préoccupations d'ordre budgétaire, en ne basant le paiement que sur les services environnementaux supplémentaires obtenus. Nous supposons un modèle à deux périodes où un agriculteur répartit ses terres entre une production biologique, une production conventionnelle causant des dommages à l'environnement et des bandes enherbées génératrices de services environnementaux. Le PSE est introduit dans la dernière période. Ce PSE additionnel génère des distorsions dans le comportement de l'agriculteur en première période afin d'obtenir un paiement plus important en période finale. Le PSE de second rang prenant en compte cette distorsion est égal à la différence actualisée des bénéfices environnementaux marginaux obtenus à chaque période. En présence de PSE additionnels, les taxes environnementales de second rang ne sont plus égales au dommage marginal et s'ajustent pour tenir compte des distorsions causées par le PSE. Un pouvoir de marché sur le marché biologique est ensuite introduit. Il réduit la distorsion due à l'additionnalité dans la période initiale mais aussi la quantité de production biologique dans la période finale. Le PSE additionnel de second rang dépend de l'ampleur de ces deux effets, ce qui modifie également la valeur des taxes environnementales. Le chapitre 4 montre qu'un PSE additionnel ne permet jamais d'atteindre l'efficacité environnementale, même lorsque les marchés fonctionnent de façon concurrentielle.

Le chapitre 5 analyse le principe de « pas de perte nette » d'une politique de protection de la biodiversité, couplé à la séquence Eviter Réduire Compenser (ERC). Il étudie le comportement d'un aménageur confronté à cette politique. L'analyse souligne la difficulté à transposer dans l'analyse économique les concepts de la séquence ERC. Supposant un cadre d'information parfaite, il est montré que la demande de compensation ne dépend pas de son prix, contrairement à l'offre de compensation. En information asymétrique, le développeur utilise son information de façon stratégique. Il définit simultanément sa demande de compensation et le niveau de réduction des dommages en fonction du prix de la compensation. Au final, le choix du projet se base aussi sur ce prix. Le chapitre 5 montre que la séquence ERC est inefficace à protéger la biodiversité lorsque l'information est asymétrique.

**Mots-clés** : Protection de la biodiversité · Paiements pour services environnementaux · Coût social des fonds publiques · Pouvoir de marché · Additionnalité · Hiérarchie de compensation

Codes JEL :  $Q57 \cdot Q58$ 

### Abstract

The thesis studies the optimal implementation of payments for environmental services (PES) under imperfect competition. It draws lessons from public economics for regulators on how to implement PES to protect biodiversity. The results show cases where PES is ineffective at promoting biodiversity. The thesis consists of four chapters. Chapter 2 reviews the literature on PES, while Chapters 3, 4 and 5 each examine an aspect of imperfect competition.

Chapter 3 designs the second-best PES and environmental tax under imperfect competition. Farmers allocate their land between conventional agriculture, organic agriculture, and buffer strips. The regulator sets an environmental tax on conventional agriculture as it causes environmental damages, and a PES on buffer strips as they favor biodiversity. The conventional sector is perfectly competitive, while the organic sector is organized under an oligopoly. We show that the second-best environmental tax is higher than the marginal damage while the PES is lower than the marginal benefit. We then include the social cost of public funds (MCF). The environmental tax increases and the PES decreases with the MCF as long as demand for the conventional good is inelastic. We thus highlight a contributory component of the incentive tax. Chapter 3 also identifies specific scenario where the PES is ineffective.

Chapter 4 studies the efficiency of additionality-based PES. They can address budget constraint concerns by only paying for additional environmental services. We look at a farmer who allocates his land between organic production, conventional production causing environmental damage, or grass strips generating biodiversity. Using a two-period model, we introduce a PES in the final period, rewarding the additional grass strips provided by the farmer. We show that the additionality-based PES distorts the behavior in the initial period in order to get more payment in the final period. The second-best PES to limit this behavior equals the discounted difference of the marginal environmental benefits obtained in each period. We define the second-best value of environmental taxes in the presence of the additionality-based PES. They no longer equal the marginal damage and adjust to account for the distortions caused by PES. We then introduce market power in the organic market. Market power reduces the distortion due to the PES in the initial period but reduces the organic production quantity in the final period. The second-best PES depends on the size of both effects and environmental taxes under market power have to be amended. In any case, Chapter 4 shows that an additionality-based PES never achieves environmental efficiency, even in a competitive market framework.

Chapter 5 focuses on the "no net loss principle" of a biodiversity protection policy, accompanied by the Avoid Reduce Compensate (ARC) sequence and studies the behavior of a developer facing this policy. The analysis highlights the difficulty of transposing the concepts of the ARC sequence into economic analysis. Assuming perfect information, we show that the demand for compensation does not depend on its price, unlike the supply of compensation. Under asymmetric information, the developer behaves strategically. He simultaneously defines his demand for offsets and the level of damage reduction based on the offset price. In the end, the project choice is made based on the price of the offset. Chapter 5 shows that the mitigation hierarchy is ineffective under asymmetric information.

**Keywords** : Biodiversity Conservation · Payments for Environmental Services · Marginal Social Cost of Public Funds · Market Power · Additionality · Mitigation Hierarchy

**JEL Codes** :  $Q57 \cdot Q58$ 

# Contents

1	$\mathbf{Intr}$	oducti	on	1	
	1.1	From biodiversity to environmental services			
	1.2	Payments for environmental services			
	1.3 Motivation $\ldots$				
	1.4	Contri	butions	6	
<b>2</b>	Pay	ments	for Environmental Services: A literature review	8	
	2.1	Defini	ng payments for environmental services	9	
		2.1.1	From biodiversity to environmental services	9	
		2.1.2	Payments for environmental services	10	
	2.2	Are Pl	ES programs efficient?	12	
		2.2.1	Additionality	13	
		2.2.2	Using auctions to reveal conservation $\operatorname{cost}$ information $\ .$ .	16	
		2.2.3	Other considerations	20	
	2.3	Agglor	meration Bonus	24	
		2.3.1	Evaluating the Agglomeration Bonus	25	
		2.3.2	Agglomeration bonus design	29	
		2.3.3	Existing programs with agglomeration bonuses	31	
	2.4	Mitiga	tion banking: PES using tradable permits	33	
		2.4.1	Compensation feasibility	35	
		2.4.2	A lack of avoidance	37	
	2.5	Conclu	1sion	38	
3	Pay	ment f	or Environmental Services and environmental tax under		
	$\operatorname{imp}$	erfect	competition	<b>40</b>	
	3.1	Introd	uction	41	
	3.2	The m	odel	46	
		3.2.1	Assumptions	46	
		3.2.2	The benchmark	47	
		3.2.3	The first-best	49	
	3.3	Second	d-best environmental policies	50	

		3.3.1 The farmer's behavior	50				
		3.3.2 Second-best level of environmental tax and PES	51				
	3.4	The marginal social cost of public funds	54				
	3.5	Conclusion	58				
	3.6	Appendices of Chapter 3	60				
4	Are	e Additionality-Based Payments for Environmental Services					
	Effi	ficient?					
	4.1	Introduction	69				
	4.2	Literature review	72				
	4.3	The model	75				
		4.3.1 Assumptions	75				
		4.3.2 The benchmark: No regulation	76				
		4.3.3 First-best regulation	76				
	4.4	Additionality and perfect competition	78				
		4.4.1 Strategic behaviors	78				
		4.4.2 Tax and PES designs	81				
		4.4.3 Calculated quantities	84				
	4.5	Additionality and imperfect competition	84				
		4.5.1 Monopoly: Strategic behaviors	85				
		4.5.2 Tax and PES designs	87				
		4.5.3 Calculated quantities	89				
	4.6	Conclusion	89				
	4.7	Appendices of Chapter 4	91				
5	On	the efficiency of the mitigation hierarchy	97				
	5.1	Introduction	98				
	5.2	Assumptions	102				
	5.3	Perfect and symmetric information	103				
	5.4	Asymmetric Information	104				
	5.5	Discussion and conclusion	106				
6	Cor	nclusion 1	L <b>09</b>				
	6.1	Main results	109				
	6.2	Limitations and future research	112				
7	Rés	sumé extensif en français: Paiements pour services environ-					
	nen	nentaux et concurrence imparfaite	15				

References		
7.5	Limites et pistes de recherche futures	125
7.4	Principaux résultats	121
7.3	Motivation de la thèse	119
7.2	Paiements pour services environnementaux	117
7.1	De la biodiversité aux services environnementaux	115

# | Chapter

# Introduction

# 1.1 From biodiversity to environmental services

The Convention on Biological Diversity (CBD) defines biodiversity as "the variability among living organisms from all sources including, *inter alia*, terrestrial, marine and other aquatic ecosystems and the ecological complexes of which they are part; this includes diversity within species, between species and of ecosystems" (Convention on Biological Diversity, 2010). A diversity of life and ecosystems also allows for a diversity of environmental services. Dasgupta (2021) has made the comparison to financial portfolios, where a diversity of investments helps mitigate risks; biodiversity can similarly help to mitigate risks to help nature be more productive, resilient, and adaptable.

The decline in biodiversity is a well-documented phenomenon, which is likely to worsen with climate change (Díaz et al., 2019; Ruckelshaus et al., 2020; Dasgupta, 2021). One of the main causes of biodiversity decline is the loss of various habitats due to land use change (Lewis et al., 2011; Bamière et al., 2013). A large part of the terrestrial biodiversity loss has occurred on agricultural lands, which were once home to an abundance and variety of species, but land intensification and increasing farm size has transformed and fragmented natural habitats, leading to a decline in many species. According to Dasgupta (2021), an estimated 20 percent of species may become extinct in the next few decades, and perhaps double that by the end of the century.

In 1977 the term 'nature's services' first appeared in the academic literature in a paper by Walter Westman (Westman, 1977). This was followed by the term "ecosystem services", which first appeared in the literature in 1981 (Costanza et al., 2017). However, it was not until the late 1990s, when an article in Nature estimated that the total value of all ecosystem services in the biosphere was between 16-54 trillion USD, that ecosystem services gained substantial popularity as a topic in academic literature (Costanza et al., 1997).

The Millennium Ecosystem Assessment explains that "Ecosystem services are the benefits people obtain from ecosystems. These include *provisioning services* such as food, water, timber, and fiber; regulating services that affect climate, floods, disease, wastes, and water quality; *cultural services* that provide recreational, aesthetic, and spiritual benefits; and *supporting services* such as soil formation, photosynthesis, and nutrient cycling" (Reid et al., 2005). The economic literature distinguishes between ecosystem services and environmental services (ES). The term "ecosystem services" refers to the benefits provided by ecosystems while "environmental services" refers to the protection of these ecosystems by humans and to the notion of externalities induced by human activities. For example, the Food and Agriculture Organization of the United Nations (FAO) proposes a definition of ES in terms of ecosystem services. For agriculture, ES are defined as the subpart of ecosystem services that can be qualified in terms of externalities, that is, all ecosystem services except provisioning services (Lugo, 2007). ES can be used to refer to the production of services by farmers to protect the environment. We can cite several examples. Long crop rotations improve ecosystem services such as supporting services through improved soil quality. The diversity of productive activities on a farm promotes beneficial interactions between crops and livestock, and the management of landscape features such as grass buffer strips, slopes, hedgerows, or watercourses contribute to the ecological functioning of agroecosystems. All of these definitions make it possible to justify the remuneration for these ES as an internalization of externalities. This leaves room for policy intervention to encourage their optimal provision, such as payments for environmental services (PES), which have become a familiar tool for conserving and restoring ecosystems and the services they provide (Dasgupta, 2021).

## **1.2** Payments for environmental services

One of the most widely cited definitions of PES comes from Wunder (2005), which defines PES as a voluntary transaction in which a well-defined ES or land use that can produce that service is purchased by (at least) one ES buyer from (at least) one ES provider if and only if the ES provider guarantees the provision of ES (conditionality). Conditionality can be difficult to assess in outcome-based PES schemes, as some ES are difficult to measure. In practice, it is much more common to see action-based PES schemes that are conditional on specific land use or management practices.

The above definition illustrates the Coase theorem, whereby an externality can be resolved by private negotiation and the optimal allocation of ES can be achieved, regardless of the initial property allocation and assuming sufficiently low transaction costs and well-defined property rights (Coase, 1960). An example of a Coasean PES is the Vittel PES in northeastern France, where Nestlé made an agreement with local farmers to compensate them for reducing their use of fertilizers in order to prevent nitrate pollution in the aquifers (Bingham, 2021).

The definition of PES can be expanded to include certain types of government interventions that reflect a Pigouvian subsidy (Pigou, 1920; Sattler and Matzdorf, 2013). This type of PES is much more common in practice than a Coasean PES. For example, European agri-environmental schemes (AES) are publicly funded, and the government acts as an intermediary between the buyers of ES (the public) and the sellers of ES (the farmers who receive PES grants). Both Coasean and Pigouvian PES schemes follow the beneficiary-pays principle.

Yet another alternative definition is provided by Muradian et al. (2010), describing PES as a transfer of resources between social actors, which aims to create incentives to align individual and/or collective land use decisions with social interest in natural resource management. This definition is more flexible than that of Wunder (2005), and better reflects what happens in actual PES schemes than what should happen in theory. This definition also reflects the fact that payments are not necessarily monetary, but can also be in-kind transfers.

Various forms of PES programs have been implemented in both developing and developed countries. In the United States, the Conservation Reserve Program has been in place since 1985 (Hellerstein, 2017). Costa Rica is one of the first countries to adopt a national PES program, establishing theirs in 1997 (Pagiola, 2008). In China, the Sloping Land Conversion Program and Natural Forest Conservation Program invested over 50 billion USD from 2000-2009 (Salzman et al., 2018). One program that links developed and developing countries is REDD+, conceptualized in 2007 which is aimed at reducing carbon emissions from deforestation and forest degradation by incentivizing developed countries to keep their forests in tact by offering payments financed by developed countries for actions taken to reduce or stop deforestation (Chiroleu-Assouline et al., 2018). Biodiversity protection policies often focus on agricultural lands, such as the AES in Europe, which member states have been required to apply since 1992. AES are a type of PES that remunerate farmers

for voluntary actions taken to preserve and improve the environment. In fact, in the European Union, the largest source for practical nature conservation comes from the AES applied under the Common Agricultural Policy (CAP) (Herzon et al., 2018). Practices adopted under the AES include the reduction of fertilizers and/or pesticides, the establishment of grass buffer strips near rivers, and adaptations to crop rotations. More recently, according to the 2018 National Biodiversity Plan in France, water agencies are testing their own PES programs. They have been allocated 150 million euros from France's national budget, with the objective of maintaining or creating good ecological practices, such as reducing pesticides, planting cover crops, etc. This new program also aims at results-based instead of action-based payments, with annual monitoring to determine the results. Either maintaining or creating good practices will be rewarded, but creating good practices will receive higher payments (up to 676 euros/ha/year, compared to up to 66 euros/ha/year for the maintenance of good practices).

Classifying the wide variety of PES schemes is not an easy task. As Sattler et al. (2013) point out, PES schemes rely on a multitude of approaches that differ greatly in terms of the ES addressed, price formation mechanisms, origins and levels of payment, characteristics of buyers and sellers, rules governing the contract between the parties involved, level of complexity, etc. According to Wunder (2005), the main ES addressed by PES are carbon sequestration and storage, biodiversity protection, watershed protection and landscape beauty.

### 1.3 Motivation

Despite the vast literature on PES, there are still research questions to explore. First, there is a need to analyze how PES interact with other public policies, such as Pigouvian taxes, and to define their level in multi-policy contexts. There are already a few studies that look at the interactions between public policies. According to Bryan and Crossman (2013), the interaction effects of multiple financial incentives can reduce the effectiveness of policies when multiple incentives encourage the provision of services from agro-ecosystems. Agri-environmental measures need to take into account that policies are typically packages of different policy tools arranged in policy mixes and that financial incentives for different ecosystem services interact (Huber et al., 2017). Finally, Lankoski and Ollikainen (2003) provide an interesting framework for a theoretical analysis to find the optimal level of a PES and Pigouvian tax in the agricultural sector. Another question not well developed in the PES literature is how market power can change the optimal level of PES. This question has already been addressed in the literature on Pigouvian taxes, which has found that under market power, the optimal second-best tax should be less than the marginal damage (Barnett, 1980; Ebert, 1991). However, no papers on the level of PES under imperfect competition have been found. Market power leads to a suboptimal production level, as firms restrict production to increase their profits. Since both taxes and PES influence production levels, they must take into account any market power in order to not further distort production away from the socially optimal level.

Moreover, when PES are publicly funded, it is necessary to raise funds through taxes, which can then distort markets. Increasing contributory taxes can alter the allocation of resources in an economy by influencing consumption, labor and investment decisions. Therefore, it seems important to take into account the social cost of public funds in our analysis. This is a measure of the welfare loss suffered by society as a result of mobilizing additional revenues to finance public expenditures (Browning, 1976; Dahlby, 2008). For example, Browning (1976) estimates the MCF of the U.S. labor income tax, and finds an MCF of 1.09–1.16 per dollar of tax revenue collected. According to Beaud (2008), this cost is equal to 1.2 for France. Thus, when the regulator collects one euro of tax, it costs society 1.2 euros. This aspect should therefore not be neglected in the decision to set up a PES.

Next, an important factor in the economic efficiency of PES programs is whether or not they are additional, that is, whether they entail the provision of an ES that would not have occurred in the absence of any payment. Early in the development of PES, the majority of programs had no additionality requirement, perhaps due to the idea that monitoring additionality would prove too costly. Or, as in the case of Costa Rica's national program, the goal may be to recognize and pay for any provision of ES regardless of its additionality. It is only more recently that the assessment of additionality of PES programs has become a concern, even though it is essential for a PES scheme to achieve its environmental objective with economic efficiency while maintaining investor confidence (Bennett, 2010).

Finally, mitigation banks are a type of PES, as both instruments involve paying for actions to restore, preserve, and/or manage biodiversity and ecosystems. Indeed, Salzman et al. (2018) refer to mitigation banks as a type of compliance PES. Rather than using a Pigouvian subsidy, the mitigation bank involves tradable permits, and is therefore a quantity rather than a price mechanism. Before credits can be purchased

from a mitigation bank, one must, in theory, follow what is called a mitigation hierarchy with the goal of no net loss (of biodiversity or environmental services). When designing a project that will have negative impacts on the surrounding environment, a developer must first avoid any possible damage by modifying the project. Next, any damage that cannot be avoided must be reduced. Finally, any residual damage after avoiding and reducing must be compensated, either by restoring another equivalent area of land or by purchasing credits from a mitigation bank. However, in reality, the mitigation hierarchy is not properly followed, as it is not rational to do so.

## 1.4 Contributions

The objective of this thesis is to analyze these questions, in order to deepen the knowledge of PES. This thesis consists of a review of the literature on PES and then three papers in which we address some of the remaining gaps in the literature on PES.

The first paper designs the second-best PES when it interacts with a Pigouvian tax under imperfect competition. We consider farmers who face a choice between producing a conventional or an organic agriculture good. The regulator sets a Pigouvian tax on conventional agriculture as it generates environmental damages, as well as a PES on uncultivated land as buffer strips favor biodiversity. The conventional agriculture sector is perfectly competitive, unlike the organic agriculture sector, which is organized under an oligopoly. We show that the second-best level of the Pigouvian tax is higher than the marginal damage whereas the PES is lower than the marginal benefit. We then introduce the social marginal cost of public funds (MCF) and show that the Pigouvian tax increases with the MCF while the PES decreases with the MCF provided that demand for the conventional agriculture good is inelastic. We thus highlight a contributory component of the environmental incentive tax. The first paper also identifies specific cases where the PES is ineffective in promoting biodiversity.

The implementation of PES may face a financing constraint, especially when the buyer is a public regulator. An additionality-based PES can address this problem by only paying for additional ES. The objective of the second paper is to study the efficiency of an additionality-based PES. To do so, we consider a farmer who has to choose to allocate his land between organic production, conventional production causing environmental damage, or biodiversity-generating grass strips. Using a twoperiod model, we introduce a PES in the final period, remunerating the additional grass strips provided by the farmer. We show that the additional PES distorts the behavior in the initial period, in order to obtain more payment in the final period. The second-best PES to limit this behavior is equal to the discounted difference of the marginal environmental benefits obtained in each period. We also establish the second-best value of environmental taxes in the presence of the additionality-based PES. They are no longer equal to the marginal damage and are adjusted to take into account the distortions caused by the additionality-based PES. The analysis is then extended by taking into account market power in the organic market. It turns out that market power reduces the distortion due to the additionality-based PES in the initial period but reduces the organic production quantity in the final period. The second-best PES depends on the size of these two effects and environmental taxes under market power have to be amended. In any case, the second paper shows that an additionality-based PES never achieves environmental efficiency, even in a competitive market framework. Furthermore, it provides new insights into understanding the interactions between different environmental policies in the presence of several types of distortions.

Finally, the third paper focuses on the no net loss principle of a biodiversity protection policy, accompanied by the "avoid, reduce, compensate" (ARC) sequence and studies the behavior of a developer facing this policy. The definition of each step in the sequence is highly conceptual. First, we show that it is difficult to transpose into a microeconomic decision model. Starting with a perfect information scenario, we show that the demand for compensation does not depend on its price, unlike the supply of compensation. We then assume that the regulator does not have the same information as the developer about the environmental damage of the project or the marginal costs of reducing the damage. In this case, the developer strategically uses this asymmetric information. Using backward induction reasoning, he simultaneously defines his demand for offsets and the level of environmental damage reduction based on the offset price. In the end, the project choice is made by also taking into account the price of the offset. The third paper shows that the mitigation hierarchy is ineffective under asymmetric information, making the safeguarding of biodiversity inefficient.

# $\sum_{\text{Chapter}} 2$

# Payments for Environmental Services: A literature review

This literature review begins by defining payments for environmental services in Section 2.1 followed by a discussion of the literature regarding the efficiency of PES programs in Section 2.2. Next, Section 2.3 reviews the literature on the agglomeration bonus, which aims to improve the environmental outcomes of PES schemes. Then, Section 2.4 reviews the literature on mitigation banking, which can be thought of as a PES program that is quantity-based instead of price-based. Finally, Section 2.5 concludes.

# 2.1 Defining payments for environmental services

We begin by defining biodiversity and discussing its importance to human well-being. Then, we discuss the definition of payments for environmental services, a policy that aims to conserve biodiversity.

#### 2.1.1 From biodiversity to environmental services

The Convention on Biological Diversity (CBD) defines biodiversity as "the variability among living organisms from all sources including, *inter alia*, terrestrial, marine and other aquatic ecosystems and the ecological complexes of which they are part; this includes diversity within species, between species and of ecosystems" (Convention on Biological Diversity, 2010). A diversity of life and ecosystems also allows for a diversity of environmental services. Dasgupta (2021) made the comparison to financial portfolios, where a diversity of investments helps mitigate risks; biodiversity can likewise help to mitigate risks to help nature be more productive, resilient, and adaptable.

The decline in biodiversity is a well-documented phenomenon, which is likely to worsen with climate change (Díaz et al., 2019; Ruckelshaus et al., 2020; Dasgupta, 2021). One of the main causes of biodiversity decline is the loss of various habitats due to land use change (Lewis et al., 2011; Bamière et al., 2013). A large part of the terrestrial biodiversity loss has occurred on agricultural lands, which were once home to an abundance and variety of species, but land intensification and increasing farm size has transformed and fragmented natural habitats, leading to a decline in many species. According to Dasgupta (2021), an estimated 20% of species may become extinct in the next few decades, and perhaps double that by the end of the century.

In 1977 the term 'nature's services' first appeared in the academic literature in a paper by Walter Westman (Westman, 1977). This was followed by the term "ecosystem services", which first appeared in the literature in 1981 (Costanza et al., 2017). However, it was not until the late 1990s, when an article in Nature estimated that the total value of all ecosystem services in the biosphere was between 16-54 trillion USD, that ecosystem services gained substantial popularity as a topic in academic literature (Costanza et al., 1997).

The Millennium Ecosystem Assessment explains that "Ecosystem services are the benefits people obtain from ecosystems. These include *provisioning services* such as food, water, timber, and fiber; *regulating services* that affect climate, floods, disease, wastes, and water quality; *cultural services* that provide recreational, aesthetic, and spiritual benefits; and *supporting services* such as soil formation, photosynthesis, and nutrient cycling" (Reid et al., 2005). The economic literature distinguishes between ecosystem services and environmental services (ES). The term "ecosystem services" refers to the benefits provided by ecosystems while "environmental services" refers to the protection of these ecosystems by humans and to the notion of externalities induced by human activities. For example, the Food and Agriculture Organization of the United Nations (FAO) proposes a definition of ES in terms of ecosystem services. For agriculture, ES are defined as the subpart of ecosystem services that can be qualified in terms of externalities, that is, all ecosystem services except provisioning services (Lugo, 2007). ES can be used to refer to the production of services by farmers to protect the environment. We can cite several examples. Long crop rotations improve ecosystem services such as supporting services through improved soil quality. The diversity of productive activities on a farm promotes beneficial interactions between crops and livestock, and the management of landscape features such as grass buffer strips, slopes, hedgerows, or watercourses contribute to the ecological functioning of agroecosystems. All of these definitions make it possible to justify the remuneration for these ES as an internalization of externalities. This leaves room for policy intervention to encourage their optimal provision, such as payments for environmental services (PES), which have become a familiar tool for conserving and restoring ecosystems and the services they provide (Dasgupta, 2021).

#### 2.1.2 Payments for environmental services

One of the most widely cited definitions of PES comes from Wunder (2005), which defines PES as a voluntary transaction in which a well-defined ES or land use that can produce that service is purchased by (at least) one ES buyer from (at least) one ES provider if and only if the ES provider guarantees the provision of ES (conditionality). Conditionality can be difficult to assess in outcome-based PES schemes, as some ES are difficult to measure. In practice, it is much more common to see action-based PES schemes that are conditional on specific land use or management practices.

The above definition illustrates the Coase theorem, whereby an externality can be resolved by private negotiation and the optimal allocation of ES can be achieved, regardless of the initial property allocation and assuming sufficiently low transaction costs and well-defined property rights (Coase, 1960). An example of a Coasean PES

is the Vittel PES in northeastern France, where Nestlé made an agreement with local farmers to compensate them for reducing their use of fertilizers in order to prevent nitrate pollution in the aquifers (Bingham, 2021).

The definition of PES can be expanded to include certain types of government interventions that reflect a Pigouvian subsidy (Pigou, 1920; Sattler and Matzdorf, 2013). This type of PES is much more common in practice than a Coasean PES. For example, European agri-environmental schemes (AES) are publicly funded, and the government acts as an intermediary between the buyers of ES (the public) and the sellers of ES (the farmers who receive PES grants). Both Coasean and Pigouvian PES schemes follow the beneficiary-pays principle.

Yet another alternative definition is provided by Muradian et al. (2010), describing PES as a transfer of resources between social actors, which aims to create incentives to align individual and/or collective land use decisions with social interest in natural resource management. This definition is more flexible than that of Wunder (2005), and better reflects what happens in actual PES schemes than what should happen in theory. This definition also reflects the fact that payments are not necessarily monetary, but can also be in-kind transfers.

Over 550 active PES programs exist globally, with an estimated 36-42 billion USD in annual transactions (Salzman et al., 2018). In the United States, the Conservation Reserve Program has been in place since 1985 (Hellerstein, 2017). Costa Rica is one of the first countries to adopt a national PES program, establishing theirs in 1997 (Pagiola, 2008). In China, the Sloping Land Conversion Program and Natural Forest Conservation Program invested over 50 billion USD from 2000-2009 (Salzman et al., 2018). One program, conceptualized in 2007, that links developed and developing countries is REDD+, which aims to reduce carbon emissions from deforestation and forest degradation by incentivizing developing countries to keep their forests in tact by offering payments financed by developed countries for actions taken to reduce or stop deforestation (Chiroleu-Assouline et al., 2018). Biodiversity protection policies often focus on agricultural lands, such as the AES in Europe, which member states have been required to adopt since 1992. AES are a type of PES that remunerate farmers for voluntary actions taken to preserve and improve the environment. In fact, in the European Union, the largest source for practical nature conservation comes from the AES applied under the Common Agricultural Policy (CAP) (Herzon et al., 2018). Practices adopted under the AES include the reduction of fertilizers and/or pesticides, the establishment of grass buffer strips near rivers, and adaptations to crop rotations. More recently, according to the 2018 National Biodiversity Plan in France, water agencies are testing their own PES programs. They have been allocated 150 million euros from France's national budget, with the objective of maintaining or creating good ecological practices, such as reducing pesticides, planting cover crops, etc. This new program also aims at results-based instead of action-based payments, with annual monitoring to determine the results. Either maintaining or creating good practices will be rewarded, but creating good practices will receive higher payments (up to 676 euros/ha/year, compared to up to 66 euros/ha/year for the maintenance of good practices).

Classifying the wide variety of PES schemes is not an easy task. As Sattler et al. (2013) point out, PES schemes rely on a multitude of approaches that differ greatly in terms of the ES addressed, price formation mechanisms, origins and levels of payment, characteristics of buyers and sellers, rules governing the contract between the parties involved, level of complexity, among others. According to Wunder (2005), the main ES addressed by PES are carbon sequestration and storage, biodiversity protection, watershed protection and landscape beauty.

# 2.2 Are PES programs efficient?

The literature regarding the efficiency of PES programs includes theoretical models, lab experiments, and empirical analyses of real world programs. Most studies evaluate the cost-effectiveness of PES programs, which entails either maximizing ES provision for a given budget, or minimizing the budget to provide a given level of ES.

Many of the obstacles to PES program efficiency originate from the asymmetric information between ES buyers and sellers regarding the costs that ES sellers incur when participating in a PES scheme. The PES literature investigates how asymmetric information impacts the additionality of PES programs and proposes solutions to reduce information asymmetries, such as reverse auctions. Other topics to consider when analyzing PES policy efficiency include transaction costs, the social cost of the public funds raised to finance PES programs, and their long-term impacts.

When designing efficient PES contracts, one must consider the asymmetric information about conservation between the ES seller(s) and the ES buyer(s), whether the buyer is a private individual or a public regulator. Asymmetric information gives rise to two incentive problems: moral hazard (hidden action) and adverse selection (hidden information) (Latacz-Lohmann and Van der Hamsvoort, 1998; Ferraro, 2008). Moral hazard arises from the buyer's imperfect information about contract compliance. Empirically, monitoring studies in England, Germany, and the US have found that as few as 4% or as many as one-third of participants did not fully comply with their contractual obligations (Hart and Latacz-Lohmann, 2005). Adverse selection arises because of asymmetric information about conservation costs, which can lead to informational rents for landowners, as they have incentive to inflate their costs in order to receive higher payments. According to Börner et al. (2017), adverse selection is likely to be more severe where baseline compliance levels are high, payments are too low to cover compliance costs, and program take up rates are low. ES buyers need to reduce informational rents if they want to maximize the environmental services obtained within their budget constraint (Ferraro, 2008). Moreover, selecting the lowest-cost ES sellers can also imply lower ES gains than selecting high-cost sellers.

The rest of this section analyzes factors that influence the efficiency of PES. First, the literature on the additionality of PES programs is explored. Next, we look at whether conservation auctions can improve efficiency compared to fixed payments. Finally, this section concludes with a discussion of other points analyzed in the literature on PES efficiency.

#### 2.2.1 Additionality

One of the main concerns when evaluating PES programs is whether or not they are *additional*; that is whether they lead to the provision of environmental services that would not have been provided otherwise. If a program pays for ES that would have been provided absent any payment, this means they are paying windfall gains to the ES seller. Early on in PES development, a majority of programs had no additionality requirement, possibly due to the idea that monitoring additionality would prove to be too costly (Bennett, 2010). Or, as in the case of the national program in Costa Rica, the aim may be to recognize and remunerate any ES provision regardless of its additionality (Bennett, 2010). It is only more recently that evaluating the additionality of PES programs has become a concern, even though doing so is essential for a PES scheme to achieve its environmental target with economic efficiency while maintaining investor confidence (Bennett, 2010). Adverse selection and moral hazard as described above both offer challenges to achieving additionality, and Sills et al. (2008) describe two other challenges, namely spillovers or leakage, and the possibility that even if there is additionality of a certain land use that is thought to provide certain services, these services may not be additional (Pattanayak et al., 2010). Spillover effects or leakage may occur when preserving some plots of forest leads to increased timber prices, which may incentivize the deforestation of other plots not subject to a PES scheme.

Wunder (2005) explains that establishing a baseline level of ES is essential in order to assess the additionality of a PES program and thus to avoid paying for ES that would have been provided without the program, leading to windfall gains for the ES seller, and a lost opportunity to pay for ES where they would be additional. However, since establishing the baseline level of ES can be prohibitively costly, a regulator or other ES buyer may rely on the ES seller to report this information. This gives the ES seller incentive to under-report their current level of ES provision in order to earn payments for more units of ES provision. When the ES buyer is a public regulator, the issue of additionality is even more important as it prevents wasting public funds. Determining the baseline ES provision for many individual sellers can be quite costly, so Kaczan et al. (2017) look at the possibility of using collective PES schemes to lower this cost. They use a framed field-laboratory experiment with participants from a PES scheme in Mexico and study the impact of conditioning PES payments on an aggregate outcome on group participation and coordination. When the PES payments are conditioned on a group's additionality they find that lower contributors raised their contributions and overall additionality increased.

Additionality concerns are particularly important for carbon markets (Bennett, 2010). Those paying for carbon offset credits risk paying forest managers to protect forest area that would have remained in tact in the absence of any payments. Moreover, leakage of the deforestation activities may occur if a forest PES leads to market conditions that make it more profitable for forest managers in other regions to increase their deforestation rates, thus leading to a displacement of carbon emissions from deforestation rather than a net increase in carbon sequestration. Since the objective behind carbon offsets is to achieve net zero carbon emissions in order to limit global climate change, the additionality of such a program is crucial.

The additionality literature includes theoretical models of conservation contracts. For example, Mason and Plantinga (2013) look at contracts for carbon sequestration from land placed in forest use serving as offsets to meet emissions reduction goals. A government or business seeking to purchase offsets to reduce their emissions will want to minimize expenditures while maximizing additionality, so paying for forests that would remain without a payment would be wasteful. The authors argue there is an adverse selection problem, as only the agent knows how much land would be placed in forest in the absence of any payment. The authors propose offering a menu of contracts to induce agents to reveal their type (in terms of high vs. low opportunity cost of placing land in forest). While it does not achieve the first best solution, the menu of contracts allows for a reduction in government expenditure compared to a uniform payment. Similarly, Chiroleu-Assouline et al. (2018) undertake a theoretical analysis of additionality of REDD+ contracts, which are made between developed and developing countries with the aim of reducing carbon emissions from deforestation and degradation. Using a principal-agent model, they show that dividing developing countries into two groups based on two different policy instruments can help the developed country obtain efficient deforestation and avoided deforestation levels from their payments.

Pates and Hendricks (2020) take another approach to additionality, framing nonadditionality as a moral hazard problem in a technology diffusion context. They look at the case where a new and more environmentally friendly technology becomes less expensive to adopt over time, and whose adoption might be subsidized. They argue that an agent may delay adoption of the technology in order to get a subsidy for adoption in a future time period, which is an example of moral hazard since the agent changes his behavior in response to the policy. After developing a conceptual model, the authors run numerical simulations and find that the moral hazard results in a non-monotonic relationship between different policy parameters (e.g. budget or payment size) and the change in technology adoption rates linked to the PES policy. Furthermore, they find that the cost-effectiveness of such a policy is lower when the policy is introduced at a time of rapid technology adoption.

Other papers have investigated the empirical evidence of additionality in PES schemes. For example, Chabé-Ferret and Subervie (2013) study five agro-environment schemes (AES) implemented in France to estimate their additional and windfall effects. They find different levels of additionality for the different AES, with the more stringent requirements leading to higher additionality. Mezzatesta et al. (2013) use propensity score matching to evaluate the additionality of the Conservation Reserve Program in the US in regard to six conservation practices: conservation tillage, cover crops, hayfield establishment, grid sampling, grass waterways, and filter strips. Based on survey data of farmers in the state of Ohio, they calculate the average treated effect on the treated (ATT), defined as "the average increase in the proportion of the land adopted in a conservation practice for enrolled farmers relative to their counterfactual proportion of the land in this practice that they would have

adopted without funding" (Mezzatesta et al., 2013). The authors find that while the overall ATT of the program is positive and statistically significant for each of the conservation practices, the degree of additionality varies across the practices, with hayfield establishment having the highest additionality and conservation tillage the lowest. Jones et al. (2020) look at the additionality of a PES in terms of forest cover and subsequent effects on hydrological services and find that the PES reduces losses but does not provide many gains in forest cover. Furthermore, they find that when there is no additionality in forest cover due to the PES the result is economic loss. Next, Mohebalian and Aguilar (2016) use GIS data to investigate the additionality of a forest PES program in Ecuador and their findings suggest that the PES program has provided little additionality in terms of preventing deforestation. Finally, Ezzine-de Blas et al. (2016) perform a meta-analysis on 55 PES programs around the world and find that additionality is positively influenced by spatial targeting, payment differentiation, and strong conditionality.

#### 2.2.2 Using auctions to reveal conservation cost information

One well-studied policy instrument that could be used to alleviate the problems arising from asymmetric information and to improve the economic efficiency of PES schemes is the reverse auction, where ES sellers submit bids (based on their conservation costs) to the ES buyer, who selects the lowest bidder(s). Latacz-Lohmann and Van der Hamsvoort (1997) is the first paper to look at using reverse auctions for conservation contracts. Using simulations of a theoretical model they demonstrate that, compared to fixed rate PES payments, competitive bidding can significantly increase the cost-effectiveness of an agricultural PES program by reducing the information rents for the bidders and by encouraging higher opportunity cost farmers to participate, which can increase environmental effectiveness. Furthermore, these advantages of auctions compared to fixed payments increase when the initial information asymmetry is more severe. The authors analyze the farmer's optimal bidding strategy, which is to maximize net profit while maintaining a high enough probability of bid acceptance, given his expectations of the maximum acceptable bid. Furthermore, bidding strategy will depend on the risk attitude of the farmer. If the farmer is risk-neutral, his optimal bidding strategy will be a linearly increasing function of opportunity costs and the expected bid cap. For a risk averse farmer, the conservation payment can be considered as a non-stochastic income component, and the optimal bid will depend on foregone profits and the difference in risk premiums.

Another theoretical model by Lewis and Polasky (2018) builds on previous works from

Vickrey (1961) Clarke (1971), Groves (1973), and Polasky et al. (2014) and defines an auction mechanism that implements optimal environmental service provision over time. They allow for the benefits from the environmental services to change over time, depending on the landscape pattern and climate change. Their auction design leads to landowners truthfully revealing their private information about the returns they would receive in an alternate use that does not provide environmental services while bidding on current and future development values of their land. Additionally, their auction mechanism can either be used in a procurement auction where bidders compete for a subsidy, or it can be used in a normal auction where bidders compete for development rights on the land, which is useful when environmentally important areas are owned by the government.

Auctions are used to allocate PES contracts in different programs around the world. The United States Department of Agriculture (USDA) has been using an auction mechanism to allocate land retirement contracts through the Conservation Reserve Program (CRP) since 1986 (Latacz-Lohmann and Van der Hamsvoort, 1997). Latacz-Lohmann and Van der Hamsvoort (1998) describe two types of auctions that can be used for countryside public goods provision. The first type, which the US CRP exemplifies, is a government procurement auction, where a government agency acts as the buyer and the farmers act as sellers, and the commodity being traded is a public good. The second type is an auction of certificates, where the traded commodity is property rights to receive pre-determined financial rewards for the provision of countryside benefits, with the government agency acting as the seller and farmers as the buyers, paying in terms of environmental commitments.

#### 2.2.2.1 Factors affecting auction efficiency

Different design features and rules can impact auction efficiency in different contexts. Uniform price auctions pay all winning bidders the same level, often the highest accepted bid, while discriminatory price auctions pay each winning bidder their bid level. Uniform price offers more incentive for bidders to reveal their true costs since the payment level is not necessarily the same as their bid. Sealed bidding keep bids private while open bidding allows bidders to know each other's bids. Auctions may be budget constrained, where they accept the lowest bids up until a budget level is reached. Alternatively, they could be target constrained, i.e., the lowest bids are accepted until a certain area of land is enrolled. The criteria used to accept or reject bids may also combine the price with an estimated value of the environmental benefits that would be provided. For example, the US CRP procurement auction uses an environmental benefits index (EBI) to select land for enrollment (Claassen et al., 2008). Finally, whether or not auctions are repeated will impact their efficiency as bidders have the opportunity to adjust their expectations about the maximum acceptable bid when auctions are repeated.

Much of the work on conservation auctions has involved lab experiments and Schilizzi (2017) provides an overview of results from these studies that analyze different causal and outcome factors of auction features. In regard to the format of the auction, the author's key findings include that discriminatory price auctions perform better than uniform price auctions in terms of cost-effectiveness, despite bid shading, although uniform pricing is better for learning about unknown conservation costs. Kawasaki et al. (2012) looks at compliance in conservation auctions both theoretically and using a lab experiment. While the theoretical analysis predicts that budgetary costeffectiveness, allocative efficiency, total payment, and number of compliant winners will be equal between the two payment types, the lab experiment resulted in higher budgetary cost-effectiveness and allocative efficiency when payments were uniform than when they were discriminatory. Furthermore, they found that in a repeated auction setting budget-constrained auctions are more robust than target-constrained auctions. When communication is allowed among bidders, uniform pricing mitigates collusion rents relative to discriminatory pricing. Similarly, when cost curves are more heterogeneous, uniform pricing should also perform better than discriminatory pricing. Arnold et al. (2013) found that a budget-constrained discriminatory auction may decrease the pool of potential bidders, inducing adverse selection. Another interesting result comes from Cason and Gangadharan (2004) who summarize a lab experiment that showed that landowners increased their bids when they were given information before bidding about the environmental benefits of the actions they planned to take if they won the contract. Glebe (2013) later confirmed this finding theoretically, but also found that revealing environmental benefit information could induce higher participation rates for the auction and a thicker market.

A more recent paper by Santos et al. (2021) incorporates the idea of asset specificity to conservation auctions to see how cost-effectiveness is impacted. They argue that participating in a PES scheme can require making investments that are specific to ES provision and these investments yield lower returns in any alternative use. Consequently, this can lead to potential PES participants requiring higher payments to compensate for risk, or asking for ex-ante rather than ex-post payments, which can lower the incentive to comply with program requirements. The authors use an agent-based model to simulate a PES scheme that aims to increase carbon capture by paying for the retirement of land from agricultural production, the investment in reforestation, and the conservation of these trees. They evaluate the cost-effectiveness of uniform and discriminatory auctions to allocate the PES contracts to liquidity constrained land users who require ex-ante payments in order to finance their initial investment. Their results imply that a higher number of long-term contracts can be allocated by auctions as compared to fixed payments. Moreover, they find that discriminatory auctions are more additional and cost-effective than uniform price auctions. When land users have high time preferences (and thus high discount rates), short-term contracts are more cost-effective than long-term contracts. Finally, they find that asset specificity, liquidity constraints, high time preferences, and absence of trust in the agency can all exacerbate the challenge of asymmetric information.

#### 2.2.2.2 Potential drawbacks of auctions

Auctions for public goods such as environmental services diverge from the standard auction model in several ways, which can make them more vulnerable to failure. Compared to fixed payments, auctions provide the most benefits when the informational asymmetry and number of potential participants are high, contracts are homogeneous, farms are heterogeneous, and when the environmental service production is separable between farms (Latacz-Lohmann and Van der Hamsvoort, 1998). The two main drawbacks of using auctions in the provision of public goods are higher transaction costs than fixed payments and the potential for strategic bidding due to the common-value characteristic of public goods (Latacz-Lohmann and Van der Hamsvoort, 1998).

Additionally, conservation contract auctions also generally involve multiple sign up periods (sequential auctions), which can be affected by Bayesian learning, leading to higher average bids and a smaller distribution of bids as bidders learn what the highest accepted bid is (Latacz-Lohmann and Van der Hamsvoort, 1997). In fact, Schilizzi and Latacz-Lohmann (2007) found that while both budget-constrained and target-constrained discriminatory price auctions had higher cost-effectiveness than fixed payments in a single time period, the gain from using an auction diminished when considering multiple time periods.

Indeed, while much of the literature on conservation auctions suggests they are likely more efficient than fixed payments, other studies investigate scenarios where auctions do not improve the cost-effectiveness of PES programs. Lundberg et al. (2018) use a conceptual agent based simulation model to assess the cost-effectiveness of fixed payments and procurement auctions in different contexts. They find that contextual factors such as baseline compliance rates, the distribution of provision costs across the landscape, and the correlation of provision costs and ES provision affect whether fixed payments or procurement auctions are more cost-effective. Their results show that discriminatory auctions are likely to be the most effective design when baseline compliance is low and/or when differentiated payments are politically feasible, i.e. perceived to be fair. A fixed payment scheme that offers high payments and uses targeting is more likely to be additional in contexts where baseline compliance is high, the correlation between ES provision and costs is positive, and/or the budget for the program is small. In this context, auctions could exacerbate the adverse selection problem, leading to lower additionality and thus lower cost-effectiveness. However, even when fixed payments are preferred, an initial one-shot auction may be useful to gather information about provision costs in order to set an appropriate price for the fixed price scheme.

Arguedas and van Soest (2011) evaluate the cost-effectiveness of using a menu of contracts, as suggested by previous literature (Smith, 1995; Smith and Tomasi, 1995; Wu and Babcock, 1995, 1996; Moxey et al., 1999) compared to procurement auctions. While offering a menu of contracts can improve efficiency compared to uniform fixed payments, contract menus are rarely used in practice (Ferraro, 2008). Designing a menu of contracts requires more information, which can be costly to acquire, and the menu of contracts that would be offered under perfect information is typically not incentive-compatible where there is asymmetric information, meaning the net benefits of offering a menu of contracts may be too low to justify the costs of acquiring the necessary information. However, Arguedas and van Soest (2011) show that if conservation costs include fixed costs as well as variable costs the complete information solution can be incentive compatible in the presence of asymmetric information, as long as ES providers with lower variable costs face higher fixed costs (and vice versa). This is likely to occur if the same circumstances are conducive to both conservation and production. For example, hydrological circumstances in arid regions can lead to negatively correlated fixed and variable costs of conservation as less access to water can make the fixed costs of restoring habitat higher, but variable opportunity costs lower as the land would be less productive for agriculture.

#### 2.2.3 Other considerations

Here, we look at transaction costs, other benefits of PES, the duration of contracts, motivational crowding, and the marginal social cost of public funds.

#### 2.2.3.1 Transaction costs

The transaction costs involved in implementing PES policies are another main obstacle to achieving efficiency (Börner et al., 2017). Indeed, Salzman et al. (2018) cite low-transaction-cost institutions as one of four key features needed to scale up PES. Williamson (1975) distinguishes between ex-ante and ex-post transaction costs, with the former including the costs of information gathering, preparing, and negotiating a conservation agreement, while the latter includes the costs of monitoring compliance and any conflict resolution costs. There is often a tradeoff between improving the effectiveness of a policy and increasing transaction costs. For example, the ex-ante transaction costs for an auction are likely to be higher than for a fixed payment scheme (Latacz-Lohmann and Van der Hamsvoort, 1998). Similarly, a PES that includes spatial targeting to improve cost-effectiveness implies higher transaction costs (Uthes et al., 2010).

Armsworth et al. (2012) uses an ecological-economic model to assess the effectiveness of different payment schemes that vary in complexity, and thus transaction costs. They define an optimal policy design that maximizes biodiversity improvement for a given budget while leaving each farmer indifferent between participating or not in the PES, i.e. the payments only cover the participation costs and offer no windfall gains. Next, they evaluate policies that add simplifications, namely using a uniform fixed payment and no spatial targeting, both of which are commonly used to simplify existing agri-environmental schemes. Compared to the optimal policy, they find that the policy simplifications lead to a 49-100% reduction in biodiversity provision for a given budget level. Moreover, they find that the added implementation costs of the more complex optimal policy are worth incurring, even if they amount to over 70% of the payments that would have otherwise gone to the participating farmers.

#### 2.2.3.2 Other benefits to consider

Although levying taxes to finance PES programs can lead to social costs, PES programs can sometimes offer benefits other than environmental services. A strand of the literature on PES in developing countries looks at poverty alleviation as a possible secondary objective of PES programs (e.g., Suich et al. (2015); Wünscher and Wunder (2017); Ola et al. (2019) among others). PES programs may also have other positive impacts in developed countries as well. Schirpke et al. (2018) evaluated the socio-economic benefits of 50 cases in 21 Natura 2000 sites in Italy and found that, in addition to improvements in defined conservation objectives, there were positive effects on socio-economic development of the local communities.

#### 2.2.3.3 Contract duration

Most existing PES programs involve contracts for relatively short periods of time, often five or ten years, raising the question of whether ES provision will continue or be reversed once payments end. This is commonly referred to in the literature as permanence, and sometimes also referred to as persistence, continuance, maintenance, or confirmation (Dayer et al., 2018). Empirically, a recent study by Kemigisha et al. (2023) found weak permanence in a PES program in Uganda where it was clear that forest owners had abandoned the PES-induced practices four years after payments ended.

#### 2.2.3.4 Motivation crowding: what happens after PES?

Related to the permanence concern is the idea of motivational crowding, the idea that extrinsic motivations, such as financial payments like PES, can "crowd out" the intrinsic motivations in individuals that may arise from altruism or pro-social attitudes and undermine long-term conservation efforts (Kosoy and Corbera, 2010; Vatn, 2010; Gómez-Baggethun and Ruiz-Pérez, 2011; Luck et al., 2012; De Snoo et al., 2013; Muradian et al., 2013). Alternatively, participation in a PES scheme could potentially "crowd in" intrinsic motivation for conservation. Whether a PES scheme crowds in or crowds out intrinsic motivations will impact the likelihood of the scheme persisting after payments end. This is important, as some PES programs state permanence of the land uses or management techniques being incentivized as an objective of the program. For example, the National Landcare Program in Australia is based on the idea that investments will lead to long-term changes in management practices (Dayer et al., 2018). Pro-environmental attitude shifts are also an implicit goal of voluntary agri-environmental schemes in the EU (Burton et al., 2008).

Albers et al. (2008) investigate whether motivation crowding occurs when a public and a private entity interact in conservation. They look at how properties of the conservation benefit function can impact the interaction between the private and public entities. Specifically, they look at how results change when the private entity's benefit function displays diminishing marginal benefits versus increasing marginal benefits. They find that whether increased public conservation crowds out private conservation will depend on the trust's benefit function. If it displays diminishing marginal benefits private conservation will be crowded out, whereas if it displays increasing marginal benefits, private conservation will be crowded in. Dayer et al. (2018) suggest five theory-based explanations, which likely covary and interact with one another, for whether conservation behaviors can be expected to persist or revert when payments stop. First, they address landowner cognitions, i.e. attitudes toward the environment and certain management practices, and posit that if a conservation program can positively change landowner cognitions, permanence is more likely to occur. Empirically, Kuhfuss et al. (016a) find that French farmers who perceived a higher quality of life while participating in an agri-environmental scheme were more likely to state an intention to continue the conservation behaviors after payments ended whereas those who experienced technical difficulties had 50% lower intentions to persist. A second explanation is motivation crowding. Indeed, Deci et al. (1999) performed a meta-analysis of social psychology experiments and found evidence of the crowding out effect. Regarding PES programs, several papers have found evidence of crowding out effects (De Snoo et al., 2013; Rode et al., 2015). For example, Kits et al. (2014) uses a lab experiment to evaluate motivation crowding in conservation auctions and found an 18% reduction in beneficial management practices due to motivational crowding. However, Rode et al. (2015) also found four studies that indicated crowding in effects, though out of these four only Rodriguez-Sickert et al. (2008) presented a statistically significant result. Third, if the conservation behavior is habit forming (i.e. simple and frequently repeated actions) it will be more likely to persist after payments. Fourth, they discuss the importance of financial resources, and, of course, having sufficient resources absent any payment to continue the behavior makes persistence more likely. If incentivized practices generate a better sale value for farm products (e.g. a price premium for an eco-labeled product), a farmer will be more likely to continue the practices once the program ends (Kuhfuss et al., 016b; Dayer et al., 2018).

#### 2.2.3.5 Social cost of public funds

Implementing a publicly funded PES program requires raising public funds, which can change the allocation of resources in an economy through impacts on consumption, labor, and investment decisions (Dahlby, 2008). The marginal social cost of public funds (MCF) is a measure of the welfare loss to society as a result of raising additional revenues to finance government spending (Browning, 1976; Dahlby, 2008). Empirically, Browning (1976) estimates the MCF of labor income taxes in the United States, finding a MCF of \$1.09-\$1.16 per dollar tax revenue raised. According to Beaud (2008), this cost is equal to 1.2 for France. Most of the PES literature does not explicitly take the MCF into account when evaluating efficiency, though some papers do acknowledge its impact, e.g. Ferraro (2008). The general lack of consideration of the MCF when evaluating PES efficiency is a clear gap in the literature.

# 2.3 Agglomeration Bonus

Habitat fragmentation, the process of dividing a contiguous area of natural habitat into smaller, more isolated patches is deemed as a key pressure contributing to biodiversity loss (Wilcove et al., 1986; Fooks et al., 2016). So, the spatial configuration of areas used for biodiversity conservation matters as it impacts the chance a species has to survive. Most commonly, connected land is thought to be better for preserving species (Albers et al., 2018). There are species, however, that prefer other spatial configurations, and this will be briefly discussed later in the literature review. The importance of the spatial connection of habitats has led to the idea of an Agglomeration Bonus (AB), an extension of PES schemes that aims to incentivize neighboring landowners to coordinate their land management decisions to connect habitat conservation areas (Banerjee et al., 2011). This mechanism was first proposed by Parkhurst et al. (2002) and has since been explored by numerous other authors. The agglomeration bonus is a bonus payment on top of a flat participation payment that is awarded to landowners when they conserve contiguous parcels of land. Another possibility is an agglomeration payment, where instead of offering a bonus, the participation payment is also contingent upon achieving contiguous conserved land.

The main drawback of an agglomeration bonus mechanism is the potential for multiple Nash equilibria which entails the classic coordination problem (Parkhurst et al., 2002). Amongst all possible Nash equilibria, there is only one first best equilibrium for society, called the Pareto dominant Nash equilibrium (Parkhurst and Shogren, 2007). The challenge is then getting landowners to coordinate in a way to conserve land that results in the Pareto dominant Nash equilibrium rather than any other equilibrium. Several papers indeed find the existence of multiple Nash equilibria in different scenarios in their models, and coordination failure is documented in some experimental papers (Albers et al., 2008).

Following the suggestion in Parkhurst et al. (2002) to use an agglomeration bonus, other authors have used simulations and experiments to study different properties of agglomeration bonuses. The first subsection will address how the literature evaluates the success of the agglomeration bonus mechanism in regard to conservation goals, compared to other mechanisms. Next, some authors look at how changing the design
or accounting for transaction costs impacts the efficiency of the agglomeration bonus. Finally, this section concludes with a discussion of existing agglomeration bonus schemes in the US and Switzerland.

#### 2.3.1 Evaluating the Agglomeration Bonus

The literature addressed in this review mostly evaluates agglomeration bonuses based on cost-effectiveness (Wätzold and Drechsler, 2014; Bell et al., 2016; Bamière et al., 2013; Liu et al., 2019; Kuhfuss et al., 016a; Delacote et al., 2016) and environmental effectiveness, (Krämer and Wätzold, 2018; Bell et al., 2016; Parkhurst and Shogren, 2007, 2008; Liu et al., 2019). The latter is usually measured through proxies such as amount of land enrolled in a program or to what extent the desired configuration of land is achieved rather than through environmental indicators, which can be costly to measure. Other criteria used to analyze the efficiency of the agglomeration bonus mechanism include economic efficiency (Parkhurst and Shogren, 2007; Fooks et al., 2016; Albers et al., 2008; Grout, 2009), and fairness (Ferré et al., 2018; Drechsler, 2017).

#### 2.3.1.1 Cost-effectiveness

Cost-effectiveness is a measure to compare different policy tools to see which tool either minimizes the cost of achieving a certain objective or maximizes outcomes given a certain budget constraint. Accordingly, there are several papers that compare agglomeration bonuses to standard PES programs or other policies to see if they can improve the cost-effectiveness. A couple papers find that including an agglomeration bonus improves cost-effectiveness. First, Bell et al. (2016) develops an agent-based model and finds that including an agglomeration bonus can improve cost-effectiveness, so long as some monitoring effort is present. Next, Kuhfuss et al. (016a) examines both the decision to enroll in a program and the decision of how many hectares to enroll. They find that once a bonus is introduced the minimum willingness to accept for participants is lowered by an amount that is even greater than the value of the bonus. They likewise find that including the bonus also encourages enrolling a larger share of land in the contract, given the decision to enroll.

Conversely, some papers find less clear results about how agglomeration bonuses affect cost-effectiveness. For example, Delacote et al. (2016) develop a spatial model and run simulations to measure leakage resulting from different avoided deforestation policies. They compare two different PES policies with an agglomeration bonus and a conservation area policy, while looking at four cases where similar patches are either clustered or dispersed and there is either high or low interaction between landowners. They find that the spatial distribution affects which policy is most cost-effective. For example, the agglomeration bonus results in the least leakage per unit cost in the case where similar patches are clustered and there is low interaction between landowners, but it has intermediate cost-effectiveness for leakage in the three other cases. Furthermore, the agglomeration bonus results in the lowest avoided deforestation per unit cost in both dispersed cases, making it the least cost-effective for this measure. Next, Wätzold and Drechsler (2014) compare an agglomeration payment where the base payment and bonus are both contingent on the spatial connectivity, an agglomeration bonus where only the bonus is contingent on the spatial connectivity, and a homogeneous payment where landowners receive a base payment for conserved land parcels. Their results show that an agglomeration bonus never substantially dominates the two alternatives. However, they find that the agglomeration payment performs better than the homogeneous payment in most scenarios in terms of cost-effectiveness.

Another part of the literature looks at using agglomeration bonuses with procurement auctions. For example, Liu et al. (2019) conducts a framed field experiment, using forest owners in rural China as subjects, and investigates the use of an agglomeration bonus in conjunction with a PES auction for conservation contracts. While they find that overall bidders in the agglomeration bonus treatment group bid less than their counterparts in the control group in anticipation of receiving a bonus, they caution that this result is only statistically significant for a subsample that excludes players who clearly misunderstood the experiment or who exhibited protest attitudes. Furthermore, they suggest that while bids may decrease, the agglomeration bonus may include 'nontrivial additional costs' compared to the control group, as total costs become similar between the two groups once the bonus payments are taken into account. Next, Schilizzi (2017) found that using an agglomeration bonus in an auction setting increases the size and quality of the pool of submitted offers.

Bamière et al. (2013) study a unique case where the ideal spatial pattern of conserved land is a random mosaic rather than contiguous habitat. They compare a per-hectare subsidy, an auction, and an agglomeration malus (a penalty for conserving a parcel adjacent to an existing reserve since contiguity is not desired here) when the objective is reaching 15% reserved land. They argue that there is a trade-off between minimizing public costs and reaching the desired spatial pattern, since their results show that the auction is the most cost-efficient tool, but the agglomeration malus achieves a more desirable spatial pattern.

#### 2.3.1.2 Environmental effectiveness

Environmental effectiveness can be prohibitively costly to measure, which is why most studies rely on proxy measures, such as the total area of land enrolled in a PES program and the degree of connectivity. Some studies find that using an agglomeration bonus has a positive impact on environmental effectiveness, though others find mixed results.

Starting with papers that find positive impacts on environmental effectiveness, Parkhurst and Shogren (2007, 2008) conduct lab experiments using a spatial coordination game to test the robustness of an agglomeration bonus. They consider four different conservation targets that consist of different configurations of connected habitats. They measure the percentage of the desired conservation objective that is achieved, denoting this measure biological efficiency. These papers find that an agglomeration bonus leads to higher biological efficiency, and that this efficiency increases with experience as players play more rounds, but decreases with increased complexity of the coordination problem. Parkhurst and Shogren (2008) find that including an agglomeration bonus led to higher biological efficiency than current policies such as fixed payments or compulsion. Next, Bell et al. (2016) use an agent based model and find that agglomeration payments have the potential to increase the adoption of pro-environmental practices as well as improve the contiguity of these practices.

Other studies find caveats to the positive impacts on environmental effectiveness. Fooks et al. (2016) run an experiment using both landowners and students as participants. They look at both spatial targeting and network bonuses in the context of a reverse auction, evaluating four different scenarios: one where only spatial targeting is used, one where only a network bonus is used, one where both are used, and finally a scenario where neither is used. They model a function of the value the buyer places on specific configurations of enrolled parcels in order to measure environmental effectiveness. Their results show that the bonus was able to induce more parcels to enter the auction, especially those that are higher value in terms of the conservation objective. However, they find that network bonuses on their own result in worse outcomes than the baseline reverse auction. When network bonuses are combined with spatial targeting, welfare outcomes are positively affected, suggesting that spatial targeting can mitigate the welfare loss from an auction with a network bonus. Next, using avoided deforestation and leakage as measures of environmental effectiveness, Delacote et al. (2016) find that an agglomeration policy reduces the amount of leakage in an avoided deforestation scheme, but also reduces avoided deforestation within the intervention area. They point out that increasing the direct payment component can help counter the reduction in avoided deforestation, but this of course leads to higher program costs. Additionally, they caution that they only look at the service of carbon sequestration, and avoided deforestation can impact other ecosystem services, such as flood control. Finally, Liu et al. (2019) hypothesize that using an agglomeration bonus in an auction will lead bidders with more neighbors to lower their bids in order to increase their probability of getting selected into the program, and that this would promote more contiguous conservation, However, their experimental data does not show a statistically significant difference between contiguity and connectivity indicators in the treated and control groups.

#### 2.3.1.3 Economic efficiency

Very few papers evaluate the success of the agglomeration bonus based on economic efficiency. This measure can be difficult to estimate as there is a lot of uncertainty around the monetary value of the environmental benefits and what the optimal allocation of ecosystem services should be. Furthermore, increased provision of one ecosystem service can impact the provision of other ecosystem services, either in a positive or negative way, which can complicate the calculation of the optimal allocation. Nevertheless, the following papers seek to evaluate the economic efficiency of agglomeration bonuses.

Albers et al. (2008) examines the impact of public policy on the amount and configuration of privately conserved land and the extent to which the social optimum is achieved in different scenarios. In their model, agents face the same cost function, but can vary in regard to benefit functions and budget constraints. When analysing the impact of an agglomeration bonus, the authors look at two land trusts who each have preferences about agglomeration. When one trust has a preference for agglomeration while the other is indifferent, an agglomeration bonus, even if it is very small, will induce the socially optimal conservation pattern with certainty in a sequential move game, though it will have less success in a simultaneous move game due to coordination failure. If at least one of the trusts has preferences against adjacent conservation patterns, the agglomeration bonus should compensate the trust for any disamenity it experiences from conserving agglomerated land. A final caveat for this study is that it assumes perfect information about agents' preferences, whereas in reality one agent is not likely to fully know another agent's preferences.

## 2.3.2 Agglomeration bonus design

Since the inception of the idea of an agglomeration bonus by Parkhurst et al. (2002), the literature has expanded to include analyses on different ways to design an agglomeration bonus and the impacts of design choices on its performance. The papers below look at such factors as side payments and communication between landowners, the role of transaction costs, and how payments should be structured.

Several papers allow for side payments between landowners as a way for one landowner to incentivize his neighbor to participate when the neighbor has opportunity costs that otherwise outweigh their private benefits from enrolling in a conservation program. For this to occur, the agglomeration bonus must be large enough such that the landowner can cover the difference between his neighbor's opportunity costs and the conservation payment. The literature finds that side payments can improve the performance of agglomeration bonuses (Albers et al., 2008; Bell et al., 2016). Interestingly, other motives can lead to side payments between landowners, for example, Ferré et al. (2018) find that players in their experiment who have no economic incentive to offer a side payment sometimes do so in the interest of diminishing payoff inequality.

Related to side payments, is whether or not experiments allow for communication between subjects as they decide whether or not to enroll in the incentive program. A handful of studies look at the impact of communication on the performance of the agglomeration bonus. Banerjee et al. (2017) allow communication in some groups but not others, and they impose a small messaging fee to communicate with a neighbor. They find that participation rate in the agglomeration bonus scheme is on average higher in groups with communication than those without communication, for both high and low transaction cost settings. Moreover, they find that allowing communication increases performance of the agglomeration bonus when transaction costs are high, though the impact of communication when transaction costs are low depends on whether participants previously experienced the high transaction cost setting. Parkhurst et al. (2002) find that allowing communication before longer-term contracts could be a key to contiguous habitat creation with the agglomeration bonus mechanism.

Bell et al. (2016) include communication, side payments, and monitoring in their agent based model. Communication occurs during an interaction step in which farmers choose to participate with a given probability. If they choose to participate in the interaction step, they exchange information on past yields and fines for cheating

(when cheating and monitoring has occurred). They find that side payments only account for at most 4% of adoption in cases with low overall program spending. Monitoring is represented by a probability of getting caught cheating, and a higher probability means higher monitoring costs. They find three main consequences of including an agglomeration bonus in a PES program. First, absolute uptake is increased. Second, overall payment costs may decrease. And third, program monitoring demands may be reduced. Another finding they discuss is that when there is some monitoring effort, the agglomeration bonus can improve the costeffectiveness of the program, which can be partly attributed to a peer effect related to side payments, since a farmer is unlikely to honor a side payment offer if their neighbor cheats, so if the neighbor wants to guarantee that they receive the side payment, they need to adopt the conservation practice.

Some papers try to incorporate transaction costs into their analyses. For example, Banerjee et al. (2017) use a lab experiment to find the effect of varying transaction costs on participation rates and spatial coordination in an agglomeration bonus scheme. They look at two levels of transaction costs, high and low, and find that when transaction costs are high, there are significantly lower participation rates in the bonus scheme, though communication somewhat mitigates this effect. Next, they look at the impact on spatial coordination on a circular network, both locally (coordinating with your two direct neighbors) and globally (all eight group members choose the same land use). Without communication, global coordination is not impacted by different transaction costs, while allowing communication improves global coordination.

In regard to the magnitude of agglomeration payments over time, Ferré et al. (2018) look at whether agglomeration payments should remain constant over time, or change with changing opportunity costs. They analyze this question in the context of rewetting organic soils in Switzerland and find that, in terms of funds spent per unit of preserved peat soil, the constant payment has the highest cost-effectiveness compared to the variable payment and the baseline scenarios. Their experiment shows that the constant agglomeration bonus outperforms both the variable agglomeration payment and the baseline scenario in terms of participation. Additionally, unlike with the variable agglomeration payment, a majority of players are able to coordinate over all ten time periods with the constant agglomeration payment.

Finally, Grout (2009) proposes a conditional agglomeration bonus to mitigate the complexity of the coordination problem. Essentially, he proposes that a landowner's

agreement to enroll his or her land is only binding if the desired spatial configuration is met. The paper argues that this modification improves the applicability of the agglomeration bonus by limiting the information burden for landowners to their own opportunity costs and eliminating the need to coordinate enrollment decisions with the other landowners. Making enrollment and payments conditional on achieving a certain spatial configuration results in an expected payoff of zero if coordination fails to achieve the spatial configuration. So, each landowner then only has to evaluate if their expected payoff is greater than zero if the spatial configuration is achieved, and they no longer have to consider the decisions of the other landowners.

#### 2.3.3 Existing programs with agglomeration bonuses

While the theoretical and experimental literature on agglomeration bonuses is growing, there are very few real world examples of PES programs that include an agglomeration bonus or agglomeration payment. In fact, only two such programs are cited in the literature: the Conservation Reserve Enhancement Program in Oregon, in the United States, which expands on the national Conservation Reserve Program in the US, and the Swiss network bonus for connected ecological compensation areas. These programs are briefly explained in the following subsections.

#### 2.3.3.1 Conservation Reserve Enhancement Program

The United States has a national program for land conservation called the Conservation Reserve Program, which allocates contracts via auction to farmers who then receive payments to take land out of crop production in order to favor conservation measures aimed at environmental objectives such as reducing soil erosion, improving air and water quality, and providing wildlife habitat (Grout, 2009). On a state level, Oregon state signed an agreement with the US Department of Agriculture (USDA) in 1998 to create the Conservation Reserve Enhancement Program (CREP), which pays a one-time bonus to participants along a stream if at least 50% of the stream bank within a 5-mile stream segment is enrolled in the CRP, though the retired acres are not required to be contiguous (Parkhurst et al., 2002).

The CREP aims to restore, maintain, and enhance riparian areas along agricultural lands to benefit fish, wildlife, and water quality. The program includes annual rental payments and a one-time bonus equal to four times the annual rental rate for each acre enrolled (Parkhurst et al., 2002). This 'cumulative impact incentive payment' rewards landowners (or a group of landowners) if their riparian establishment is more than 2.5 miles of a 5-mile stream segment. Contracts last for 10-15 years. Since the start of the program, roughly 38,500 acres have been enrolled, and the State of Oregon has invested around \$25.4 million since 1999 (Oregon Watershed Enhancement Board, 2018). The 2018 CREP Annual Report does not present the monitoring results, but states that approximately 300 site visits and inspections took place in the fiscal year of 2018.

#### 2.3.3.2 Swiss network bonus

Switzerland implemented their Agricultural Act in 1993, which included financial support for 'ecological compensation areas' (Krämer and Wätzold, 2018). While the share of ecological compensation areas increased over time, there was criticism that they were not effective in conserving biodiversity. As a result, two types of payments on top of the standard payment were introduced: a 'quality bonus' for meeting exceptional biological quality, and a 'network bonus' for creating connected ecological compensation areas (Krämer and Wätzold, 2018). Requirements for the network bonus are decided at the canton level, but have to fulfill minimum standards that are set at the national level. To facilitate the monitoring process, the initial situation of a project area is documented on a map.

Krämer and Wätzold (2018) investigate three case studies of projects in the Swiss network bonus program. While they caution that the results are not necessarily representative for Switzerland as a whole, they provide some conclusions. First, they find that overall, the introduction of the network and quality bonuses led to an increase in the size of contracted ecological compensation areas, with an increase in participation and/or an increase in the size of conservation areas. They also find that the overall quality of areas that were contracted before the introduction of the network bonus has improved. Despite evidence of some improvements, it is still somewhat unclear to what extent the improved quality and increased emphasis on spatial coordination has had a positive impact on biodiversity in general and the target species in particular, so the authors call for better monitoring data in order to more clearly assess these outcomes.

## 2.4 Mitigation banking: PES using tradable permits

The United States Clean Water Act of 1972 laid the groundwork for mitigation banking by requiring permits for projects that would negatively impact wetland area. In order to obtain said permit, one must follow what is called the mitigation hierarchy. The policy objective of the mitigation hierarchy is often 'no net loss' (NNL) of biodiversity, though sometimes it is more ambitious and aims for a net gain of biodiversity. When designing a project that will have negative impacts on the surrounding environment, a developer must first avoid any possible damages by altering the project; then, damages that cannot be avoided must be reduced; and finally, any residual damage after following the first two steps must be compensated, either by restoring another equivalent area of land or by purchasing offset credits from a mitigation bank. This is also referred to as the 'avoid, reduce, compensate' (ARC) sequence. Using a mitigation bank allows for the compensation activities to be carried out by another agent, who can then sell credits to the mitigation bank that developers can then purchase. The US wetland mitigation banking program was the first market of tradable permits to protect biodiversity (de Muelenaere, 2011) and the idea of habitat banking and biodiversity offsets has since spread around the world, with at least 3,000 offset projects recorded worldwide covering at least 153,670 sq km (Bull and Strange, 2018).

Mitigation banks can be considered a type of PES, as both instruments involve providing payment for the restoration, preservation and/or management of biodiversity and ecosystems (Bureau, 2010; Salzman et al., 2018; Combe, 2020). Rather than using a Pigouvian subsidy, mitigation banking involves tradable permits, usually in the form of biodiversity offsets, and thus is a quantity-based rather than a price-based instrument. In contrast to tradable pollution permits, where a regulator determines the supply of permits by setting a cap on the amount of pollution allowed, offset market supply is endogenous, since the supply of permits depends on the number, size, and quality of compensation areas being created, which is in turn influenced by the price of an offset credit (Simpson et al., 2021).

In France, the mitigation hierarchy was introduced by the founding law for the Protection of Nature of 1976. The effectiveness of the mitigation hierarchy is measured via impact studies, which are required when obtaining a permit for development projects of a certain nature or size that are likely to affect protected species or habitats (Bigard et al., 2018; Levrel et al., 2018). It applies to projects,

plans and programs subject to environmental assessment as well as to projects subject to various administrative authorization procedures under the Environmental Code, such as environmental authorization, derogations for species protection or Natura 2000 impact assessment. The ARC sequence is widely practiced in European environmental policy, and EU Directives, such as the Habitats Directive, have been a major driver in the reinforcement of the ARC sequence in France (Quétier et al., 2014). France's Law 2010-788 of July 12, 2010 led to important reforms concerning the mitigation of development impacts on biodiversity, including reforms on the requirements for impact assessments and enforcement capabilities (Quétier et al., 2014). Governmental guidance from 2012 states that compensatory actions should last as long as impacts, but there is little guidance about design, duration, or frequency of monitoring efforts (Quétier et al., 2014). The 2016 Biodiversity Law resulted in compensation becoming mandatory for residual impacts (Levrel et al., 2018) and introduced the use of mitigation banks in the form of natural compensation sites to anticipate future compensation demands (Aubry et al., 2021). The idea behind natural compensation sites is to create a supply of compensation credits by restoring larger connected areas, in order to avoid the time lag between damaging habitats and restoring compensation areas. The first French natural compensation site materialized with the pilot project on the Cossure site in the Bouches-du-Rhône department. This site was created in 2009 by the organization CDC Biodiversité, with the support of the Ministry of Ecology and Sustainable Development (MEDD) (Dutoit et al., 2018).

An example of another mitigation bank policy in France that is essentially a privately funded PES is the agri-environmental biodiversity offset schemes (ABOS), where developers can obtain biodiversity offset credits by financing agri-environmental schemes (AES) contracts with farmers (Le Coent et al., 2017; Calvet et al., 2019). Since the developer is the one paying for the contract, an ABOS has an advantage compared to the publicly funded AES in that it does not incur the social cost of public funds. Indeed, ABOS have the same framework as AES, and technical practices are often identical for the same areas (Le Coent et al., 2017). Simpson et al. (2021) build an ecological-economic model of an ABOS where farmers create biodiversity offset credits that they sell to housing developers in order to see what effect of different biodiversity conservation targets will have on the price and quantity of offsets in the market. They look at eight policy scenarios, including one laissez-faire scenario, a no net loss target, and then six different net gain targets, ranging from a 5% to a 50% increase in the target biodiversity indicator. Applying their model to a case study in Scotland, they find that increasing the net gain target had ambiguous results and depended on the context. If offset demand is inelastic (elastic) at the no net loss equilibrium, then a net gain policy initially increases (decreases) the equilibrium offset quantity and price.

While the mitigation hierarchy and mitigation banking are widely used, there remains a lack of evidence of their efficacy (Brown and Land, 1999; Turner et al., 2001; Burgin, 2010; Needham et al., 2019; Damiens et al., 2021). As the oldest mitigation banking program, the US wetland mitigation bank is perhaps the most widely studied program and many argue that the US wetland mitigation banks have fallen short of their NNL objective, though the Army Corp of Engineers has failed to keep sufficient records to properly assess the objective (Burgin, 2010). Brown and Land (1999) found that only 74% of the individual banks met the NNL requirement by acreage, which does not even take into account the functional aspect of the compensation area. As with price-based PES, additionality is also an issue in mitigation banking. Burgin (2010) found that 75% of the wetland conservation banks likely would have been developed even without the legal requirement to mitigate lost wetland area. The literature generally focuses on either the feasibility of the compensation step or the lack of application of the avoidance step.

#### 2.4.1 Compensation feasibility

The final step in the mitigation hierarchy, compensation, requires restoring or creating new habitat areas to offset those lost due to a project. It is considered to be the most controversial step of the mitigation hierarchy (Arlidge et al., 2018). However, while the loss is certain, the gains from the compensation areas is uncertain (Weissgerber et al., 2019). Even when focusing on a particular ecosystem category, such as wetlands, there can be a wide variety of species and ecosystem functions such that it is difficult to find an identical or even comparable area for compensation actions. Moreover, criteria for finding an equivalent compensation markets (de Muelenaere, 2011). It is challenging and costly to calculate all the dimensions of biodiversity loss, and as such the areas traded via compensation banks tend to be measured based on simplistic measurements such as land area. For the most part, it seems that compensation areas used to generate offset credits do not measure up to the damaged and lost areas, resulting in a net loss when the size of the two areas is equal. From an ecological point of view, the results of various monitoring studies of the Cossure pilot project in France show that the goal of restoring herbaceous vegetation has been successful so far, and that its maintenance should be upheld thanks to the re-establishment of pastoral practices in the area (Dutoit et al., 2018). However, the final ecological assessment of the restoration actions in the Cossure project can only be carried out with a longer time span, as varying weather conditions can lead to different invasive species thriving, as happened in 2014 when a particularly rainy summer allowed an invasive species to proliferate (Dutoit et al., 2018). While the natural compensation site has demonstrated the possibility to restore some parts of a natural habitat, it has also demonstrated the limitations of compensation actions to fully restore lost ecosystems (Dutoit et al., 2018).

A study by Turner et al. (2001) found that roughly 80% of the wetlands built for mitigation purposes did not become fully functional. Similarly, Campbell et al. (2002) compare natural and created wetlands in the state of Pennsylvania, looking at variables related to soil and plant quality and found that even the oldest created wetlands had few similarities with their natural counterparts. Tillman et al. (2022)looked at wetland mitigation banks that have aged past the required 5-year management and monitoring periods and found that plant communities in wetland banks have greater conservation value than the lowest quality, degraded natural wetlands, but were not close to the same value as high-quality, reference natural wetlands. Reiss et al. (2009) studied wetland mitigation banks in Florida and found that while most banks were deemed successful in terms of permit criteria, the permit criteria were not explicitly tied to ecological criteria, and so the functional performance provided by the wetland banks remains unclear. While the scientific knowledge base on ecological restoration is still relatively young (Pullin and Knight, 2009; Suding, 2011), there is evidence that restoring degraded habitats is more likely to be effective than creating new habitat because it is more likely to re-establish the necessary ecosystem functions (Mitsch and Wilson, 1996; Kozich and Halvorsen, 2012; Moreno-Mateos et al., 2012).

Location is crucial when discussing the compensation site. A main drawback of the compensation step is the inevitable geographic relocation of habitat area (Brown and Land, 1999). In particular, lower land prices in rural areas can lead to a displacement of wetlands or other natural areas from more urban to more rural areas. Quétier et al. (2014) argue that guidance on offset location is essential to avoid targeting the least expensive land without considering the potential effectiveness of an offset site. Grouping offsets for multiple projects into one compensation site is more likely

to achieve the NNL objective, but also comes with risks. For example, if the same restoration action is used across the entire site and fails in one part of the site it is more likely to fail in the other parts as well (Moilanen et al., 2009). In the US, the 2008 Mitigation Rule codified the idea of service areas—that is, the geographic areas for which a mitigation bank can provide compensation (Martin and Brumbaugh, 2013). This, of course, then affects the decision of which sites to restore as some may have a more limited service area. Finally, the location of the impacted site also has consequences, as there can be cumulative impacts from multiple projects impacting the same site (Kiesecker et al., 2010).

The timing of compensation is another factor to consider, especially since restoring areas to a desired quality can take time, leading to a lag between damages and offsets (Burgin, 2010; de Muelenaere, 2011; Damiens et al., 2021). Additionally, long-term retention of biodiversity on compensation sites is crucial to reaching the NNL objective (Damiens et al., 2021). This can be complicated by institutions that are subject to shorter-term policy and election cycles, though some legal instruments, such as the use of conservation easements, have been proposed as a way to protect biodiversity gains on compensation sites from incompatible land uses over time (Damiens et al., 2021). Another way to account for losses during time lags is to use trading ratios when calculating the number of offsets required to compensate an impacted area (Needham et al., 2019; Damiens et al., 2021). For example, when there is a long time lag between the negative impact and the compensation, a developer may be required to purchase more offsets than if there were no time lag.

## 2.4.2 A lack of avoidance

While the compensation step of the sequence has arguably received the most attention in the literature, other studies regarding the mitigation hierarchy also highlight that the first step, avoidance, is the most important but is "more often ignored than implemented" (Clare et al., 2011). Indeed, avoidance is the most certain and effective way to limit impacts on biodiversity, as it does not engender the same problems as compensation, such as restoration time lags, limitations to what can be offset, and negative social implications from taking away biodiversity in one area and improving it in another (Phalan et al., 2018). A few papers describe different reasons for which the avoidance step is not properly implemented. Clare et al. (2011) identify five key factors that lead decision-makers to fail to prioritize wetland impact avoidance and reduction above compensation in the US and Canada, namely a lack of consensus on what constitutes avoidance, a failure of land-use planning approaches to identify high-priority wetlands in advance of development, an economic undervaluation of wetlands, a 'techno-arrogance' associated with wetland creation and restoration that results in wetland loss, and finally inadequate enforcement of compensation requirements. Similarly, Phalan et al. (2018) identify five challenges for effective impact avoidance: political will, legislation quality and its implementation in practice, process, capacity (informational and transaction costs), and technical knowledge. Finally, Levrel et al. (2018) identify some obstacles that undermine the additionality of the ARC sequence in France, which relate to a diversion of resources from existing conservation actions toward compensation measures and the pursuit of rents and cost minimization by different stakeholders.

Bigard et al. (2018) sought to evaluate how the execution of the ARC sequence in France aligned with the definitions and national guidance for each step. They analyzed 42 impact studies for projects between 2006 and 2016 in the territory of the Montpellier metropolis and contiguous municipalities and found that in 60% of the cases, the qualifications of the ARC measures given in the impact study did not correspond to the national reference definitions. For example, the so-called avoidance measures in the impact studies were actually reduction measures according to the national reference definitions. They also found that this confusion had negative consequences on the ecological effectiveness of the ARC hierarchy. As Stevenson and Weber (2020) note, there is a temptation to skip to steps lower in the hierarchy that are easier or cheaper.

## 2.5 Conclusion

Biodiversity loss is one of the most important problems humanity faces today. The various benefits humans receive from a diversity of life and services supported by diverse ecosystems is not captured by markets, and so government intervention is necessary to reach a socially optimal provision. This literature review has focused on Payments for Environmental Services (PES), a policy tool that aims to internalize the positive externalities of environmental services to encourage their provision. The economic literature analyzes conditions and design choices that make PES more or less efficient, using both theoretical models and empirical studies of existing programs.

PES can take many forms, including auctions, programs with agglomeration bonuses, and mitigation banks, all of which face challenges and are imperfect solutions to a complex problem. A variety of ecosystems and policy contexts necessitates a variety of policy design options. Conservation auctions may perform best in one context, while standard fixed rate PES may be preferred in another. Therefore, policy makers should be prudent when designing PES schemes and gather information about their specific policy context. While this literature review has focused on efficiency, policy makers should also pay attention to equity and fairness concerns.

Although the literature on PES is extensive, knowledge gaps remain. In particular, the theoretical design of PES under imperfect competition has yet to be explored. Additionally, very few papers look at the interaction of PES and other environmental policies such as environmental taxes. Next, the social cost of public funds is largely excluded from analyses of PES programs. Finally, the literature on mitigation banking lacks an economic analysis of the rationality of the ARC sequence. The remainder of this thesis aims to contribute to the existing literature by filling in these gaps.

# $\begin{bmatrix} & & \\ &$

## Payment for Environmental Services and environmental tax under imperfect competition

This paper designs the second-best Payment for Environmental Services (PES) when it interacts with a Pigouvian tax under imperfect competition. We consider farmers who face a choice between producing a conventional or an organic agriculture good. The regulator sets a Pigouvian tax on conventional agriculture as it generates environmental damages, as well as a PES on uncultivated land as buffer strips favor biodiversity. The conventional agriculture sector is perfectly competitive, unlike the organic agriculture sector, which is organized under an oligopoly. We show that the second-best level of the Pigouvian tax is higher than the marginal damage whereas the PES is lower than the marginal benefit. We then introduce the marginal social cost of public funds (MCF) and show that the Pigouvian tax increases with the MCF while the PES decreases with the MCF provided that demand for the conventional agriculture good is inelastic. We thus highlight a contributory component of the environmental incentive tax. This paper also identifies specific cases where the PES is ineffective in promoting biodiversity.<sup>1</sup>

**Keywords**: Biodiversity Conservation  $\cdot$  Payment for Environmental Services  $\cdot$  Pigouvian Tax  $\cdot$  the Marginal Social Cost of Public Funds  $\cdot$  Market Power

<sup>&</sup>lt;sup>1</sup>This chapter is joint work with Sonia Schwartz: Krautkraemer A., S. Schwartz (2023) Payment for environmental services and environmental tax under imperfect competition, document de recherche du LEO, DR LEO 2023-12, available at: https://hal.science/hal-04160411

## 3.1 Introduction

The collapse of biodiversity is a well-documented phenomenon, which is likely to worsen with climate change (Dasgupta, 2021; Díaz et al., 2019; Ruckelshaus et al., 2020). A leading cause of the decline in biodiversity is the loss of various habitats due to land use change (Lewis et al., 2011; Bamière et al., 2013). According to Dasgupta (2021), an estimated 20% of species could become extinct in the next several decades, perhaps twice as many by the end of the century. A way to take into account the many and varied benefits that humans derive from the natural environment and healthy ecosystems is to mobilize the concept of ecosystem services (Reid et al., 2005). According to Reid et al. (2005), they are categorized into the following four types: provisioning services such as food, water, timber, and fiber; regulating services that affect climate, floods, disease, wastes, and water quality; cultural services such as soil formation, photosynthesis, and nutrient cycling.

The economic literature distinguishes between ecosystem services and environmental services. While ecosystem services refer to the functioning of ecosystems, environmental services (ES) refer to the notion of externalities induced by human activities. In this case, mechanisms for internalizing externalities must be implemented, to encourage their optimal provision. Payments for environmental services (PES) are becoming a familiar tool for conserving and restoring ecosystems and the services they provide. They aim to finance the conservation and restoration of nature (Dasgupta, 2021).

One of the most widely cited definitions of PES comes from Wunder (2005). He defines PES as a voluntary transaction where a well-defined ES or a land-use that is likely to produce that service is bought by a (minimum one) ES buyer from a (minimum one) ES provider if and only if the ES provider secures ES provision. Other definitions as given by Muradian et al. (2010) include the possibility of in-kind payment. Wunder's definition is broad enough to include in particular a Coasean negotiation or a public buyer. For instance, if the PES involves private agents, this type of PES can be related to Coasean negotiations<sup>2</sup>. Others PES include certain types of government intervention that reflect a Pigouvian subsidy (Sattler and Matzdorf, 2013; Pigou, 1920). This type of PES is common in practice and is

<sup>&</sup>lt;sup>2</sup>One example of a Coasean PES is the Vittel PES in north-eastern France, where Nestle reached an agreement with local farmers to prevent nitrate contamination in aquifers (Sattler and Matzdorf, 2013; Bingham, 2021).

the focus of this article.

Some examples of PES funded by public authorities can be mentioned. Following France's National Biodiversity Plan of 2018 (MTES, 2021), French water agencies are experimenting with their own PES schemes. They have been allocated 150 million euros of the French national budget, with the objective to maintain or create good ecological practices, such as lowering pesticide use or planting cover crops (MTES, 2019). While both maintaining and creating good practices will be remunerated, creating good practices will receive much higher compensation (up to 676 euros/ha/year compared to up to 66 euros/ha/year for maintenance). A new program in Paris involves setting up a PES between the water agency Eau de Paris and farmers located in the water catchment area (CPES, 2020). Farmers will benefit from the PES if they commit to limiting the use of fertilizers and pesticides, or if they establish grasslands, which are considered a better filter for water than wheat or maize fields. A wheat farmer who converts to organic farming will be able to get  $\in$ 450 per year per hectare for the first five years and  $\in$ 220 for the next two years. Farmers will only receive the full payment if a target level for nitrate concentration in groundwater is reached. This PES was created thanks to the validation of a state aid scheme (n°SA.54810) by the European Commission. This type of PES is therefore considered as a public subsidy. Regarding the Pigouvian subsidy promoting positive externalities, it should correspond to the marginal environmental benefit.

The European agri-environmental programs are financed through public funds under the Common Agricultural Policy (CAP). They are one of the major advances in the CAP in recent years and are considered as PES programs. These agrienvironmental measures consist of offering financial compensation to farmers for voluntary commitment on their part, over several years, to implement practices or production. For example, an agri-environmental contract may cover a fiveyear period and compensate foregone profits or the costs of implementing the environmental measures. Common management practices adopted under agrienvironmental measures include reducing fertilizer and/or pesticide use, planting buffer crops near rivers, and adaptations to crop rotations. Indeed, long crop rotations improve ecosystem services such as support services through improved soil quality. The diversity of productive activities on a farm promotes beneficial interactions between crops and livestock and the management of landscape features such as grass strips, embankments, hedges or watercourses contribute to the ecological functioning of agroecosystems (Beesley and Ramsey, 2009; Wätzold et al., 2016; Princé and Jiguet, 2013).

While changing certain agricultural practices can help protect biodiversity, agricultural practices can also cause pollution. We can cite the use of chemical fertilizers and pesticides that pollute watersheds (Shortle and Abler, 2001). In order to internalize negative externalities, the regulator can set a Pigouvian tax (Pigou, 1920), equal to the marginal damage in a perfectly competitive market setting. However, this result is obtained by considering only the negative externality. In order to take into account the specificities of the agricultural domain, it would be necessary to consider a model that takes into account both positive and negative externalities.

Some work has looked at the interaction of different public policies (Howlett and Rayner, 2013). According to Bryan and Crossman (2013), interaction effects of multiple financial incentives may reduce policy efficiency wherever multiple incentives encourage the supply of services from agro-ecosystems. Agri-environmental measures must, however, take into account that policies are typically bundles of different policy tools arranged in policy mixes and that financial incentives for different ecosystem services interact (Huber et al., 2017). Lankoski and Ollikainen (2003) provide a framework for a theoretical analysis of several environmental policies in the agricultural sector. Assuming parcels of varying land quality, the authors study the optimal land allocation between two crops that are more or less intensive in fertilizer use and fallow buffer strips when facing negative externalities from nutrient runoff, and positive externalities from biodiversity and landscape diversity. They defined first-best environmental policies, involving a differentiated tax on fertilizer and a differentiated buffer strip subsidy. The crucial assumptions supporting these results are notably the absence of a distortion resulting from the contributory taxes and the absence of any market power in the agricultural sector.

The PES is based on the beneficiary pays principle. Its implementation requires raising public funds, which can cause economic distortions (Mirrlees, 1971). Increasing contributory taxes can change the allocation of resources in an economy through impacts on consumption, labor, and investment decisions (Dahlby, 2008). A simple way to take into account these distortions is to consider the marginal social cost of public funds (MCF). It is a measure of the welfare loss to society as a result of raising additional revenues to finance government spending (Browning, 1976; Dahlby, 2008). For example, Browning (1976) estimates the MCF of labor income taxes in the United States, finding a MCF of \$1.09-\$1.16 per dollar tax revenue raised. According to Beaud (2008), this cost is equal to 1.2 for France. So, when the regulator raises one euro in taxes, it costs the society 1.2 euros. This aspect should therefore not be neglected in the decision to set up a PES.

The relevance of competitive organic farming market can be questioned. According to Nguyen-Van et al. (2021), the development of organic agriculture can be very heterogeneous over a territory. For instance, out of 34259 municipalities in metropolitan France (excluding overseas territories) with at least one farmer, only 418 (1.2%) are 100% organic, and 52.4% of municipalities do not have an organic farmer.<sup>3</sup> The non-uniform distribution of organic farming across the country and transport constraints for organic products can limit competition in organic product market, resulting in local markets for organic farming where some producers have market power. Indeed, while we did not find empirical work on the presence of market power in the organic sector, as Sexton (2013) points out, in reality probably no agricultural markets are perfect examples of competitive markets, for reasons including the emphasis on dimensions of product quality and differentiation, which could apply to organic products. Thus, it can be interesting to see what happens theoretically to the welfare-maximizing PES payment in the presence of market power.

The economic literature has analyzed the effectiveness of agri-environmental programs in protecting biodiversity (Pe'er et al., 2014; Kleijn and Sutherland, 2003; Batáry et al., 2015) but neglected the possible interactions between the different environmental policies. On the one hand, if public PES are similar to Pigouvian subsidies, their analysis does not have to mirror Pigouvian taxes because they imply a necessary financing constraint. On the other hand, the economic literature is not well-developed concerning the design of a PES under imperfect competition, contrary to the Pigouvian tax. Indeed, a tax based only on marginal external damages ignores the social cost of further output contraction by a producer whose output already is below an optimal level. Under market power, the optimal second-best tax should actually be less than the marginal damage (Barnett, 1980; Ebert, 1991). Since then, the literature on environmental taxation has widely developed for numerous scenarios of imperfect competition.

The focus of our theoretical paper is to analyze the second-best PES design combined with environmental taxes under imperfect competition and taking into account distortions from contributory taxation. To do this, we assume a farmer chooses to produce a conventional or an organic good. Whereas the conventional agriculture

<sup>&</sup>lt;sup>3</sup>Spatial factors explain the gaps in organic development between territories, such as the quality of the soil (Wollni and Andersson, 2014; Lampach et al., 2020) as well as the geographical organisation of the activity and populations (Ben Arfa et al., 2009) and the presence of many other organic farmers in a geographical unit (Schmidtner et al., 2012; Bjørkhaug and Blekesaune, 2013).

good market is perfectly competitive, the organic good market is organized under an oligopoly. Farmers can produce conventional agriculture goods, which causes environmental damages, or an organic production which we are assuming will have a neutral impact on the environment. If the farmer leaves fallow buffer strips, this favors biodiversity. In order to simultaneously favor biodiversity and reduce environmental damages, the regulator sets a PES on fallow land and a Pigouvian tax on conventional agriculture production.

In our framework, the Pigouvian tax decreases the conventional good production level. The PES on the fallow land area reduces organic and conventional production levels. Under market power, we show that the second-best level of the Pigouvian tax is higher than the marginal damage - contrary to Barnett (1980) - and the PES is lower than the marginal benefit. The organic good production level is too low because of the market power and the PES further reduces this production level. In order to mitigate the reduction due to market power, the regulator sets a PES lower than the marginal benefit. The conventional good level is reduced with both the PES and the Pigouvian tax. As the PES is not high enough, the regulator sets a Pigouvian tax above the marginal damage in order to reach the correct level of conventional agriculture. The environmental policies are used in a complementary way to take into account the distortion induced by the market power. We also analyze the particular case where farmers never choose buffer strips, which occurs when productions are profitable enough. In this case the PES is useless and the regulator can only regulate environmental damages. This time, market power in organic agriculture favors conventional agriculture production. So a way to reduce environmental damages is to set the Pigouvian tax above the marginal damage.

We then consider distortionary taxation in our economy. To do that, we introduce a marginal social cost of public funds (MCF). We show that the environmental tax increases with the MCF, whereas the PES decrease with the MCF under two assumptions: the demand for the conventional agriculture good is inelastic and environmental tools have to provide buffer strips efficiently. We thus highlight a *contributory component* of the environmental incentive tax under distortionary taxation. Indeed, the primary objective of a Pigouvian tax is to give the appropriate incentives to agents and not to raise a revenue for the regulator.

The assumption of a neutral impact of organic agriculture can be controversial. On the one hand, several empirical studies found a positive relationship between organic farming and biodiversity (Batáry et al., 2015; Freemark and Kirk, 2001; Marja et al., 2014; Hole et al., 2005). On the other hand, other studies found no or minor difference between conventional and organic farming (Hiron et al., 2013; Piha et al., 2007; Purtauf et al., 2005; Gerling et al., 2019) and in some cases conventional farming even supported a greater biodiversity than organic farming (Weibull et al., 2003; Rahmann, 2011). The reasons for these contradicting results are diverse. Fuller et al. (2005) found that some species benefit from organic farming, while others benefit from conventional farming. Tscharntke et al. (2021) highlight that what characterizes organic agriculture is the prohibition of synthetic agrochemicals, which results in limited benefits for biodiversity. Seriously estimating the impact of organic farming on biodiversity requires a well-defined benchmark. For example, according to Dasgupta (2021), one of the causes of biodiversity loss is the change in land use, especially conversion to agricultural use. In this case, organic agriculture would be considered always detrimental to biodiversity. If it had been assumed in this paper that organic farming also produces biodiversity, the effects of the single PES on grass strips and on organic farming would have been cancelled out making the PES still useless.

This paper is organized as follows. Section 3.2 presents our model. Section 3.3 examines second-best environmental policies and Section 3.4 introduces the MCF. Finally, Section 3.5 concludes.

## 3.2 The model

In this section, we present the assumptions used in our model, the farmers' production decision absent any policy and the first-best allocation.

## 3.2.1 Assumptions

We consider  $n \ge 2$  identical farmers who each have three choices for how to manage his land: conventional agriculture  $(x_{1i})$ , organic agriculture  $(x_{2i})$ , and/or leaving the land uncultivated to act as a reserve for biodiversity  $(y_i)$ . Each farmer *i* produces  $x_{1i}$ ,  $x_{2i}$  and  $y_i$ , with total output for each good equal to  $X_1 = \sum_{i=1}^n x_{1i}$ ,  $X_2 = \sum_{i=1}^n x_{2i}$ , and  $Y = \sum_{i=1}^n y_i$ , respectively. Each farmer decides how much of his land to allocate to each management option such that  $x_{1i} + x_{2i} + y_i = T_i$  where  $T_i$  is his total area of land (with  $T = \sum_{i=1}^n T_i$ ). We assume that producing  $x_{1i}$  ( $x_{2i}$ ) units requires  $x_{1i}$ ( $x_{2i}$ ) units of land  $\forall i = 1, ..., n$ .

The cost of implementing organic agriculture is higher than that of conventional

agriculture,  $c_1(x_{1i}) < c_2(x_{2i})$ . Both  $c_1(x_{1i})$  and  $c_2(x_{2i})$  are increasing and convex<sup>4</sup>,  $\forall i = 1, ..., n$ . The quantity of land left uncultivated only incurs an opportunity cost of not producing. For simplicity, we set the cost of entry into the organic market at zero, which corresponds to an absence of barriers to entry<sup>5</sup>. We assume a linear demand for both agricultural goods. The inverse demand function for each agricultural product is given by  $p_1(X_1)$  and  $p_2(X_2)$  for conventional and organic agriculture, respectively.

The organic agricultural good can be considered as a good with few substitutes, contrary to the conventional agriculture good. For example, transport constraints for organic products can limit competition in the organic product market. We are assuming that the conventional good is one that is traded on the global market (e.g. wheat) while the organic good is one that is traded locally, is difficult to transport long distances, and is not taking prices from the global market (e.g. baby leaf lettuce). So, we assume perfect competition on the conventional agriculture good market and imperfect competition on the organic agriculture good market, which is organized in the form of oligopoly.

Each of the land management choices has a different impact on the environment. Conventional agriculture causes pollution, represented by the damage function  $D(X_1)$ which is increasing and convex,  $D'(X_1) > 0$ ,  $D''(X_1) > 0$ . We assume that organic agriculture has a neutral impact on the environment. Finally, the uncultivated land leads to biodiversity benefits, and has a positive impact on the environment, represented by the increasing and concave benefit function, given by B(Y).

## 3.2.2 The benchmark

In this subsection we look at the farmer's decision in the absence of any policy. He behaves as a price taker on the conventional product market and as a Cournot competitor on the organic product market. Farmer *i* maximizes his profit by choosing  $x_{1i}$  and  $x_{2i}$  and by considering the physical constraint of his available land:  $T_i$  has to be greater than or equal to  $x_{1i} + x_{2i}$ . Associating  $\lambda$  to this constraint, the profit for farmer  $i \forall i = 1, 2, ..., n$  with  $i \neq j$  is:

$$\pi_i(x_{1i}, x_{2i}) = p_1 x_{1i} + p_2 (X_2) x_{2i} - c_1(x_{1i}) - c_2(x_{2i}) + \lambda (T_i - x_{1i} - x_{2i})$$

<sup>4</sup>Additionally, we assume that  $c_{1}'''(x_{1i}) = 0$  and  $c_{2}'''(x_{2i}) = 0, \forall i = 1, ..., n$ .

<sup>&</sup>lt;sup>5</sup>In reality, there are requirements for farmers producing conventional agriculture to make a transition to organic agriculture. For simplicity, we assume the corresponding costs equal zero

Maximizing profit yields the following conditions:

$$p_1 - c_1'(x_{1i}) - \lambda = 0 \tag{3.1}$$

$$p_2'(X_2)x_{2i} + p_2(X_2) - c_2'(x_{2i}) - \lambda = 0$$
(3.2)

$$\lambda(T_i - x_{1i} - x_{2i}) = 0 \tag{3.3}$$

Whereas a farmer equalizes the marginal cost to the price when making his conventional agriculture production decision, he considers the marginal revenue when making his organic agriculture production decision. The production decision depends on whether the land constrains the farmer's decision, that is  $\lambda > 0$ , or whether the farmer will have some uncultivated land, that is  $\lambda = 0$ .

Farmer *i* considers all other farmers' decisions in the organic product market in order to maximize his profit. To see how the production level of farmer *i* responds to the production level of farmer *j*, we use Equation (3.2) and apply the implicit function theorem. When the farmer leaves uncultivated land ( $\lambda = 0$ ), we find:

$$\frac{\partial x_{2i}}{\partial x_{2j}} = -\frac{p_2''(X_2)x_{2i} + p_2'(X_2)}{p_2''(X_2)x_{2i} + 2p_2'(X_2) - c_2''(x_{2i})} < 0$$

An increase in farmer j's production of the organic agriculture good will make farmer i reduce his production of the organic agriculture good. Thus, goods produced from organic agriculture are strategic substitutes.

For the case where there is no uncultivated land  $(\lambda > 0)$ , using equations (3.1), (3.2) and (3.3), we set  $N(x_{2i}, x_{2j}) = p_1 - c'_1(T_i - x_{2i}) - p'_2(X_2)x_{2i} - p_2(X_2) + c'_2(x_{2i})$ .

Applying the implicit function theorem we find:

$$\frac{\partial x_{2i}}{\partial x_{2j}} = -\frac{\frac{\partial N}{\partial x_{2j}}}{\frac{\partial N}{\partial x_{2i}}} = -\frac{-p_2''(X_2)x_{2i} - p_2'(X_2)}{c_1''(T_i - x_{2i}) - p_2''(X_2)x_{2i} - 2p_2'(X_2) + c_2''(x_{2i})} < 0$$

This shows that an increase in farmer j's production of the organic agricultural good will lead to a decrease in farmer i's production of the organic agricultural good. Organic agricultural goods are once again in this case, strategic substitutes.

Farmers make their decisions without taking into account environmental aspects – such as environmental damage and benefits – and in a world of imperfect competition. As a result, these production levels are not optimal. There is room for public intervention.

## 3.2.3 The first-best

In this subsection, we investigate the first-best outcome. The government regulator seeks to maximize welfare, which is composed of the consumer surplus, the farmer's profit, the environmental damage and benefit while taking into account the constraint on available land:

$$W_{X_1,X_2,\lambda} = \int_0^{X_1} p_1(u) du + \int_0^{X_2} p_2(v) dv - nc_1\left(\frac{X_1}{n}\right) - nc_2\left(\frac{X_2}{n}\right) + B(T - X_1 - X_2) - D(X_1) + \lambda(T - X_1 - X_2)$$

Maximizing welfare gives the conditions for the first best optimal solutions,  $x_1^*$  and  $x_2^*$ :

$$p_1(X_1^*) - c_1'(\frac{X_1^*}{n}) - B_y - D'(\frac{X_1^*}{n}) - \lambda = 0$$
(3.4)

$$p_2(\frac{X_2^*}{n}) - c_2'(\frac{X_2^*}{n}) - B_y - \lambda = 0$$
(3.5)

$$\lambda(T - X_1^* - X_2^*) = 0 \tag{3.6}$$

There are two possible cases for  $\lambda$ . If  $\lambda = 0$ , then  $y = T - X_1^* - X_2^*$  will either be zero or positive. If  $\lambda > 0$ , then  $y = T - X_1^* - X_2^*$  will be zero, and no land will be left uncultivated. Taking into account the Kuhn-Tucker multiplier, the conventional and organic production levels are based on marginal social costs, i.e. marginal cost of production of each agriculture type, as well as biodiversity benefits from fallow land and the pollution damages from conventional agriculture.

Consider a first-best policy. In this case, the pigouvian tax must be equal to the marginal damage i.e.  $t^* = D'(\frac{X_1^*}{n})$  and the PES must be set at the marginal environmental benefit level i.e.  $s^* = B_y$  if  $\lambda > 0$  and must be nonexistent if  $\lambda = 0$ .

However, it is easy to see that these levels of environmental policy would not lead to the first-best (equations 3.4, 3.5, 3.6). Environmental policies would correct environmental externalities, but another distortion would remain: market power. The regulator must therefore designate second-best policies.

## 3.3 Second-best environmental policies

Although we cannot directly correct for market power, we can examine a secondbest environmental policy to internalize the negative and positive externalities of pollution and biodiversity, respectively, and improve welfare. Here, we examine an environmental tax, t, on pollution related to the conventional agriculture good and a PES for biodiversity, s, which subsidizes uncultivated land in order to favor biodiversity. We first look at the farmer's behavior facing environmental policies and we then define the second best level of environmental tax and PES.

## 3.3.1 The farmer's behavior

We now introduce into the farmer's profit the environmental tax and the PES. The profit for farmer i,  $\forall i$  and  $i \neq j$ , taking into account the constraint on his land is now:

$$\pi_i = p_1 x_{1i} + p_2 (X_2) x_{2i} - c_1(x_{1i}) - c_2(x_{2i}) - t x_{1i} + s(T_i - x_{1i} - x_{2i}) + \lambda (T_i - x_{1i} - x_{2i})$$

Maximizing profit yields the following conditions:

$$p_1 - c_1'(x_{1i}) - t - s - \lambda = 0 \tag{3.7}$$

$$p_2'(X_2)x_{2i} + p_2(X_2) - c_2'(x_{2i}) - s - \lambda = 0$$
(3.8)

$$\lambda(T_i - x_{1i} - x_{2i}) = 0 \tag{3.9}$$

We can see how production levels change with environmental policies. When the farmer leaves uncultivated land ( $\lambda = 0$ ), we use Equations (3.7) and (3.8), and apply

the implicit function theorem. We find:

$$\frac{\partial x_{1i}}{\partial s} = \frac{1}{-c_1''(x_{1i})} < 0$$
$$\frac{\partial x_{1i}}{\partial t} = \frac{1}{-c_1''(x_{1i})} < 0$$
$$\frac{dx_{2i}}{ds} = \frac{1}{2p_2'(X_2) + p_2''(X_2)x_{2i} - c_2''(x_{2i})} < 0$$

The PES decreases production levels of both agriculture goods while the environmental tax only decreases the production level of the conventional agriculture good. Thus, the PES and the environmental tax lead to an increase in uncultivated land and consequently favor biodiversity benefits.

If the farmer leaves no uncultivated land  $(\lambda > 0)$ , we obtain, after using Equations (3.7), (3.8) and (3.9) and applying the implicit function theorem:

$$\frac{\partial x_{1i}}{\partial t} = \frac{1}{c_1''(x_{1i}) + c_2''(T_i - x_{1i}) - 2p_2'(T - X_1) - p_2''(T - X_1)(T_i - x_{1i}) - p_1'} < 0$$

Since  $x_{2i} = T_i - x_{1i}(t)$ , it is obvious that:

$$\frac{dx_{2i}}{dt} = -\frac{\partial x_{1i}}{\partial t} > 0$$

This implies that an increase in the environmental tax will lead to an increase in the production level of the organic agriculture good and a decrease in the production level of the conventional agriculture good, and in the same proportion. It is a zero-sum game. Here, the PES does not impact the farmer's production choices because the cost structure and market is such that it is not profitable to leave any land uncultivated. Hence, a PES is useless.

## 3.3.2 Second-best level of environmental tax and PES

We maximize the social welfare function to find the second-best levels of the environmental tax and of the PES. We first investigate the case where it is optimal to leave uncultivated land, and then when it is optimal to cultivate all the land. Starting with the first scenario ( $\lambda = 0$ ), the social welfare function is:

$$W(X_1(s,t), X_2(s)) = \int_{0}^{X_1(s,t)} p_1(u) du + \int_{0}^{X_2(s)} p_2(v) dv - nc_1\left(\frac{X_1(s,t)}{n}\right) - nc_2\left(\frac{X_2(s)}{n}\right) + B(T - X_1(s,t) - X_2(s)) - D(X_1(s,t))$$

Maximizing this welfare function with respect to s and t leads to the following first order conditions:

$$\frac{\partial X_1}{\partial s} [p_1(X_1(s)) - c_1' \left(\frac{X_1(s)}{n}\right) - B_y - D'(X_1(s))] + \frac{\partial X_2}{\partial s} [p_2(X_2(s)) - c_2' \left(\frac{X_2(s)}{n}\right) - B_y] = 0$$

$$\frac{\partial X_1}{\partial t} [p_1(X_1(t)) - c_1' \left(\frac{X_1(t)}{n}\right) - B_y - D'(X_1(t))] = 0$$
(3.11)

with  $\frac{\partial X_1}{\partial s} < 0$ ,  $\frac{\partial X_1}{\partial t} < 0$ , and  $\frac{dX_2}{ds} < 0$  obtained in the previous section and  $B_y = B'(y)$ . Using equations (3.7) and (3.8), we rearrange the profit maximization conditions to obtain the following:

$$p_1 - c_1'\left(\frac{X_1}{n}\right) = t + s$$
 (3.12)

$$p_2(X_2) - c_2'\left(\frac{X_2}{n}\right) = -p_2'(X_2)\frac{X_2}{n} + s \tag{3.13}$$

Next, we plug equations (3.12) and (3.13) into equations (3.10) and (3.11) to obtain the following equations:

$$\frac{\partial X_1}{\partial s}[t+s-B_y-D'(X_1(s))] + \frac{dX_2}{ds}[-p_2'(X_2(s))\frac{X_2}{n}+s-B_y] = 0$$
(3.14)

$$\frac{\partial X_1}{\partial t} [t + s - B_y - D'(X_1(t))] = 0$$
(3.15)

We can now solve (3.15) for t, and plug that into (3.14) to solve for s and t. We find:

$$s = B_y + p'_2(X_2) \frac{X_2}{n}$$
(3.16)

$$t = D'(X_1) - p'_2(X_2)\frac{X_2}{n}$$
(3.17)

Here, the PES is based on the marginal environmental benefit from the grass strip rather than the opportunity cost for the farmer, in contrast to typical agrienvironmental contracts. It appears that the second-best PES is lower than the marginal benefit, whereas the second-best tax is higher than the marginal damage. This result differs from Barnett (1980) who shows that in the presence of market power, the Pigouvian tax must be lower than the marginal damage. In our study, production of the organic agriculture good is lower than its first best level because of market power. As the PES reduces the level of organic agriculture, a way to not further decrease this level is to set a lower PES. But this PES will not sufficiently reduce the production from conventional agriculture. Thus the environmental tax is higher than its first best level in order to get the right level of conventional agriculture. Replacing the value of environmental policy tools in (3.7) and (3.8), we find first-best quantities (given by (3.4) and (3.5)). Finally, we can see that if the number of firms increases and approaches infinity, both environmental policy tools reach their first best level: the marginal benefit for the PES and the marginal damage for the environmental tax.

We now investigate the second-best environmental policy tool level when it is profitable to leave no uncultivated land ( $\lambda > 0$ ). So setting  $X_1 = T - X_2$ , the social welfare is now:

$$W(X_1(t), X_2(t)) = \int_0^{T-X_2(t)} p_1(u) du + \int_0^{X_2(t)} p_2(v) dv - nc_1\left(\frac{T-X_2(t)}{n}\right) - nc_2\left(\frac{X_2(t)}{n}\right) + B\left(T - (T-X_2(t)) - X_2(t)\right) - D(T-X_2(t))$$

Maximizing this welfare equation yields the following first order condition:

$$\frac{dX_2}{dt}\left[-p_1(T-X_2) + p_2(X_2) + c_1'\left(\frac{T-X_2}{n}\right) - c_2'\left(\frac{X_2}{n}\right) + D'(T-X_2)\right] = 0 \quad (3.18)$$

Using the profit first order conditions (3.7) and (3.8), we find that:

$$-p_1 + c_1'\left(\frac{T - X_2}{n}\right) + p_2(X_2) - c_2'\left(\frac{X_2}{n}\right) = -t - p_2'(X_2)\frac{X_2}{n}$$
(3.19)

Plugging (3.19) into (3.18) yields:

$$t = D'(T - X_2) - p'_2(X_2)\frac{X_2}{n}$$
(3.20)

In this case, the second-best environmental tax level is also higher than the marginal damage. As the PES cannot incentivize the uncultivated land, only the environmental tax will correct both the negative externality and market power in the organic market. Again, this second-best environmental tax can achieve the first-best levels of production. Our results are summed up in the following proposition:

**Proposition 1.** The second-best PES is lower than the marginal benefit, whereas the Pigouvian tax is higher than the marginal damage contrary to Barnett (1980). There are cases where PES are ineffective in protecting biodiversity.

## **3.4** The marginal social cost of public funds

The public PES needs to be financed, which means taxing taxpayers in other ways. There are two ways to introduce the distortions induced by the tax system into our theoretical model. The first is to consider a general equilibrium model that explicitly introduces the tax system. The problem is the complexity of the model, making its results difficult to interpret. The second is to introduce into a partial equilibrium model the marginal social cost of public funds (MCF), which summarizes the fiscal distortions. In order to enrich our results, we choose this second path. We denote by  $\epsilon$  the MCF. Each euro raised by the environmental tax will enable to reduce distortionary contributory taxes of  $(1 + \epsilon)$  euros. Conversely, implementing a PES means a requirement for additional government revenue through increased contributory taxes, which will come at a cost to society. So, each euro allocated to the PES costs  $(1 + \epsilon)$  euros to society.<sup>6</sup> We modify the welfare function given in Section 3.3 in order to take into account the taxation effects. In the case where the

<sup>&</sup>lt;sup>6</sup>The model was extended by introducing a constraint to finance the PES by the environmental tax. However, the results were not tractable.

farmers leave uncultivated land, the welfare reads as:

$$W(X_1(s,t), X_2(s)) = \int_0^{X_1(s,t)} p_1(u) du + \int_0^{X_2(s)} p_2(v) dv - nc_1\left(\frac{X_1(s,t)}{n}\right) - nc_2\left(\frac{X_2(s)}{n}\right) + B(T - X_1(s,t) - X_2(s)) - D(X_1(s,t)) + \epsilon t X_1(s,t) - \epsilon s(T - X_1(s,t) - X_2(s))$$

Maximizing this welfare function with respect to s and t leads to the following first order conditions:

$$\frac{\partial X_1}{\partial s} [p_1(X_1(s)) - c_1'\left(\frac{X_1(s)}{n}\right) - B_y - D'(X_1(s)) + \epsilon t + \epsilon s] + \frac{dX_2}{ds} [p_2(X_2(s)) - c_2'\left(\frac{X_2(s)}{n}\right) - B_y + \epsilon s] - \epsilon (T - X_1(s) - X_2(s)) = 0$$
(3.21)

$$\frac{\partial X_1(t,s)}{\partial t} [p_1(X_1(t,s)) - c_1' \left(\frac{X_1(t,s)}{n}\right) - B_y - D'(X_1(t,s)) + \epsilon t + \epsilon s] + \epsilon X_1(t,s) = 0$$
(3.22)

with  $\frac{\partial X_1(t,s)}{\partial s} < 0$ ,  $\frac{\partial X_1(t,s)}{\partial t} < 0$ , and  $\frac{d X_2(s)}{ds} < 0$ . Using Equations (3.21) and (3.22), and solving for s and t we find:

$$s^{MCF} = \frac{B_y + p_2'(X_2)\frac{X_2}{n}}{1+\epsilon} + \frac{\epsilon}{1+\epsilon} \left[ \frac{T - X_1 - X_2}{\frac{dX_2}{ds}} \right] + \frac{\epsilon}{1+\epsilon} X_1 \left[ \frac{\frac{\partial X_1}{\partial s}}{\frac{dX_2}{ds}\frac{\partial X_1}{\partial t}} \right] (3.23)$$
$$t^{MCF} = \frac{D'(X_1) - p_2'(X_2)\frac{X_2}{n}}{1+\epsilon} - \frac{\epsilon}{1+\epsilon} \left[ \frac{\frac{\partial X_1}{\partial s}X_1}{\frac{dX_2}{ds}\frac{\partial X_1}{\partial t}} + \frac{T - X_1 - X_2}{\frac{dX_2}{ds}} + \frac{X_1}{\frac{\partial X_1}{\partial t}} \right] (3.24)$$

The second-best PES and environmental tax are now defined taking into account their costs as far as public finance is concerned. Environmental policy tool design combines the direct effect on the environment and market power and indirectly the induced changes in several land uses computed to the MCF. Comparing (3.16) and (3.23) shows that  $PES^{MCF} < PES$  whereas the comparison is not simple for  $t^{MCF}$ and t.

Contrary to the intuition, the effect of a change in  $\epsilon$  in  $t^{MCF}$  and  $PES^{MCF}$  is not immediate (see Appendix C and D in Section 3.6 for full calculations). To investigate this point, we use (3.12) and (3.13) with  $X_1(s(\epsilon), t(\epsilon))$  and  $X_2(s(\epsilon))$ . The variation of  $t^{MCF}$  and  $PES^{MCF}$  with respect to the MCF is mainly undetermined. Restricting conditions, we obtain:

If 
$$e_{X_1/t} > -1$$
:  $\frac{ds}{d\epsilon} < 0$  and  $\frac{dt}{d\epsilon} > 0$  if  $\frac{\partial X_1}{\partial t} / \frac{\partial X_2}{\partial s} > \varpi$ 

If the elasticity of demand of the conventional agricultural good with respect to the environmental tax is low, the PES will always decrease and the environmental tax will increase with the MCF provided that both environmental prices favor uncultivated land in an efficient way. In the presence of distortionary taxation, the regulator exploits a contributory component of the environmental incentive tax.

Indeed if  $e_{X_1/t} > -1$ , the production level of conventional agriculture will not be significantly reduced after the implementation of the environmental tax. An increase in the marginal cost of public funds will increase the environmental tax, provided also that the impact of changes in production levels induced by the environmental tax and the PES are higher than a threshold given by  $\varpi$ . The environmental tax should reduce the level of conventional agricultural production more than the PES diminishes the level of organic production. In other words, the uncultivated land should be further to the detriment of conventional agriculture than to the detriment of organic agriculture. The introduction of the MCF leads the regulator to exploit a contributory component of the incentive tax while keeping in mind the objective of providing the right environmental incentives. Consequently, the environmental tax, which initially has an incentive objective (Bureau and Mougeot, 2005), would also have a contributory outcome when the MCF is taken into account .

If the farmers cultivate the entire land  $(\lambda > 0)$ , the introduction of the MCF modifies the welfare function as follows:

$$W(T - X_{2}(t), X_{2}(t)) = \int_{0}^{T - X_{2}(t)} p_{1}(u) du + \int_{0}^{X_{2}(t)} p_{2}(v) dv - nc_{1}\left(\frac{T - X_{2}(t)}{n}\right) - nc_{2}\left(\frac{X_{2}(t)}{n}\right) + B\left(T - (T - X_{2}(t)) - X_{2}(t)\right) - D(T - X_{2}(t)) + \epsilon t X_{1}(t) - \epsilon s(T - (T - X_{2}(t)) - X_{2}(t))$$

Maximizing welfare yields this following first-order condition:

$$\frac{dX_2}{dt}\left[-p_1(T-X_2)+p_2(X_2)+c_1'(\frac{T-X_2}{n})-c_2'(\frac{X_2}{n})+D'(T-X_2)-\epsilon t\right]+\epsilon(T-X_2)=0$$
(3.25)

Using equations (3.7) and (3.8) from the profit maximization gives:

$$-p_1 + c_1'(X_1) + p_2(X_2) - c_2'(X_2) = -t - p_2'(X_2)\frac{X_2}{n}$$
(3.26)

We can then write equation (3.25) as:

$$\frac{dX_2}{dt} \left[ -t - p_2'(X_2)\frac{X_2}{n} + D'(T - X_2) - \epsilon t \right] + \epsilon(T - X_2) = 0$$
(3.27)

Then, we solve equation (3.27) and we obtain the second-best environmental tax level:

$$t^{MCF} = \frac{D'(X_1) - p'_2(X_2)\frac{X_2}{n}}{1 + \epsilon} + \frac{\epsilon}{1 + \epsilon} \left(\frac{X_1}{\frac{dX_2}{dt}}\right)$$
(3.28)

We saw in Section 3 that the second-best environmental tax is the same, whether all the land is cultivated or not. This is not the case when including the MCF. Since there is no uncultivated land, the indirect effects are limited to organic and conventional agricultural production. This environmental tax is always lower than its design without the MCF. The regulator uses the contributory component of the incentive environmental tax in the presence of the MCF if the demand for the agricultural good is inelastic with respect to the environmental tax (see Appendix):

If 
$$e_{X_1/t} > -1$$
,  $\frac{dt}{d\epsilon} > 0$ 

Proposition 2 summarizes our results:

**Proposition 2.** If the demand elasticity of the conventional agricultural food is inelastic with respect to the environmental tax, the MCF decreases the secondbest PES but increases the environmental tax provided both environmental prices favor uncultivated land in an efficient way. The regulator exploits the contributory component of the environmental incentive tax.

## 3.5 Conclusion

Pollution and biodiversity benefits are two externalities associated with agricultural land that lead to market failure. According to the Tinbergen rule, multiple market failures require multiple policies to address them. Here, we looked at the scenario where an environmental tax and a PES scheme are used to address pollution and biodiversity conservation, respectively. We added an additional market distortion in the form of an oligopoly in organic agriculture production. We found that the second-best tax on conventional agriculture production is higher than the marginal damage from pollution, and the second-best PES for biodiversity is lower than the marginal benefit. An important characteristic of a public PES scheme is the necessity to raise funds to finance it, which can also be at the origin of economic distortions. In order to account for this aspect, we then introduced the marginal social cost of public funds (MCF). The PES decreases with the MCF, whereas the Pigouvian tax increases with the MCF, provided that demand for the conventional agriculture good is inelastic and environmental policies provide buffer strips efficiently. This article highlights a contributory component of the environmental incentive tax. Indeed, the primary objective of a Pigouvian tax is to give the appropriate incentives to agents and not to raise a revenue for the regulator. This study also identifies cases where the PES is ineffective in promoting biodiversity.

This study was extended by considering other assumptions. First, we have challenged the assumption of a neutral impact of organic farming on biodiversity by assuming that fallow buffer strips produce more biodiversity than organic farming. In this case, we use two PES. The level of organic farming would be subject to two effects: a negative effect that favors buffer strips and a positive effect that favors biodiversity from organic farming. The first effect would therefore outweigh the second and the mechanisms highlighted in this paper would remain relevant. Second, under our assumptions, we have modified the environmental policy tools by considering two PES schemes, one on uncultivated land and the other on organic agriculture but no environmental tax. We found that the PES for organic agriculture takes the market power into account, and is higher than the marginal benefit of organic production, whereas the PES for uncultivated land is equal to the marginal benefit of biodiversity and no longer adjusts to incorporate the market power. Finally, we have challenged the assumption that there are no negative externalities of conventional agricultural production on the level of organic production. In this case, we found that the farmer will internalize this negative impact himself and the PES and environmental tax levels do not differ from those in the main scenario of this paper. However, the definition of PES when externalities between productions cannot be directly internalized should be further analyzed in another study. This is the case when farmers are different.

The issue of market power in the organic sector may be questionable. This market power can be justified by the non-uniform distribution of organic farming across the country and transport constraints for organic products. It is possible that for certain organic agricultural goods this assumption is not valid contrary to other organic agricultural goods. The objective of this theoretical article is to contribute to the economic literature by proposing a normative analysis of PES schemes while integrating different types of distortions. An amended version of this work could consider differentiated demands for agricultural goods that occur for some level of market power.

In this paper, PES remunerate the benefits from environmental services provided by farmers. This is in contrast to actual schemes which allocate payments based on foregone profits from adopting environmental practices. This article does not take into account the additionality issue under asymmetric information. Indeed, the farmer can leave some land uncultivated before any policy is introduced because it is not profitable for him to use all of his land in agricultural production. In this case, when a PES scheme is implemented, there is a windfall effect because the farmer will be subsidized for all uncultivated land, even the land he would have left uncultivated in the absence of any policy. The size of the windfall effect can be unknown to the regulator under asymmetric information. Further research is needed to investigate these different questions.

## 3.6 Appendices of Chapter 3

## Appendix A. Welfare function concavity

• If  $\lambda = 0$ , we construct the Hessian matrix, I(W):

$$I(W) = \begin{bmatrix} \frac{\partial^2 X_1}{\partial s^2} [F] + (\frac{\partial X_1}{\partial s})^2 [F'] + \frac{d^2 X_2}{ds^2} [G] + (\frac{d X_2}{ds})^2 [G'] & \frac{\partial^2 X_1}{\partial s \partial t} [F] + \frac{\partial X_1}{\partial s} \frac{\partial X_1}{\partial t} [F'] \\ \frac{\partial^2 X_1}{\partial s \partial t} [F] + \frac{\partial X_1}{\partial s} \frac{\partial X_1}{\partial t} [F'] & \frac{\partial^2 X_1}{\partial t^2} [F] + (\frac{\partial X_1}{\partial t})^2 [F] \end{bmatrix}$$

where

$$F = p_1(X_1) - c'_1(\frac{X_1}{n}) - B_y - D'(X_1)$$
  

$$F' = p'_1(X_1) - \frac{1}{n}c''_1(\frac{X_1}{n}) + B_{yy} - D''(X_1)$$
  

$$G = p_2(X_2) - c'_2(\frac{X_2}{n}) - B_y$$
  

$$G' = p'_2(X_2) - \frac{1}{n}c''_2(\frac{X_2}{n}) + B_{yy}$$

**.**...

Following our assumptions about demand and cost structures, we can simplify the above matrix to

$$I(W) = \begin{bmatrix} (\frac{\partial X_1}{\partial s})^2 [F'] + (\frac{dX_2}{ds})^2 [G'] & \frac{\partial X_1}{\partial s} \frac{\partial X_1}{\partial t} [F'] \\ \\ \frac{\partial X_1}{\partial s} \frac{\partial X_1}{\partial t} [F'] & (\frac{\partial X_1}{\partial t})^2 [F'] \end{bmatrix}$$

Based on our assumptions, we know F' < 0 and G' < 0. Using this information, we calculate the determinant of I:

$$Det(I) = \left[ \left[ \left(\frac{\partial X_1}{\partial s}\right)^2 [F'] + \left(\frac{\partial X_2}{\partial s}\right)^2 [G'] \right] * \left(\frac{\partial X_1}{\partial t}\right)^2 [F'] \right] - \left[ \frac{\partial X_1}{\partial s} \frac{\partial X_1}{\partial t} [F'] * \frac{\partial X_1}{\partial s} \frac{\partial X_1}{\partial t} [F'] \right]$$

After simplification, we obtain:

$$Det(I) = (\frac{dX_2}{ds})^2 [G'] (\frac{\partial X_1}{\partial t})^2 [F'] > 0$$

Thus, the welfare function is concave because the determinant is positive while  $\left[\frac{dX_2}{ds}\right]^2 [G'] + \left[\frac{\partial X_1}{\partial t}\right]^2 [F'] < 0.$
• Next, we look at the case where  $\lambda > 0$ , referring to (3.18):

$$\frac{d^2W}{dt^2} = \frac{d^2X_1}{dt^2} [p_1(X_1) - p_2(T - X_1) - c_1'(\frac{X_1}{n}) + c_2(\frac{T - X_1}{n}) - D'(X_1)] + (\frac{dX_1}{dt})^2 [p_1'(X_1) + p_2'(T - X_1) - \frac{1}{n}c_1''(\frac{X_1}{n}) - \frac{1}{n}c_2''(\frac{T - X_1}{n}) - D''(X_1)]$$

Under our assumptions, we have:

$$\left(\frac{dX_1}{dt}\right)^2 \left[p_1'(X_1) + p_2'(T - X_1) - \frac{1}{n}c_1''(\frac{X_1}{n}) - \frac{1}{n}c_2''(\frac{T - X_1}{n}) - D''(X_1)\right] < 0$$

Therefore, the welfare function is still concave when  $\lambda > 0$ .

# Appendix B. Welfare function concavity under the marginal social cost of public funds

• If  $\lambda = 0$ , we use (3.21) and (3.22) to create the Hessian matrix:

$$H = \begin{bmatrix} a & b \\ c & d \end{bmatrix}$$

where

$$a = \frac{\partial^2 X_1}{\partial s^2} [A + \epsilon(t+s)] + (\frac{\partial X_1}{\partial s})^2 [A'] + 2\epsilon \frac{\partial X_1}{\partial s} + \frac{d^2 X_2}{ds^2} [B + \epsilon s] + [\frac{d X_2}{ds}]^2 [B'] + 2\epsilon \frac{d X_2}{ds}$$

$$b = \frac{\partial^2 X_1}{\partial s \partial t} [A + \epsilon(t+s)] + \frac{\partial X_1}{\partial t} \frac{\partial X_1}{\partial s} [A'] + \epsilon [\frac{\partial X_1}{\partial s} + \frac{\partial X_1}{\partial t}]$$

$$c = \frac{\partial^2 X_1}{\partial t \partial s} [A + \epsilon(t+s)] + \frac{\partial X_1}{\partial t} \frac{\partial X_1}{\partial s} [A'] + \epsilon [\frac{\partial X_1}{\partial s} + \frac{\partial X_1}{\partial t}]$$

$$d = \frac{\partial^2 X_1}{\partial t^2} [A + \epsilon(t+s)] + (\frac{\partial X_1}{\partial t})^2 [A'] + 2\epsilon \frac{\partial X_1}{\partial t}$$

and

$$A = p_1(X_1) - c'_1(\frac{X_1}{n}) - B_y - D'(X_1)$$
  

$$B = p_2(X_2) - c'_2(\frac{X_2}{n}) - B_y$$
  

$$A' = p'_1(X_1) - \frac{1}{n}c''_1(\frac{X_1}{n}) + B_{yy} - D''(X_1)$$
  

$$B' = p'_2(X_2) - \frac{1}{n}c''_2(\frac{X_2}{n}) + B_{yy}$$

Thanks to our assumptions, we can simplify the Hessian to:

$$H = \begin{bmatrix} \left(\frac{\partial X_1}{\partial s}\right)^2 [A'] + \left[\frac{dX_2}{ds}\right]^2 [B'] + 2\epsilon \left(\frac{\partial X_1}{\partial s} + \frac{dX_2}{ds}\right) & \left(\frac{\partial X_1}{\partial t}\right)^2 [A'] + 2\epsilon \left(\frac{\partial X_1}{\partial t}\right) \\ \\ \left(\frac{\partial X_1}{\partial t}\right)^2 [A'] + 2\epsilon \left(\frac{\partial X_1}{\partial t}\right) & \left(\frac{\partial X_1}{\partial t}\right)^2 [A'] + 2\epsilon \frac{\partial X_1}{\partial t} \end{bmatrix}$$

So the determinant is:

$$Det = \left\{ \left(\frac{\partial X_1}{\partial s}\right)^2 [A'] + \left[\frac{dX_2}{ds}\right]^2 [B'] + 2\epsilon \left(\frac{\partial X_1}{\partial s} + \frac{dX_2}{ds}\right) * \left(\frac{\partial X_1}{\partial t}\right)^2 [A'] + 2\epsilon \frac{\partial X_1}{\partial t} \right\} - \left\{ \left(\frac{\partial X_1}{\partial t}\right)^2 [A'] + 2\epsilon \frac{\partial X_1}{\partial t} \right\}^2$$

Simplifying, we find:

$$Det = \left(\frac{dX_2}{ds}\right)^2 \left(\frac{\partial X_1}{\partial t}\right)^2 [A'][B'] + 2\epsilon \left(\frac{\partial X_1}{\partial t}\frac{dX_2}{ds}\right) \left(\frac{dX_2}{ds}[B'] + \frac{\partial X_1}{\partial t}[A']\right) + 4\epsilon^2 \frac{dX_2}{ds} \frac{\partial X_1}{\partial t} > 0$$

With A' < 0 and B' < 0, we find a positive determinant. And, because  $\left(\frac{\partial X_1}{\partial s}\right)^2 [A'] + \left[\frac{dX_2}{ds}\right]^2 [B'] + 2\epsilon \left(\frac{\partial X_1}{\partial s} + \frac{dX_2}{ds}\right) < 0$ , we have a concave function.

• If  $\lambda > 0$ , we refer to (3.25):

$$\frac{d^2W}{dt^2} = \frac{d^2X_1}{dt^2}[E+\epsilon t] + \left(\frac{dX_1}{dt}\right)^2[E'] + 2\epsilon\frac{dX_1}{dt}$$

where

$$E = p_1(X_1) - p_2(T - X_1) - c_1'\left(\frac{X_1}{n}\right) + c_2'\left(\frac{T - X_1}{n}\right) - D'(X_1)$$
  

$$E' = p_1'(X_1) + p_2'(T - X_1) - \frac{1}{n}c_1''\left(\frac{X_1}{n}\right) - \frac{1}{n}c_2''\left(\frac{T - X_1}{n}\right) - D''(X_1) < 0$$

With our assumptions we can simplify this to:

$$\frac{d^2W}{dt^2} = \left(\frac{dX_1}{dt}\right)^2 [E'] + 2\epsilon \frac{dX_1}{dt} < 0$$

Thus, our welfare function is still concave when  $\lambda > 0$ .

#### Appendix C. Tax and PES changes with the MCF if Y>0

According to (3.23), t and s depend on  $\epsilon$ . Moreover t and s must satisfy conditions (3.21) and (3.22). We set:

$$\begin{aligned} q &= p_1'(X_1(t(\epsilon), s(\epsilon))) - \frac{1}{n} c_1''\Big(\frac{X_1(t(\epsilon), s(\epsilon))}{n}\Big) - D''(X_1(t(\epsilon), s(\epsilon))) < 0 \\ z &= p_2'(X_2(s(\epsilon))) - \frac{1}{n} c_2''\Big(\frac{X_2(s(\epsilon))}{n}\Big) < 0 \end{aligned}$$

Additionally, we know:  $\frac{\partial X_1}{\partial t} = \frac{\partial X_1}{\partial s} < 0.$ 

• We differentiate (3.21) and (3.22) with respect to  $\epsilon$  and rearrange the equations into the following matrix form:

$$\begin{bmatrix} \frac{ds}{d\epsilon} \\ \frac{dt}{d\epsilon} \end{bmatrix} = K \begin{bmatrix} -\frac{\partial X_1}{\partial s}[t+s] - \frac{\partial X_2}{\partial s}s + (T - X_1 - X_2) \\ -\frac{\partial X_1}{\partial t}(t+s) - X_1 \end{bmatrix}$$

where 
$$K = \begin{bmatrix} i & j \\ k & l \end{bmatrix}$$
, with:  
 $i = \frac{\partial X_1}{\partial s} [(q + B_{yy}) \frac{\partial X_1}{\partial s} + 2\epsilon + B_{yy} \frac{\partial X_2}{\partial s}] + \frac{\partial X_2}{\partial s} [(z + B_{yy}) \frac{\partial X_2}{\partial s} + B_{yy} \frac{\partial X_1}{\partial s} + 2\epsilon]$   
 $j = \frac{\partial X_1}{\partial s} [(q + B_{yy}) \frac{\partial X_1}{\partial t} + 2\epsilon + B_{yy} \frac{\partial X_2}{\partial s}]$ 

$$k = \frac{\partial X_1}{\partial t} [(q + B_{yy}) \frac{\partial X_1}{\partial s} + B_{yy} \frac{\partial X_2}{\partial s} + 2\epsilon]$$
  

$$l = \frac{\partial X_1}{\partial t} [(q + B_{yy}) \frac{\partial X_1}{\partial t} + 2\epsilon]$$

• We multiply each side of the equation by  $K^{-1}$  to isolate  $\frac{ds}{d\epsilon}$  and  $\frac{dt}{d\epsilon}$ :

$$\begin{bmatrix} \frac{ds}{d\epsilon} \\ \frac{dt}{d\epsilon} \end{bmatrix} = K^{-1} \begin{bmatrix} -\frac{\partial X_1}{\partial s} [t+s] - \frac{\partial X_2}{\partial s} s + (T - X_1 - X_2) \\ -\frac{\partial X_1}{\partial t} (t+s) - X_1 \end{bmatrix}$$
(3.29)

where

$$K^{-1} = \frac{1}{\det K} \begin{bmatrix} l & -j \\ -k & i \end{bmatrix}$$

• We calculate  $\det K$  :

$$\begin{split} Det = & \left\{ \frac{\partial X_1}{\partial t} [(q + B_{yy}) \frac{\partial X_1}{\partial t} + 2\epsilon] \right\} \left\{ \frac{\partial X_1}{\partial s} [(q + B_{yy}) \frac{\partial X_1}{\partial s} + 2\epsilon + B_{yy} \frac{\partial X_2}{\partial s}] \\ & \quad + \frac{\partial X_2}{\partial s} [(z + B_{yy}) \frac{\partial X_2}{\partial s} + B_{yy} \frac{\partial X_1}{\partial s} + 2\epsilon] \right\} \\ & \quad - \left[ \frac{\partial X_1}{\partial s} [(q + B_{yy}) \frac{\partial X_1}{\partial t} + 2\epsilon + B_{yy} \frac{\partial X_2}{\partial s}] \right]^2 \\ Det = & \frac{\partial X_1^2}{\partial t} \frac{\partial X_2^2}{\partial s} qz + \frac{\partial X_1^2}{\partial t} \frac{\partial X_2^2}{\partial s} qB_{yy} + \frac{\partial X_1^2}{\partial t} \frac{\partial X_2^2}{\partial s} zB_{yy} + 2\frac{\partial X_1^2}{\partial t} \frac{\partial X_2}{\partial s} q\epsilon \\ & \quad + 2\frac{\partial X_1}{\partial t} \frac{\partial X_2^2}{\partial s} z\epsilon + 2\frac{\partial X_1^2}{\partial t} \frac{\partial X_2}{\partial s} B_{yy} \epsilon + 2\frac{\partial X_1}{\partial t} \frac{\partial X_2^2}{\partial s} B_{yy} \epsilon + 4\frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} \epsilon^2 > 0 \end{split}$$

because q < 0 and z < 0,  $\frac{\partial X_1}{\partial t} = \frac{\partial X_1}{\partial s} < 0$  and  $\frac{\partial X_2}{\partial s} < 0$ .

• We calculate  $\frac{ds}{d\epsilon}$ , using (3.29):

$$\begin{split} \frac{\partial s}{\partial \epsilon} &= \frac{1}{\det} \Big\{ \left[ \frac{\partial X_1}{\partial t} \left[ (q + B_{yy}) \frac{\partial X_1}{\partial t} + 2\epsilon \right] \Big\} \Big\{ - \frac{\partial X_1}{\partial s} \left[ t + s \right] - \frac{\partial X_2}{\partial s} s + (T - X_1 - X_2) \Big\} \\ &\quad + \frac{1}{\det} \Big\{ - \frac{\partial X_1}{\partial s} \left[ (q + B_{yy}) \frac{\partial X_1}{\partial t} + 2\epsilon + B_{yy} \frac{\partial X_2}{\partial s} \right] \Big\} \Big\{ - \frac{\partial X_1}{\partial t} (t + s) - X_1 \Big\} \\ &\quad \frac{\partial s}{\partial \epsilon} = \frac{1}{\det} \Big\{ - \frac{\partial X_1}{\partial t}^2 \frac{\partial X_2}{\partial s} qs + \frac{\partial X_1^2}{\partial t} q(T - X_2) + \underbrace{\frac{\partial X_1}{\partial t}^2 \frac{\partial X_2}{\partial s} tB_{yy}}_{>0} + \frac{\partial X_1^2}{\partial t}^2 B_{yy} (T - X_2) \Big\} \\ &\quad + \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} X_1 B_{yy} - 2 \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} s\epsilon + 2 \frac{\partial X_1}{\partial t} \epsilon(T - X_2) \Big\} \\ &\text{So } \frac{\partial s}{\partial \epsilon} < 0 \text{ if } \frac{\partial X_1^2}{\partial t}^2 \frac{\partial X_2}{\partial s} tB_{yy} + \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} X_1 B_{yy} < 0 \Leftrightarrow \frac{\partial X_1}{\partial t} \frac{\partial x_2}{\partial s} B_{yy} [\frac{\partial X_1}{\partial t} t + X_1] < 0 \\ &\text{i.e. if } \frac{\partial X_1}{\partial t} t + X_1 > 0 \Leftrightarrow \frac{\partial X_1}{\partial t} \frac{t}{X_1} + 1 > 0 \Leftrightarrow \underbrace{\frac{\partial X_1}{\partial t} \frac{t}{X_1}}_{e_{X_1/t}} = -1 \end{split}$$

So 
$$\frac{\partial s}{\partial \epsilon} < 0$$
 if  $e_{X_1/t} > -1$ .

• We calculate  $\frac{dt}{d\epsilon}$ , using (3.29):

$$\begin{split} \frac{\partial t}{\partial \epsilon} &= \frac{1}{\det} \Big\{ -\frac{\partial X_1}{\partial t} \big[ (q+B_{yy}) \frac{\partial X_1}{\partial s} + B_{yy} \frac{\partial X_2}{\partial s} + 2\epsilon \big] \big[ -\frac{\partial X_1}{\partial s} (t+s) - \frac{\partial X_2}{\partial s} s \\ &+ (T-X_1-X_2) \big] + \big[ \frac{\partial X_1}{\partial s} \big[ (q+B_{yy}) \frac{\partial X_1}{\partial s} + 2\epsilon + B_{yy} \frac{\partial X_2}{\partial s} \big] \\ &+ \frac{\partial X_2}{\partial s} \big[ (z+B_{yy}) \frac{\partial X_2}{\partial s} + B_{yy} \frac{\partial X_1}{\partial s} + 2\epsilon \big] \big] \big[ -\frac{\partial X_1}{\partial t} (t+s) - X_1 \big] \Big\} \\ \frac{\partial t}{\partial \epsilon} &= \frac{1}{\det} \Big\{ \frac{\partial X_1}{\partial t}^2 \frac{\partial X_2}{\partial s} qs - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s}^2 zs - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s}^2 zt - \frac{\partial X_1^2}{\partial t}^2 qT - \frac{\partial X_2^2}{\partial s}^2 zx_1 \\ &+ \frac{\partial X_1^2}{\partial t} qx_2 - \frac{\partial X_1}{\partial t}^2 \frac{\partial X_2}{\partial s} tB_{yy} - \frac{\partial X_1}{\partial t} \frac{\partial X_2^2}{\partial s}^2 tB_{yy} - \frac{\partial X_1^2}{\partial t}^2 TB_{yy} \\ &- \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} TB_{yy} - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} X_1 B_{yy} - \frac{\partial X_2^2}{\partial s}^2 X_1 B_{yy} + \frac{\partial X_1^2}{\partial t}^2 X_2 B_{yy} \\ &+ \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} x_2 B_{yy} - 2 \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} t\epsilon - 2 \frac{\partial X_1}{\partial t} T\epsilon - 2 \frac{\partial X_2}{\partial s} X_1 \epsilon + 2 \frac{\partial X_1}{\partial t} X_2 \epsilon \Big\} \\ \frac{\partial t}{\partial \epsilon} &= \frac{1}{\det} \Big\{ \frac{\partial X_1^2}{\partial t} \frac{\partial X_2}{\partial s} qs - \frac{\partial X_1^2}{\partial t} q(T-X_2) - \frac{\partial X_1^2}{\partial t}^2 B_{yy} (T-X_2) - 2 \frac{\partial X_1}{\partial t} \epsilon (T-X_2) \\ &- \frac{\partial X_1^2}{\partial t} \frac{\partial X_2}{\partial s} tB_{yy} - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} X_1 B_{yy} - \frac{\partial X_1}{\partial t} \frac{\partial X_2^2}{\partial s}^2 zs - \frac{\partial X_1}{\partial t} \frac{\partial X_2^2}{\partial s} zt \\ &- \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} tB_{yy} - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} X_1 B_{yy} - \frac{\partial X_1}{\partial t} \frac{\partial X_2^2}{\partial s}^2 zs - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s}^2 zt \\ &- \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} tB_{yy} - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} X_1 B_{yy} - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s}^2 zs - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s}^2 zt \\ &- \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} tb - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s}^2 x_1 B_{yy} - 2 \frac{\partial X_2}{\partial s} X_1 \epsilon - \frac{\partial X_2}{\partial s}^2 zX_1 \\ &- 2 \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} t\epsilon - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s}^2 tB_{yy} \Big\} \end{aligned}$$

We know that

- $-\frac{\partial X_1}{\partial t}^2 \frac{\partial X_2}{\partial s} t B_{yy} \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} X_1 B_{yy} > 0 \Leftrightarrow -\frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} B_{yy} [\frac{\partial X_1}{\partial t} t X_1] > 0 \Leftrightarrow \frac{\partial X_1}{\partial t} \frac{t}{X_1} > -1$
- $-2\frac{\partial X_1}{\partial t}\frac{\partial X_2}{\partial s}t\epsilon 2\frac{\partial X_2}{\partial s}X_1\epsilon > 0 \Leftrightarrow -2\frac{\partial X_2}{\partial s}\epsilon[\frac{\partial X_1}{\partial t}t + X_1] > 0 \Leftrightarrow \frac{\partial X_1}{\partial t}\frac{t}{X_1} > -1$ •  $-\frac{\partial X_1}{\partial t}\frac{\partial X_2}{\partial s}^2 \tau t - \frac{\partial X_2}{\partial s}^2 \tau X_1 > 0 \text{ if } -\frac{\partial X_2}{\partial s}^2 z[\frac{\partial X_1}{\partial t}t + X_1] > 0 \Leftrightarrow \frac{\partial X_1}{\partial t}\frac{t}{X_1} > -1$

• 
$$-\frac{\partial X_1}{\partial t}\frac{\partial X_2}{\partial s}zt - \frac{\partial X_2}{\partial s}zX_1 > 0$$
 if  $-\frac{\partial X_2}{\partial s}z[\frac{\partial X_1}{\partial t}t + X_1] > 0 \Leftrightarrow \frac{\partial X_1}{\partial t}\frac{t}{X_1} > -$ 

 $\begin{aligned} \bullet & -\frac{\partial X_2}{\partial s}^2 X_1 B_{yy} - \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s}^2 t B_{yy} > 0 \text{ if } -\frac{\partial X_2}{\partial s}^2 B_{yy} [\frac{\partial X_1}{\partial t} t + X_1] > 0 \Leftrightarrow \frac{\partial X_1}{\partial t} \frac{t}{X_1} > -1 \\ \bullet & -\frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s}^2 zs + \frac{\partial X_1}{\partial t}^2 \frac{\partial X_2}{\partial s} qs > 0 \Leftrightarrow \frac{\partial X_1}{\partial t} \frac{\partial X_2}{\partial s} s[\frac{\partial X_1}{\partial t} q - \frac{\partial X_2}{\partial s} z] > 0 \Leftrightarrow [\frac{\partial X_1}{\partial t} q - \frac{\partial X_2}{\partial s} z] > 0 \Leftrightarrow [\frac{\partial X_1}{\partial t} q - \frac{\partial X_2}{\partial s} z] > 0 \Leftrightarrow \frac{\partial X_1}{\partial t} / \frac{\partial X_2}{\partial s} > z/q \equiv \varpi. \end{aligned}$ 

$$\Rightarrow \frac{\partial t}{\partial \epsilon} > 0 \text{ if } e_{X_1/t} > -1 \text{ and } \frac{\partial X_1}{\partial t} / \frac{\partial X_2}{\partial s} > \varpi.$$

#### Appendix D. Tax and PES changes with the MCF if Y=0

We use (3.25) and we set:  $J(t, \epsilon) = \frac{dX_2}{dt} \left[-p_1(T - X_2) + p_2(X_2) + c'_1(\frac{T - X_2}{n}) - c'_2(\frac{X_2}{n}) + D'(T - X_2) - \epsilon t\right] + \epsilon(T - X_2)$ . Applying the implicit function theorem we find:

$$\frac{dt}{d\epsilon} = -\frac{\frac{\partial J}{\partial \epsilon}}{\frac{\partial J}{\partial t}} = -\frac{\frac{dX_1}{dt}t + X_1}{\frac{dX_1}{dt}[p_1'(X_1) + p_2'(T - X_1) - \frac{1}{n}c_1''(\frac{X_1}{n}) - \frac{1}{n}c_2''(\frac{T - X_1}{n}) - D''(X_1)] + 2\frac{dX_1}{dt}\epsilon}$$

We know that the denominator of the above expression is negative. So we obtain  $\frac{dt}{d\epsilon} > 0$  if  $e_{X_1/t} > -1$ .

# Chapter 4

# Are Additionality-Based Payments for Environmental Services Efficient?

Payments for Environmental Services (PES) may face a financing constraint, especially when the buyer is a public regulator. An additionality-based PES can address this problem. This paper aims to study the efficiency of additionality-based PES. We consider a farmer who allocates his land between organic production, conventional production causing environmental damage, or biodiversity-generating grass strips. Using a two-period model, we introduce a PES in the final period, remunerating the additional grass strips provided by the farmer. We show that the farmer distorts his behavior in the initial period, in order to obtain more payment in the final period. The second-best PES to limit this behavior equals the discounted difference of the marginal environmental benefits obtained in each period. The second-best value of environmental taxes in the presence of this PES are no longer equal to the marginal damage and are amended to take into account the distortions caused by the additionality-based PES. The analysis is then extended by taking into account market power in the organic market. It turns out that market power reduces the distortion due to the additionality-based PES in the initial period but reduces the organic production quantity in the final period. The second-best PES depends on the size of these two effects and environmental taxes under market power have to be amended. Finally, this paper shows that an additionality-based PES never achieves environmental efficiency, even in a competitive market framework. Furthermore, this paper provides new insights into understanding the interactions between different environmental policies in the presence of several types of distortions.<sup>1</sup>

<sup>&</sup>lt;sup>1</sup>This chapter is joint work with Sonia Schwartz: Krautkraemer, A., S. Schwartz (2023). Are additonality-based PES efficient, document de recherche du LEO, DR LEO 2023-01, available at https://univ-orleans.hal.science/hal-04152591/

**Keywords**: Biodiversity; Payment for Environmental Services; Pigouvian Taxes; Additionality

# 4.1 Introduction

The concept of ecosystem services was widely disseminated with the publication of the Millennium Ecosystem Assessment in 2005. Ecosystem services are defined as the many and varied benefits that humans derive from the natural environment and healthy ecosystems. They are categorized into the following four types: provisioning services such as food, water, timber, and fiber; regulating services that affect climate, floods, disease, wastes, and water quality; cultural services that provide recreational, aesthetic, and spiritual benefits; and supporting services such as soil formation, photosynthesis, and nutrient cycling (Reid et al., 2005).

The notion of ecosystem services and environmental services (ES) are often confused. While ecosystem services refer to the functioning of ecosystems, ES concern to the notion of externalities induced by human activities. The FAO (Food and Agriculture Organization of the United Nations) proposes a definition of ES in terms of ecosystem services. For agriculture, they are defined as the subpart of ecosystem services that can be qualified in terms of externalities, i.e. all ecosystem services except provisioning services. Lugo (2007) draws a distinction between benefits provided by ecosystems and human protection of these ecosystems and use the term ecosystem services to refer to the former exclusively. Thus the term ES can be used to refer to the production of services by farmers in order to protect the environment (FAO, 2007). We can quote several examples. Long crop rotations improve ecosystem services such as support services through improved soil quality. The diversity of productive activities on a farm promotes beneficial interactions between crops and livestock and the management of landscape features such as grass strips, embankments, hedges or watercourses contribute to the ecological functioning of agroecosystems (Duval et al., 2016). All of these definitions make it possible to justify paying farmers for these ES as internalization of externalities. In particular, this leaves room for policy intervention to encourage their optimal provision.

At the European level, the Common Agricultural Policy (CAP) is the main tool for the production objectives of agriculture with those of environmental and human health protection. and human health. One of the major advances of the CAP has been the introduction of agri-environmental schemes (AES). They consist of offering financial compensation to farmers for voluntary commitment on their part, over several years, to implement practices or production. For example, some AES contracts are offered for a period of five years, and compensate the farmer for the decrease in profit associated with adopting environmental practices. Agrienvironmental measures can be considered as Payments for Environmental Services (PES).

PES is one policy tool that has been implemented to try to increase the provision of ES. One of the most widely cited definitions of PES comes from Wunder (2005), who defines PES as a voluntary transaction where a well-defined ES or a land-use that is likely to produce that service is bought by a (minimum one) ES buyer from a (minimum one) ES provider if and only if the ES provider secures ES provision (conditionality). Conditionality can be difficult to evaluate in results-based PES schemes, as some ES are difficult to measure. In practice, it is much more common to see action-based PES schemes conditional on land use or specific management practices.

One major factor in the economic efficiency of PES programs is whether or not they are *additional*; that is, they lead to the provision of an ES that would not have occurred in the absence of any payment. Early on in PES development, a majority of programs had no additionality requirement, possibly due to the idea that monitoring additionality would prove to be too costly (Bennett, 2010). Or, as in the case of the national program in Costa Rica, the aim may be to recognize and remunerate any ES provision regardless of its additionality (Bennett, 2010). It is only more recently that evaluating the additionality of PES programs has become a concern, even though doing so is essential for a PES scheme to achieve its environmental target with economic efficiency while maintaining investor confidence (Bennett, 2010).

Wunder (2005) explains that establishing a baseline level of ES is essential in order to assess the additionality of a PES program and thus to avoid paying for ES that would have been provided without the program, leading to windfall gains for the ES seller, and a lost opportunity to pay for ES where they would be additional. However, since establishing the baseline level of ES can be costly, a regulator or other ES buyer may rely on the ES seller to report this information. When payments are based on additionality, this gives the ES seller incentive to under-report their current level of ES provision in order to earn payments for more units of ES provision, which is an example of moral hazard. When the ES buyer is a public regulator, the issue of additionality is even more important as it prevents wasting public funds.

The economic literature has explored questions associated with additionality (Sills et al., 2008), notably the definition of the baseline (Wunder, 2005; Kaczan et al., 2017), and the consequences in terms of new technology development (Pates and Hendricks, 2020). Another branch of the literature looks at agency theory to solve

the problem of additionality. Other works have investigated if existing programs, particularly agri-environmental schemes, demonstrate additionality (Mezzatesta et al., 2013; Chabé-Ferret and Subervie, 2013). Nevertheless, the literature has not studied the efficiency of an additionality-based PES to obtain the optimal levels of environmental benefits. That is the main objective of this paper.

To do this study, we consider a farmer who has to choose to allocate his land between organic production, conventional production, or biodiversity-generating grass strips. Using a two-period model, we introduce a PES in the final period remunerating the grass strips chosen by the farmer. We show that the PES based on additionality distorts the behavior in the initial period, in order to obtain more payment in the final period. The second-best PES based on additionality that takes into account the distortion has to be based on the discounted difference of the marginal environmental benefits obtained in each period. We also establish the second-best level of environmental taxes in the presence of the additionality-based PES. They are no longer equal to the marginal damage and are amended to take into account the distortions caused by the additionality-based PES.

These results are obtained assuming perfect competition in the good markets. The study of Pigouvian taxes in the presence of market power is well documented in the economic literature. Some of the best known works include Buchanan (1969), Barnett (1980) or Ebert (1991). However, few studies have looked at the definition of PES in this context of market power. Krautkraemer and Schwartz (2022) can be mentioned, but they do not take into account the additionality of the PES. However, the assumption of imperfect competition may be interesting for modelling organic farming markets. Data about the distribution of organic farming in France has shown that the development of organic agriculture can be very heterogeneous across a territory (Nguyen-Van et al., 2021). For instance, out of 34259 municipalities in metropolitan France (excluding overseas territories) with at least one farmer, only 418 (1.2%) are 100% organic, and 52.4% of municipalities do not have an organic farmer.<sup>2</sup> The non-uniform distribution of organic farming across the country and transport constraints for organic products could limit competition in the organic product market, resulting in local markets for organic farming where some producers could have market power. Indeed, while we did not find empirical work on the

<sup>&</sup>lt;sup>2</sup>Spatial factors explain the gaps in organic development between territories, such as the quality of the soil (Wollni and Andersson, 2014; Lampach et al., 2020) as well as the geographical organisation of the activity and populations (Ben Arfa et al., 2009) and the presence of many other organic farmers in a geographical unit (Schmidtner et al., 2012; Bjørkhaug and Blekesaune, 2013).

presence of market power in the organic sector, as Sexton (2013) points out in reality probably no agricultural markets are perfect examples of competitive markets, for reasons including the emphasis on dimensions of product quality and differentiation, which could apply to organic products. Thus, it can be interesting to see what happens theoretically to the welfare-maximizing PES payment in the presence of market power.

Our analysis is thus extended assuming market power in the organic market. It turns out that this market power reduces the distortion due to the additionality-based PES in the initial period but also reduces the production quantities in the final period. The second-best level of the additionality-based PES under market power depends on the size of these two effects. The second-best level of environmental taxes are also amended by the market power.

Finally, this paper shows that an additionality-based PES never achieves environmental efficiency, even in a competitive market framework. The PES is set up to correct an environmental distortion but its additional feature is itself the cause of another distortion. Furthermore, this paper provides new insights into understanding the interactions between different environmental policies in the presence of several types of distortions: environmental damages, environmental benefit, additionality and market power.

This article is structured as follows. Section 4.2 provides a review of the literature on additionality in PES schemes. In Section 4.3 we specify the assumptions of our model and analyze a benchmark and first-best scenario. Next, Section 4.4 examines the policy levels of a PES and tax and their resulting production levels, assuming perfect competition. In Section 4.5 we introduce imperfect competition in the organic sector. Finally, Section 4.6 concludes.

# 4.2 Literature review

Sills et al. (2008) describe four challenges to achieving additionality, namely adverse selection, spillovers or leakage, moral hazard, and the possibility that even if there is additionality of a certain land use that is thought to provide certain services, these services may not be additional. Adverse selection occurs when there is hidden information, i.e. the costs that an ES seller faces. Because the ES buyer does not have this information, the ES seller has incentive to say they have higher costs in order to receive a larger payment. Spillover effects or leakage may occur when

preserving some plots of forest leads to increased timber prices, which may incentivize the deforestation of other plots not subject to a PES scheme. Moral hazard, or hidden behavior, occurs when the prospect of a PES scheme getting implemented leads to an ES seller altering their baseline behavior in order to get a higher payment when the PES is in place.

Determining the baseline ES provision for many individual sellers can be quite costly, and Kaczan et al. (2017) look at the possibility of using collective PES schemes to lower this cost. They use a framed field-laboratory experiment with participants from a PES scheme in Mexico and study the impact of conditioning PES payments on an aggregate outcome on group participation and coordination. They found that it was easier to determine baseline and program outcomes for a collective group than for an individual and thus easier to write contracts with additional outcomes. Furthermore, when the PES payments are conditioned on a group's additionality they find that lower contributors raised their contributions.

Pates and Hendricks (2020) frame non-additionality as a moral hazard problem in a technology diffusion context. They look at the case where a new and more environmentally friendly technology becomes less expensive to adopt over time, and whose adoption might be subsidized. They argue that an agent may delay adoption of the technology in order to earn a payment for adoption in a future time period, which is an example of moral hazard since the agent changes his behavior in response to the policy. After developing a conceptual model, the authors run numerical simulations and find that the moral hazard results in a non-monotonic relationship between different policy parameters (e.g. budget or payment size) and the change in technology adoption rates linked to the PES policy (Pates and Hendricks, 2020). Furthermore, they find that the cost-effectiveness of such a policy is lower when the policy is introduced at a time of rapid technology adoption.

Additionality is of utmost importance in carbon offset markets and other carbon sequestration PES schemes. Those paying for carbon offset credits risk paying forest managers to protect forest area that would have remained intact in absence of their payments. Moreover, leakage of the deforestation activities may occur if a forest PES leads to market conditions making it more profitable for forest managers in other regions to cut down more trees, thus leading to a displacement of carbon emissions rather than a net increase in carbon sequestration. Since the objective behind carbon offsets is generally to achieve net zero carbon emissions in order to limit global climate change, the additionality of such a program is crucial. Mason and Plantinga (2013) look into the additionality of conservation contracts, by examining contracts for carbon sequestration from land placed in forest use that serve as offsets to meet emissions reduction goals. In this case, additionality is key to ensuring a reduction in carbon emissions. A government or business seeking to purchase offsets to reduce their emissions will want to minimize expenditures, so paying for forests that would remain without a payment would be wasteful. The authors argue there is an adverse selection problem, as only the agent knows how much land would be placed in forest absent any payment. They propose offering a menu of contracts to induce agents to reveal their type (in terms of high vs. low opportunity cost of placing land in forest). While not a perfect solution, the menu of contracts allows for a reduction in government expenditure compared to a uniform payment. Similarly, Chiroleu-Assouline et al. (2018) undertake a theoretical analysis of additionality of REDD+ contracts, which are made between developed and developing countries with the aim of reducing carbon emissions from deforestation and degradation. Using a principal-agent model, they show that dividing developing countries into two groups based on two different policy instruments can help the developed country obtain efficient deforestation and avoided deforestation levels from their payments.

Others have investigated the empirical evidence of additionality in PES schemes with mixed results. For example, Chabé-Ferret and Subervie (2013) study five agroenvironment schemes implemented in France to estimate their additional and windfall effects. They find different levels of additionality for the different agro-environment schemes, with the more stringent requirements leading to higher additionality. Mezzatesta et al. (2013) use propensity score matching to evaluate the additionality of the Conservation Reserve Program in the US in regard to six conservation practices: conservation tillage, cover crops, hayfield establishment, grid sampling, grass waterways, and filter strips. Based on survey data of farmers in the state of Ohio, they calculate the average treatment effect on the treated (ATT), which they define as the average increase in the proportion of the land adopted in a conservation practice for enrolled farmers relative to their counterfactual proportion of the land in this practice that they would have adopted without funding (Mezzatesta et al., 2013). The authors find that while the overall ATT of the program is positive and statistically significant for each of the conservation practices, the degree of additionality varies across the practices, with hayfield establishment having the highest additionality and conservation tillage the lowest. Jones et al. (2020) look at the additionality of a PES in terms of forest cover and subsequent effects on hydrological services and find that the PES reduces losses but does not provide

many gains in forest cover. Furthermore, they find that a lack of additionality in forest cover due to the PES results in economic loss. Finally, Mohebalian and Aguilar (2016) use GIS data to investigate the additionality of a forest PES program in Ecuador and their findings suggest that the PES program has provided little additionality in terms of preventing deforestation.

# 4.3 The model

In this section, we start by stating the assumptions of our model. Next, we analyze the benchmark situation with no regulation in place. Finally, we investigate the first-best regulation.

#### 4.3.1 Assumptions

In order to analyze the additionality issue, we construct a model with two periods, t = 0, 1. For example, the initial period is before the PES contract, while the final period is during the PES contract. We use  $\beta$  to denote the discount factor. In each period, a representative farmer has three choices for how to manage his land: a conventional crop  $(x_1^t)$ , an organic crop  $(x_2^t)$ , and leaving grass strips  $(y^t)$ . He decides how much of his land to allocate to each management option such that  $x_1^t + x_2^t + y^t = T$  where T is the total area of land in each period. We assume that producing  $x_i^t$  units requires  $x_i^t$  units of land,  $\forall i, \forall t$ .

Each farmer behaves as a price taker in both markets in each time period but we relax this assumption in Section 4.5, where we will consider market power on the organic crop market. The farmer faces production costs, which are assumed to be higher for the organic crop than the conventional crop,  $c_1(x_1^t) < c_2(x_2^t)$ . Both cost functions,  $c_1(x_1^t)$  and  $c_2(x_2^t)$ , are increasing and convex, with  $c'_i(x_i^t) > 0$  and  $c''_i(x_i^t) > 0$ ,  $\forall i = 1, 2$ . Regarding the grass strips,  $y^t$ , the only costs incurred are the foregone profits from not producing. Finally, the inverse demand function for each agricultural product is given by  $p_1^t(x_1^t)$  and  $p_2^t(x_2^t)$  for conventional and organic agriculture, respectively. Demand is linear for both agricultural goods with  $p''_i(x_i^t) < 0, \forall i, \forall t$ .

The different land management options all have different environmental impacts. Conventional agriculture causes pollution, represented by the damage function  $D(x_1^t)$ which is increasing and convex,  $D'(x_1^t) > 0$ ,  $D''(x_1^t) > 0$ . We assume that organic agriculture does not lead to pollution, nor does it increase biodiversity, so it has a neutral impact on the environment. Finally, the grass strips lead to biodiversity benefits, and thus has a positive impact on the environment. The benefit function is represented by  $BF^1(y^0, y^1) = \psi(y^0)^t B(y^1)$ , with  $B'(y^t) > 0$  and  $B''(y^t) < 0$ , and  $\psi'(y^0) > 0$ . This function means that the environmental benefit in the final period depends on the biodiversity level obtained in initial period. We normalize  $BF^0(y^0) = B(y^0)$ . We look at the case where the farmer always chooses a positive level of grass strips, i.e.,  $y^t > 0$ .

In order to take into account negative (environmental damage) and positive (biodiversity benefit) externalities, the regulator can use environmental policies, as environmental taxes denoted by  $t^t$  and PES denoted by  $s^t$  if the PES is implemented in each period and s if the PES is implemented in final period based on additionality.

#### 4.3.2 The benchmark: No regulation

In this section, we analyze the laissez-faire situation, i.e., when there is no environmental policy. As there are two periods with no link between them, we can directly maximize the intertemporal profit:

$$\pi(x_1^0, x_2^0, x_1^1, x_2^1) = p_1^0 x_1^0 + p_2^0 x_2^0 - c_1(x_1^0) - c_2(x_2^0) + \beta \{ p_1^1 x_1^1 + p_2^1 x_2^1 - c_1(x_1^1) - c_2(x_2^1) \}$$

$$(4.1)$$

Maximizing this function yields typical first order conditions that price should equal marginal cost for  $x_i^t, \forall i, \forall t$ :

$$p_i^t - c_i'(\bar{x}_i^t) = 0 \ \forall i, \forall t$$

In each case the quantities of conventional and organic agriculture production are such that the price is equal to the private marginal costs. This equilibrium is not efficient because environmental externalities are not taken into account.

#### 4.3.3 First-best regulation

In this section, we consider a social planner who decides on first-best quantities for each production. He maximizes social welfare, taking into account the farmer's profits, consumer surplus, and environmental damages and benefits.

$$W(x_1^0, x_2^0, x_1^1, x_2^1) = \int_0^{x_1^0} p_1^0(u) du + \int_0^{x_2^0} p_2^0(v) dv - c_1(x_1^0) - c_2(x_2^0) + B(T - x_1^0 - x_2^0) - D(x_1^0) + \beta \left\{ \int_0^{x_1^1} p_1^1(w) dw + \int_0^{x_2^1} p_2^1(z) dz - c_1(x_1^1) - c_2(x_2^1) + \psi(y^0) B(T - x_1^1 - x_2^1) - D(x_1^1) \right\}$$

Taking the first order conditions we obtain:

$$\frac{\partial W}{\partial x_1^0} = p_1^0(x_1^{0*}) - c_1'(x_1^{0*}) - B_{y^{0*}} - \beta \psi'(y^{0*})B(y^{1*}) - D'(x_1^{0*}) = 0$$
(4.2)

$$\frac{\partial W}{\partial x_2^0} = p_2^0(x_2^{0*}) - c_2'(x_2^{0*}) - B_{y^{0*}} - \beta \psi'(y^{0*})B(y^{1*}) = 0$$
(4.3)

$$\frac{\partial W}{\partial x_1^1} = \beta \left[ p_1^1(x_1^{1*}) - c_1'(x_1^{1*}) - \psi(y^{0*}) B_{y^{1*}} - D'(x_1^{1*}) \right] = 0$$
(4.4)

$$\frac{\partial W}{\partial x_2^1} = \beta \left[ p_2^1(x_2^{1*}) - c_2'(x_2^{1*}) - \psi(y^{0*}) B_{y^{1*}} \right] = 0$$
(4.5)

In the first-best scenario, the socially optimal levels of conventional and organic agriculture in both time periods occur taking into account social marginal cost of production. The level of conventional agriculture is based on the private marginal cost of production, the marginal biodiversity benefit and marginal damage. Similarly, the level of organic production is based on marginal cost and marginal biodiversity benefits. As the level of biodiversity achieved in period 0 positively affects the biodiversity of period 1, it appears that the decision to create grass strips in the initial period generates a marginal biodiversity benefit in both periods, given by  $[B_{y^0} + \beta \psi'(y^0)B(y^1)].$ 

Comparing the first-best equations and the benchmark, we easily identify firstbest environmental policy in each period:  $t^{0*} = D'(x_1^{0*}); t^{1*} = D'(x_1^{1*}); s^0 = B_{y^o} + \beta \psi'(y^{0*}) B(y^{1*}); s^1 = \psi(y^{0*}) B_{y^{1*}}$ . The first-best allocation must therefore be established by setting environmental taxes and a PES in each period. Each environmental tax should correspond to the environmental damage and each PES to the full marginal environmental benefit. However, the budgetary constraint<sup>3</sup> may lead the regulator to integrate the principle of additionality in the PES, by remunerating only the environmental benefits induced by the PES.

# 4.4 Additionality and perfect competition

We assume a regulator wishes to implement the principle of additionality by using a PES in the final period that remunerates the additional environmental benefits generated by the PES between the initial and final periods. The farmer is aware of the PES policy implementation and is able to adjust his initial period quantities beforehand. The regulator introduces an environmental tax on conventional production to correct for the environmental damages in each period. In order to investigate the efficiency of a PES based on additionality, we first analyze the farmer's behavior with environmental policies. Then we identify the second-best level of the environmental tax in each period and of the PES based on additionality. Finally, we get the levels of production and thus of environmental damage and benefits.

#### 4.4.1 Strategic behaviors

In the initial period, the regulator sets an environmental tax in both periods  $(t^0$  and  $t^1)$  and announces that a PES will be implemented in the final period (s) on the additional grass strip area compared to the initial period. In order to obtain optimal quantities produced in each period, we use backward induction. We first define the subgame-perfect Nash equilibrium obtained in the second stage. Then, we solve quantities produced in the initial period. We can anticipate strategic behaviors.

#### 4.4.1.1 The second stage: equilibrium quantities in the final period

In the final period, the PES is introduced, remunerating only the additional grass strip area compared to the initial period. This quantity is equal to  $[y^1 - y^0]$ 

where 
$$\begin{cases} y^1 - y^0 = T - x_1^1 - x_2^1 - y^0, \\ y^0 = T - x_1^0 - x_2^0. \end{cases}$$

We maximize the profit in the final period in order to define the subgame-perfect

<sup>&</sup>lt;sup>3</sup>One possibility would be to try to finance the PES payment with the environmental tax. However, when trying to include this in the model it became intractable.

Nash equilibrium in that period:

$$\pi^{1}(x_{1}^{1}, x_{2}^{1}) = p_{1}^{1}x_{1}^{1} + p_{2}^{1}x_{2}^{1} - c_{1}(x_{1}^{1}) - c_{2}(x_{2}^{1}) - t^{1}x_{1}^{1} + s(-x_{1}^{1} - x_{2}^{1} + x_{1}^{0} + x_{2}^{0})$$

Calculating the first order conditions, we find:

$$\frac{\partial \pi^1}{\partial x_1^1} = p_1^1 - c_1'(x_1^{1c}) - t^1 - s = 0$$
(4.6)

$$\frac{\partial \pi^1}{\partial x_2^1} = p_2^1 - c_2'(x_2^{1c}) - s = 0 \tag{4.7}$$

Solving these FOC, we find the equilibrium quantities in the final period:  $x_1^{1c}(t^1, s)$ ;  $x_2^{1c}(s)$ . Applying the implicit function theorem on (4.6) and (4.7) we can investigate how the production levels change in response to the environmental policies. We obtain:

$$\begin{array}{rcl} \displaystyle \frac{\partial x_1^{1c}}{\partial s} & = & \displaystyle -\frac{1}{c_1''(x_1^1)} < 0 \\ \\ \displaystyle \frac{\partial x_1^{1c}}{\partial t^1} & = & \displaystyle -\frac{1}{c_1''(x_1^1)} < 0 \\ \\ \displaystyle \frac{d x_2^{1c}}{d s} & = & \displaystyle -\frac{1}{c_2''(x_2^1)} < 0 \end{array}$$

In conformity with intuition, the environmental tax decreases conventional production in the final time period and the PES decreases both agriculture productions in the final time period.

#### 4.4.1.2 The first-stage: equilibrium quantities in the initial period

In order to obtain the equilibrium quantities in the initial period, the farmer maximizes his intertemporal profit. We use equilibrium quantities from the final period in the profit function,  $x_1^{1c}(t^1, s)$ ;  $x_2^{1c}(s)$ . The intertemporal profit is:

$$\begin{aligned} \pi(x_1^0, x_2^0) &= p_1^0 x_1^0 + p_2^0 x_2^0 - c_1(x_1^0) - c_2(x_2^0) - t^0 x_1^0 \\ &+ \beta \{ p_1^1 x_1^1(t^1, s) + p_2^1 x_2^1(s) - c_1(x_1^1(t^1, s)) - c_2(x_2^1(s)) - t^1 x_1^1(t^1, s) + s(y^1 - y^0) \} \end{aligned}$$

$$\frac{\partial \pi}{\partial x_1^0} = p_1^0 - c_1'(x_1^{0c}) - t^0 + \beta s = 0$$
(4.8)

$$\frac{\partial \pi}{\partial x_2^0} = p_2^0 - c_2'(x_2^{0c}) + \beta s = 0 \tag{4.9}$$

The farmer accounts for the environmental tax in the initial period as well as the PES based on additionality when deciding how to allocate his land in the initial time period. From the first-order conditions we find:  $x_1^{0c}(s, t^0)$ ;  $x_2^{0c}(s)$ . We can then apply the implicit function theorem on (4.8) and (4.9) to see how production levels change in response to the environmental policies. We find:

$$\begin{array}{rcl} \frac{\partial x_1^{0c}}{\partial s} & = & \frac{\beta}{c_1''(x_1^0)} > 0 \\ \frac{\partial x_1^{0c}}{\partial t^0} & = & -\frac{1}{c_1''(x_1^0)} < 0 \\ \frac{d x_2^{0c}}{d s} & = & \frac{\beta}{c_2''(x_2^0)} > 0 \end{array}$$

While the environmental tax in the initial period reduces the level of conventional production, the PES based on additionality raises both conventional and organic production levels in the initial period. The farmer adopts a strategic behavior in order to capture more payment from the PES in the final period. He distorts the basis for calculating the PES to his advantage.

**Proposition 1.** The additional PES creates a strategic behavior in the initial period, leading to less environmental benefit in the initial period.

The organic production level is still increased in the initial period as a result of the PES policy. The conventional production level is subject to two effects: it increases with the PES but decreases with the tax. To see the net change in conventional production level, we have to see whether the effect of the tax or the PES will be larger:

$$\frac{\beta}{c_1''(x_1^0)} - \frac{1}{c_1''(x_1^0)} = \frac{\beta - 1}{c_1''(x_1^0)} < 0$$

Since  $0 < \beta < 1$ , the direct effect of the tax will be greater than the indirect effect

of the PES, so the net effect will be a decrease in conventional production in the initial period. However, the conventional production level would have decreased more without the additionality requirement of the PES.

#### 4.4.2 Tax and PES designs

In this section we define the second-best level of the PES based on additionality and of the environmental taxes. As the additionality-based PES increases agricultural production levels in the first period, it is expected to have a negative effect on the welfare level. The second-best value of environmental policies has to correct this distortion. In order to design these policies, we replace  $x_1^{1c}(t^1, s)$ ,  $x_2^{1c}(s)$ ,  $x_1^{0c}(s, t^0)$ , and  $x_2^{0c}(s)$  in the welfare function. The regulator maximizes the welfare function with respect to the additionality-based PES and environmental taxes in each period. Looking at the intertemporal welfare we have:

$$W(x_1^0(s,t^0), x_2^0(s), x_1^1(s,t^1), x_2^1(s)) = \int_0^{x_1^0(s,t^0)} p_1(u) du + \int_0^{x_2^0(s)} p_2(v) dv - c_1(x_1^0(s,t^0)) - c_2(x_2^0(s)) + B(T - x_1^0(s,t^0) - x_2^0(s)) - D(x_1^0(s,t^0)) + \beta \left\{ \int_0^{x_1^1(s,t^1)} p_1(w) dw + \int_0^{x_2^1(s)} p_2(z) dz - c_1(x_1^1(s,t^1)) - c_2(x_2^1(s)) + \psi(T - x_1^0(s,t^0) - x_2^0(s)) B(T - x_1^1(s,t^1) - x_2^1(s)) - D(x_1^1(s,t^1)) \right\}$$

Taking the first order conditions we obtain:

$$\frac{\partial W}{\partial t^0} = \frac{\partial x_1^0}{\partial t^0} \left[ p_1^0 - c_1'(x_1^0) - B_{y^0} - \beta \psi'(y^0) B(y^1) - D'(x_1^0) \right] = 0$$
(4.10)

$$\frac{\partial W}{\partial t^1} = \frac{\partial x_1^1}{\partial t} \beta \left[ p_1^1 - c_1'(x_1^1) - \psi(y^0) B_{y^1} - D'(x_1^1) \right] = 0$$
(4.11)

$$\frac{\partial W}{\partial s} = \frac{\partial x_1^1}{\partial s} \beta \left[ p_1^1 - c_1'(x_1^1) - \psi(y^0) B_{y^1} - D'(x_1^1) \right] + \frac{dx_2^1}{ds} \beta \left[ p_2^1 - c_2'(x_2^1) - \psi(y^0) B_{y^1} \right] 
+ \frac{\partial x_1^0}{\partial s} \left[ p_1^0 - c_1'(x_1^0) - B_{y^0} - \beta \psi'(y^0) B(y^1) - D'(x_1^0) \right] 
+ \frac{dx_2^0}{ds} \left[ p_2^0 - c_2'(x_2^0) - B_{y^0} - \beta \psi'(y^0) B(y^1) \right] = 0$$
(4.12)

We find the following expression for the second-best additionality-based PES payment (see Appendix A in section 4.7 for calculations):

$$s = \frac{\frac{1}{c_2''(x_2^1)}\psi(y^0)B_{y^1} - \frac{1}{c_2''(x_2^0)}[B_{y^0} + \beta\psi'(y^0)B(y^1)]}{[\frac{1}{c_2''(x_2^1)} + \frac{\beta}{c_2''(x_2^0)}]}$$
(4.13)

The PES depends on the values of the marginal benefit in each period, as well as the production costs in each period. In the following, for ease of reading, we consider the particular case where c'' is constant, such as with quadratic equations, which gives:

$$s^{c} = \frac{\psi(y^{0c})B_{y^{1c}} - (B_{y^{0c}} + \beta\psi'(y^{0c})B(y^{1c}))}{1 + \beta}$$
(4.14)

The second-best PES based on additionality is equal, in this case, to the discounted difference between the marginal environmental benefit in the final period given by  $[\psi(y^0)B_{y^1}]$  and the marginal environmental benefit from the initial period  $[B_{y^0} + \beta\psi'(y^0)B(y^1)]$ . The latter is composed of the direct effect in the initial period, and the indirect effect of the initial grass strip area on the benefits in the final period. This is in contrast to actual AES contracts, which base payments off of foregone profits from adopting environmental practices. We can obtain conditions on the positivity of the PES:

$$s^{c} > 0 \Leftrightarrow \psi(y^{0c})B_{y^{1c}} > B_{y^{0c}} + \beta \psi'(y^{0c})B(y^{1c})$$

**Proposition 2.** The PES based on additionality is positive if it leads to a greater marginal environmental benefit in the final period compared to the initial period.

Since the PES reduces agricultural production quantities in the final period, it also generates biodiversity benefits in this period. This is the positive effect of the PES. However, the PES, based on additionality increases the agriculture production quantities in the initial period, leading to a decrease in the biodiversity benefits that could be obtained in both periods. The PES that accounts for these strategic behaviors increases proportionally to an increase in biodiversity benefits obtained in the final period. By setting the value of the PES based on the additional benefits obtained in terms of biodiversity, the regulator partly counteracts the disincentive in the initial period induced by the PES.

There will only be a payment if the PES leads to an additional effect in terms of biodiversity. If the marginal benefits of biodiversity are equal in both periods, there will be no PES. If the marginal benefits are greater in the initial period than in the final period, the PES can be negative: the regulator will seek to tax the grass strips in the final period in order to have more in the initial period, resulting in an additional marginal benefit.

Next, we use the value of the PES to determine the levels of each tax (see Appendix A in Section 4.7 for full calculations and the general case). Starting with the tax in the initial period we obtain:

$$t^{0c}(y^{0c}, y^{1c}) = t^{0*} + \frac{B_{y^{0c}} + \beta[\psi'(y^{0c})B(y^{1c}) + \psi(y^{0c})B_{y^{1c}}]}{1 + \beta}$$
(4.15)

Then, for the final period tax we find:

$$t^{1c}(y^{0c}, y^{1c}) = t^{1*} + \frac{B_{y^{0c}} + \beta[\psi'(y^{0c})B(y^{1c}) + \psi(y^{0c})B_{y^{1c}}]}{1+\beta}$$
(4.16)

Both environmental taxes are equal to their respective marginal damages, with an additional term,  $\frac{B_{y^0}+\beta[\psi'(y^0)B(y^1)+\psi(y^0)B_{y^1}]}{1+\beta} > 0$ , which represents the net present value of biodiversity benefits obtained due to the PES. Both taxes will increase proportional to the net present value of biodiversity benefits. The tax is used to focus behaviors where we obtain the most biodiversity benefits.

**Proposition 3.** In the presence of a PES based on additionality, environmental taxes are no longer equal to the marginal damage. They must take into account the distortions due to the additionality condition of the PES.

Comparing the levels of environmental policies against their first-best levels, we see that the PES in the initial period is zero, and is therefore too low compared to the first-best. The PES in the second period is also lower than its first-best level. To restore the correct production quantities, the regulator will adjust the amount of environmental taxes, which do not distort the market, unlike the additionality-based PES. Thus, the regulator will use the tax in the initial period to obtain a better level of grass strips. By increasing the tax, he partly bypasses the poor incentive of the PES on the conventional agricultural market. In the final period, since the additional PES is too low compared to its first-best level, the regulator also increases the tax to reduce the level of conventional agriculture and thus obtain more grass strips.

#### 4.4.3 Calculated quantities

We now calculate the levels of conventional and organic agriculture that will result from the policies. We take the equations (4.14), (4.15) and (4.16), and plug these into the profit FOCs (4.6), (4.7), (4.8), and (4.9). Next, we solve for the quantities of organic and conventional agriculture in both periods and compare these to the quantities from the first-best scenario (see Appendix A in Section 4.7 for calculations).

In the general case, the quantities chosen are not equal to the first-best quantities. The environmental taxes and the PES set by the regulator do not achieve the first-best. This comes from the following channel: the PES which is initially set up to correct an environmental distortion induces another distortion when it is based on additionality. The introduction of the additionality principle implements only one PES in the final period instead of a PES in each period. The second-best level of environmental policies seeks to counteract strategic behavior on the basis of the environmental benefits achieved. Since the taxes cannot indirectly correct for the distorted behavior induced in the initial period by the PES, the production quantities never match the first-best. In the end, PES based on additionality does not achieve environmental efficiency.

# 4.5 Additionality and imperfect competition

We now add the assumption that the market for the organic agricultural good is imperfectly competitive, while keeping the conventional market perfectly competitive. We are assuming that the conventional good is one that is traded on the global market (e.g. wheat) while the organic good is one that is traded locally, is difficult to transport long distances, and is not taking prices from the global market (e.g. baby leaf lettuce). We seek to determine the implications of the additionality condition of the PES in a context of imperfect competition. After analyzing the behavior of firms in response to the environmental policies, we define the second-best level of environmental taxes and PES. Finally, we calculate production levels.

#### 4.5.1 Monopoly: Strategic behaviors

In this subsection, we examine the case where imperfect competition in the organic sector takes the form of a monopoly. We assume environmental taxes in both periods and a PES based on additionality in the final period. We investigate, in this context, the farmer's behavior.

#### 4.5.1.1 The second stage: equilibrium quantities in the final period

Let us define the subgame-perfect Nash equilibrium in the final period. We use backward induction in order to define production quantities in final period. We maximize the profit function:

$$\pi^{1}(x_{1}^{1}, x_{2}^{1}) = p_{1}^{1}x_{1}^{1} + p_{2}^{1}(x_{2}^{1})x_{2}^{1} - c_{1}(x_{1}^{1}) - c_{2}(x_{2}^{1}) - t^{1}x_{1}^{1} + s(y^{1} - y^{0})$$
where
$$\begin{cases} y^{1} - y^{0} = T - x_{1}^{1} - x_{2}^{1} - y^{0}, \\ y^{0} = T - x_{1}^{0} - x_{2}^{0}. \end{cases}$$

First order conditions are the following:

$$\frac{\partial \pi^1}{\partial x_1^1} = p_1^1 - c_1'(x_1^{1m}) - t^1 - s = 0$$
(4.17)

$$\frac{\partial \pi^1}{\partial x_2^1} = p_2^{1\prime}(x_2^{1m})x_2^{1m} + p_2^1(x_2^{1m}) - c_2'(x_2^{1m}) - s = 0$$
(4.18)

Solving these FOC, we find the equilibrium quantities in the second time period:  $x_1^{1m}(t^1, s)$ ;  $x_2^{1m}(s)$ . The market power decreases the organic production level, as the farmer considers the marginal revenue rather than the price when making his land allocation decision. Applying the implicit function theorem on (4.17) and (4.18), we can investigate how the environmental policies affects the production quantities. We find:

$$\begin{array}{rcl} \displaystyle \frac{\partial x_1^{1m}}{\partial s} & = & \displaystyle -\frac{1}{c_1''(x_1^1)} < 0 \\ \\ \displaystyle \frac{\partial x_1^{1m}}{\partial t^1} & = & \displaystyle -\frac{1}{c_1''(x_1^1)} < 0 \\ \\ \displaystyle \frac{d x_2^{1m}}{ds} & = & \displaystyle \frac{1}{2p_2'(x_2^1) - c_2''(x_2^1)} < 0 \end{array}$$

The environmental policies have the expected effect and reduce the levels of production. But the organic production level is reduced by the market power and the PES. So there are more grass strips in final period. The environmental benefits are therefore increased.

#### 4.5.1.2 The first stage: equilibrium quantities in the initial period

In order to obtain the equilibrium quantities in the initial period, we use equilibrium quantities from the final period,  $x_1^{1m}(t^1, s)$ ;  $x_2^{1m}(s)$  in the farmer's intertemporal profit function:

$$\pi(x_1^0, x_2^0) = p_1^0 x_1^0 + p_2^0 (x_2^0) x_2^0 - c_1(x_1^0) - c_2(x_2^0) - t^0 x_1^0 + \beta \{ p_1^1 x_1^{1m} + p_2^1 (x_2^{1m}) x_2^{1m} - c_1(x_1^{1m}) - c_2(x_2^{1m}) - t^1 x_1^{1m} + s(y^{1m} - y^0) \}$$

Maximizing the profit function gives the following first order conditions:

$$\frac{\partial \pi}{\partial x_1^0} = p_1^0 - c_1'(x_1^{0m}) - t^0 + \beta s = 0$$
(4.19)

$$\frac{\partial \pi}{\partial x_2^0} = p_2^{0\prime}(x_2^{0m})x_2^{0m} + p_2^0(x_2^{0m}) - c_2'(x_2^{0m}) + \beta s = 0$$
(4.20)

The farmer makes his land allocation decision by taking into account the environmental tax in the initial period and the PES. His market power on the organic market leads him to consider his marginal revenue when deciding his organic production quantity instead of the price, which results in a lower organic production quantity. From the FOC, we find:  $x_1^{0m}(s, t^0)$ ;  $x_2^{0m}(s)$ . Applying the implicit function theorem on (4.19) and (4.20), we analyze how the environmental policies affect the production quantities:

$$\begin{array}{lll} \displaystyle \frac{\partial x_1^{0m}}{\partial s} & = & -\frac{\frac{\partial J}{\partial s}}{\frac{\partial J}{\partial x_1^0}} = \frac{\beta}{c_1''(x_1^0)} > 0\\ \displaystyle \frac{\partial x_1^{0m}}{\partial t^0} & = & -\frac{\frac{\partial J}{\partial t^0}}{\frac{\partial J}{\partial x_1^0}} = -\frac{1}{c_1''(x_1^0)} < 0\\ \displaystyle \frac{dx_2^{0m}}{ds} & = & -\frac{\frac{\partial K}{\partial s}}{\frac{\partial K}{\partial x_2^0}} = -\frac{\beta}{2p_2'(x_2^0) - c_2''(x_2^0)} > 0 \end{array}$$

The PES implemented in the final period creates a distortion in the initial period. Farmers increase their production levels in the initial period in order to benefit from more PES in the final period. In the initial period, the organic market is subject to two distortions: strategic behavior following the additionality-based PES and market power. Comparing  $\frac{dx_2^0}{ds}$  with market power and without market power, we obtain the following proposition:

**Proposition 4.** Market power in the organic market reduces the strategic behavior introduced by the additional PES.

The incentive to change the baseline level of grass strip upon which payment is based by increasing the quantity produced of the organic good is in contrast to pricemaking behavior, which leads to a reduction in the quantity produced. Thus, market power reduces strategic behavior in the organic market relative to the competitive situation. In the organic market, the distortion induced by market power partly offsets the distortion induced by the PES.

#### 4.5.2 Tax and PES designs

We will now define the second-best environmental policies. Let's assume  $s = -\frac{p_2^{0'}(x_2^{0m})x_2^{0m}}{\beta}$ . From (4.20), we see the appropriate incentives would be given by the regulator in the organic market in initial period. However, this PES level would not achieve the optimal market behaviour of conventional agriculture in the same period. There is also no reason why this level of PES should implement the right quantities in final period. It is therefore necessary to maximize the intertemporal welfare function in order to obtain the levels of PES and environmental taxes that result from these different trade-offs.

We obtain the PES value,  $s^m$  (see Appendix B in Section 4.7 for calculations and

the general case):

$$s^{m} = s^{*}(y^{0m}, y^{1m}) + \frac{p_{2}'(x_{2}^{1m})x_{2}^{1m} - p_{2}'(x_{2}^{0m})x_{2}^{0m}}{1 + \beta}$$
(4.21)

The second-best PES is equal to the net present value of the difference in marginal benefits, adjusted for the market power. In each period, the adjusted marginal benefit is lower than the marginal benefit without market power in each period because p' < 0. We can identify two terms. The first is similar to PES without market power valued at quantities with market power. The second is the is a market power term.

The positivity of the PES now depends on the difference in marginal benefits and also on the market power term:

$$s^m > 0 \Rightarrow \psi(y^{0m})B_{y^{1m}} - [B_{y^{0m}} + \beta\psi'(y^{0m})B(y^{1m}] > [p_2'(x_2^{0m})x_2^{0m} - p_2'(x_2^{1m})x_2^{1m}]$$

The effects of the market power are different from one period to the next. In the final period, the market power reduces the organic production level, having an effect we can qualify as negative. In the initial period, it limits the strategic behavior on the organic market, which is a positive effect. The additionality-based PES that takes into account the market power, (equation 4.21), summarizes these effects. If the negative effect in the final period is substantial, the PES with market power will be lower than without market power. Conversely, if the effect in the initial period is high, the PES with market power will be larger. As the PES remunerates the supplementary environmental benefits, one can expect that the organic production quantity will be lower in the final period, i.e.  $x_2^{1m} < x_2^{0m}$ . In this case, the market power will increase the PES.

Next, we use the value of the PES to determine the levels of each tax. We obtain (see Appendix B in Section 4.7 for calculations and the general case):

$$t^{0m}(y^{0m}, y^{1m}) = t^{0c} + \frac{\beta [p_2'(x_2^{1m})x_2^{1m} - p_2'(x_2^{0m})x_2^{0m}]}{1 + \beta}$$
(4.22)

$$t^{1m}(y^{0m}, y^{1m}) = t^{1c} + \frac{-p_2'(x_2^{1m})x_2^{1m} + p_2'(x_2^{0m})x_2^{0m}}{1+\beta}$$
(4.23)

As under perfect competition, taxes depends on the marginal damage and the net present value of biodiversity benefits. This time, they also include a term that takes into account the market power in the organic sector. If the market power increases (decreases) the PES, the tax in the initial period increases (decreases) but the tax in the final period decreases (increases). The environmental taxes are adjusted to take into account the indirect effects of market power on the conventional agriculture market.

#### 4.5.3 Calculated quantities

We now seek to calculate the production levels of conventional and organic agriculture that will result from the policies. We take equations (4.21), (4.22), and (4.23) and plug them into the profit FOCs (4.17), (4.18), (4.19), and (4.20). Next, we solve for the quantities of organic and conventional agriculture in both periods and compare these to the quantities from the first-best scenario (see Appendix B in Section 4.7 for calculations).

The quantities chosen are not equal to the first-best quantities. The second-best environmental taxes and the PES set by the regulator do not achieve the firstbest. They fail to take into account several distortions: environmental damages, environmental benefits, additionality and market power.

# 4.6 Conclusion

When program budgets are limited, ES buyers want to ensure their payments will lead to an increase in the overall level of ES provision. This is why PES can be based on the principle of additionality. The question is whether these additionality-based PES achieve environmental efficiency. To conduct this study, we used a model with two time periods and we considered the representative farmer's behavior. He allocates his land between organic agricultural production that has a neutral impact on the environment, conventional production that causes environmental damage, and leaving grass strips that generate environmental benefits. The regulator sets a PES as an incentive for the farmer to leave grass strips but only wants to pay for the additional environmental benefits that result from the PES. We show that this PES based on additionality distorts the farmer's behavior in the initial period. The farmer increases his production levels in order to obtain more payment in the final period. The second-best PES has to correct this distortion while taking into account the environmental benefits and damages. In the end, the second-best PES is equal to the discounted difference of the marginal environmental benefit in each period. This is in contrast to actual schemes which allocate payments based on foregone profits from adopting environmental practices. The second-best environmental taxes in each period are no longer equal to the marginal damage. They are adjusted to correct the distortions induced by the PES.

We then introduced market power in the organic market. If the market power reduces the organic production quantity in the final period, it limits the distortion in the initial period. Depending on the size of these effects, the second-best additionalitybased PES either increases or decreases compared to the scenario without market power. The taxes are adjusted to take into account the indirect effects of the market power on the level of conventional production.

Finally, this study has shown that the additionality condition of the PES does not achieve environmental efficiency, even under perfect competition. It also provides a better understanding of the interactions between different types of environmental policies.

It turns out that basing the PES on the additional environmental benefits obtained by the payment is not very easy to characterize in a simple perfect information setting with two time periods. However this study could be extended by considering an infinite time horizon, in order to see how the results hold up. This would mean modeling a strategic behavior in an optimal control model. In the same vein as Barnett (1980), the second-best environmental policies are defined under perfect information, which suggests that the regulator knows the firm's production costs. By mobilizing agency theory, these works could be extended under asymmetric information, which would allow for taking moral hazard into account. Finally, the model could be enriched by introducing the marginal social cost of public funds.

# 4.7 Appendices of Chapter 4

## Appendix A. The second-best environmental policies under perfect competition

#### Determination of $s^c$

From the profit FOCs given by (4.6), (4.7), (4.8), and (4.9), we find:

$$p_1^1 - c_1'(x_1^{1c}) = t^1 + s$$

$$p_2^1 - c_2'(x_2^{1c}) = s$$

$$p_1^0 - c_1'(x_1^0) = t^0 - \beta s$$

$$p_2^0 - c_2'(x_2^0) = -\beta s$$

Plugging these into (4.10), (4.11), and (4.12), we obtain:

$$\frac{\partial x_1^0}{\partial t^0} \left[ t^0 - \beta s - B_{y^0} - \beta \psi'(y^0) B(y^1) - D'(x_1^0) \right] = 0$$
(4.24)

$$\frac{\partial x_1^1}{\partial t^1} \beta \left[ t^1 + s - \psi(y^0) B_{y^1} - D'(x_1^1) \right] = 0 \qquad (4.25)$$

$$\frac{\partial x_1^1}{\partial s} \beta \left[ t^1 + s - \psi(y^0) B_{y^1} - D'(x_1^1) \right] + \frac{d x_2^1}{d s} \beta \left[ s - \psi(y^0) B_{y^1} \right] + \frac{\partial x_1^0}{\partial s} \left[ t^0 - \beta s - B_{y^0} - \beta \psi'(y^0) B(y^1) - D'(x_1^0) \right] + \frac{d x_2^0}{d s} \left[ -\beta s - B_{y^0} - \beta \psi(y^0) B_{y^1} \right] = 0$$
(4.26)

We can then solve (4.24) and (4.25) for  $t^0$  and  $t^1$ :

$$t^{0} = \beta s + B_{y^{0}} + \beta \psi'(y^{0})B(y^{1}) + D'(x_{1}^{0})$$
(4.27)

$$t^{1} = -s + \psi(y^{0})B_{y^{1}} + D'(x_{1}^{1})$$
(4.28)

We plug these values into (4.26):

$$\frac{dx_2^1}{ds}\beta\left[s - \psi(y^0)B_{y^1}\right] + \frac{dx_2^0}{ds}\left[-\beta s - B_{y^0} - \beta\psi'(y^0)B(y^1)\right] = 0$$

$$s\left[\frac{dx_2^1}{ds}\beta - \frac{dx_2^0}{ds}\beta\right] = \frac{dx_2^1}{ds}\beta\psi(y^0)B_{y^1} + \frac{dx_2^0}{ds}[B_{y^0} + \beta\psi'(y^0)B(y^1)]$$

$$s = \frac{\frac{dx_2^1}{ds}\beta\psi(y^0)B_{y^1} + \frac{dx_2^0}{ds}[B_{y^0} + \beta\psi'(y^0)B(y^1)]}{\beta\left[\frac{dx_2^1}{ds} - \frac{dx_2^0}{ds}\right]}$$

We then substitute in  $\frac{dx_2^1}{ds} = -\frac{1}{c_2''(x_2^1)}$ , and  $\frac{dx_2^0}{ds} = \frac{\beta}{c_2''(x_2^0)}$ :

$$s = \frac{\frac{1}{c_2''(x_2^1)}\psi(y^0)B_{y^1} - \frac{1}{c_2''(x_2^0)}[B_{y^0} + \beta\psi'(y^0)B(y^1)]}{\frac{1}{c_2''(x_2^1)} + \frac{\beta}{c_2''(x_2^0)}}$$
(4.29)

After assuming a constant c'', we obtain equation (4.14).

#### Determination of $t^{0c}$

Replacing s in (4.27), we have:

$$t^{0c} = \beta \left[ \frac{\frac{1}{c_2''(x_2^1)} \psi(y^0) B_{y^1} - \frac{1}{c_2''(x_2^0)} [B_{y^0} + \beta \psi'(y^0) B(y^1)]}{\frac{1}{c_2''(x_2^1)} + \frac{\beta}{c_2''(x_2^0)}} \right] + B_{y^0} + \beta \psi'(y^0) B(y^1) + D'(x_1^0)$$

Simplifying:

$$t^{0c} = \frac{\frac{(1-\beta)(B_{y^0} + \beta\psi'(y^0)B(y^1)) + \beta\psi'(y^0)B(y^1))}{c_2''(x_2^1)} + \frac{\beta[B_{y^0} + \beta\psi'(y^0)B(y^1))]}{c_2''(x_2^0)}}{\frac{1}{c_2''(x_2^1)} + \frac{\beta}{c_2''(x_2^0)}} + D'(x_1^0)$$

After rearranging and taking the case where  $c_2''$  is constant, we obtain Equation (4.15).

#### Determination of $t^{1c}$

Replacing s in (4.28), we have:

$$t^{1} = -\left\{\frac{\frac{1}{c_{2}''(x_{2}^{1})}\psi(y^{0})B_{y^{1}} - \frac{1}{c_{2}''(x_{2}^{0})}[B_{y^{0}} + \beta\psi'(y^{0})B(y^{1})]}{\frac{1}{c_{2}''(x_{2}^{1})} + \frac{\beta}{c_{2}''(x_{2}^{0})}}\right\} + \psi(y^{0})B_{y^{1}} + D'(x_{1}^{1})$$

Simplifying:

$$t^{1} = D'(x_{1}^{1}) + \frac{\frac{1}{c_{2}'(x_{2}^{0})} \left[ B_{y^{0}} + \beta \psi'(y^{0}) B(y^{1}) + \psi(y^{0}) B_{y^{1}} \right]}{\frac{1}{c_{2}''(x_{2}^{1})} + \frac{\beta}{c_{2}''(x_{2}^{0})}}$$

After rearranging and taking the case where  $c_2''$  is constant, we obtain Equation (4.16).

#### Determination of calculated quantities

As with  $y^{1c}(x_1^{1c}, x_2^{1c})$  and  $y^{0c}(x_1^{0c}, x_2^{0c})$ , quantities  $(x_1^{1c}, x_2^{1c}, x_1^{0c}, x_2^{0c})$  are obtained solving the following system:

$$p_{1} - c_{1}'(x_{1}^{0c}) - D'(x_{1}^{1c}) - B_{y^{0c}} - \beta \psi'(y^{0c})B(y^{1c}) = 0$$

$$p_{2} - c_{2}'(x_{2}^{1c}) - \frac{\psi(y^{0c})B_{y^{1c}} - (B_{y^{0c}} + \beta \psi'(y^{0c})B(y^{1c}))}{1 + \beta} = 0$$

$$p_{1} - c_{1}'(x_{1}^{1c}) - \psi(y^{0c})B_{y^{1c}} - D'(x_{1}^{1c}) = 0$$

$$p_{2} - c_{2}'(x_{2}^{0c}) + \frac{\beta}{1 + \beta} \left(\psi(y^{0c})B_{y^{1c}} - B_{y^{0c}} - \beta \psi'(y^{0c})B(y^{1c})\right) = 0$$

# Appendix B. The second-best environmental policies under imperfect competition

#### Welfare equation with market power

After having substituted in the production quantities  $(x_1^{0m}, x_2^{0m}, x_1^{1m}, x_2^{1m})$  that depend on the environmental policies, we obtain the following intertemporal welfare function:

$$W(x_1^0(s,t^0), x_{2(s)}^0(s), x_1^1(s,t^1), x_2^1(s)) = \int_0^{x_1^0(s,t^0)} p_1(u) du + \int_0^{x_2^0(s)} p_2(v) dv - c_1(x_1^0(s,t^0)) - c_2(x_2^0(s))$$
  
+  $B(T - x_1^0(s,t^0) - x_2^0(s)) - D(x_1^0(s,t^0)) + \beta \left\{ \int_0^{x_1^1(s,t^1)} p_1(w) dw + \int_0^{x_2^1(s)} p_2(z) dz - c_1(x_1^1(s,t^1)) - c_2(x_2^1(s)) + \psi(T - x_1^0(s,t^0) - x_2^0(s)) B(T - x_1^1(s,t^1) - x_2^1(s)) - D(x_1^1(s,t^1)) \right\}$ 

The first order conditions are the following:

$$\frac{\partial W}{\partial t^0} = \frac{\partial x_1^0}{\partial t^0} \left[ p_1 - c_1'(x_1^0) - B_{y^0} - \beta \psi'(y^0) B(y^1) - D'(x_1^0) \right] = 0$$
(4.30)

$$\frac{\partial W}{\partial t^1} = \frac{\partial x_1^1}{\partial t} \beta \left[ p_1 - c_1'(x_1^1) - \psi(y^0) B_{y^1} - D'(x_1^1) \right] = 0$$
(4.31)

$$\frac{\partial W}{\partial s} = \frac{\partial x_1^1}{\partial s} \beta \left[ p_1 - c_1'(x_1^1) - \psi(y^0) B_{y^1} - D'(x_1^1) \right] + \frac{d x_2^1}{d s} \beta \left[ p_2 - c_2'(x_2^1) - \psi(y^0) B_{y^1} \right] 
+ \frac{\partial x_1^0}{\partial s} \left[ p_1 - c_1'(x_1^0) - B_{y^0} - \beta \psi'(y^0) B(y^1) - D'(x_1^0) \right] 
+ \frac{d x_2^0}{d s} \left[ p_2 - c_2'(x_2^0) - B_{y^0} - \beta \psi'(y^0) B(y^1) \right] = 0$$
(4.32)

### Determination of $s^m$

We rearrange all of the profit FOCs, (4.17), (4.18), (4.19), and (4.20) and find:

$$p_1 - c'_1(x_1^{1m}) = t^1 + s$$

$$p_2(x_2^1) - c'_2(x_2^{1m}) = s - p'_2(x_2^1)x_2^1$$

$$p_1 - c'_1(x_1^0) = t^0 - \beta s$$

$$p_2(x_2^0) - c'_2(x_2^0) = -\beta s - p'_2(x_2^0)x_2^0$$

Plugging these into (4.30), (4.31), and (4.32), we obtain:

$$\frac{\partial x_1^0}{\partial t^0} \left[ t^0 - \beta s - B_{y^0} - \beta \psi'(y^0) B(y^1) - D'(x_1^0) \right] = 0$$
(4.33)

$$\frac{\partial x_1^1}{\partial t^1} \beta \left[ t^1 + s - \psi(y^0) B_{y^1} - D'(x_1^1) \right] = 0$$
(4.34)

$$\frac{\partial x_1^1}{\partial s} \beta \left[ t^1 + s - \psi(y^0) B_{y^1} - D'(x_1^1) \right] + \frac{d x_2^1}{d s} \beta \left[ s - p_2'(x_2^1) x_2^1 - \psi(y^0) B_{y^1} \right] 
+ \frac{\partial x_1^0}{\partial s} \left[ t^0 - \beta s - B_{y^0} - \beta \psi'(y^0) B(y^1) - D'(x_1^0) \right] 
+ \frac{d x_2^0}{d s} \left[ -\beta s - p_2'(x_2^0) x_2^0 - B_{y^0} - \beta \psi(y^0) B_{y^1} \right] = 0$$
(4.35)

We can then solve (4.33) and (4.34) for  $t^0$  and  $t^1$ .

$$t^{0} = \beta s + B_{y^{0}} + \beta \psi'(y^{0})B(y^{1}) + D'(x_{1}^{0})$$
(4.36)

$$t^{1} = -s + \psi(y^{0})B_{y^{1}} + D'(x_{1}^{1})$$
(4.37)

We plug these values into (4.35) in order to obtain the value of s:

$$\frac{dx_2^1}{ds}\beta\left[s - p_2'(x_2^1)x_2^1 - \psi(y^0)B_{y^1}\right] + \frac{dx_2^0}{ds}\left[-\beta s - p_2'(x_2^0)x_2^0 - B_{y^0} - \beta\psi'(y^0)B(y^1)\right] = 0$$

$$s\left[\frac{dx_2^1}{ds}\beta - \frac{dx_2^0}{ds}\beta\right] = \frac{dx_2^1}{ds}\beta[p_2'(x_2^1)x_2^1 + \psi(y^0)B_{y^1}] + \frac{dx_2^0}{ds}[p_2'(x_2^0)x_2^0 + B_{y^0} + \beta\psi'(y^0)B(y^1)]$$
$$s = \frac{\frac{dx_2^1}{ds}\beta[p_2'(x_2^1)x_2^1 + \psi(y^0)B_{y^1}] + \frac{dx_2^0}{ds}[p_2'(x_2^0)x_2^0 + B_{y^0} + \beta\psi'(y^0)B(y^1)]}{\beta[\frac{dx_2^1}{ds} - \frac{dx_2^0}{ds}]}$$

We then substitute in  $\frac{dx_2^1}{ds} = \frac{1}{2p'_2(x_2^1) - c''_2(x_2^1)}$ , and  $\frac{dx_2^0}{ds} = -\frac{\beta}{2p'_2(x_2^0) - c''_2(x_2^0)}$ :

$$s = \frac{\frac{1}{2p_2'(x_2^1) - c_2''(x_2^1)} [p_2'(x_2^1)x_2^1 + \psi(y^0)B_{y^1}] - \frac{1}{2p_2'(x_2^0) - c_2''(x_2^0)} [p_2'(x_2^0)x_2^0 + B_{y^0} + \beta\psi'(y^0)B(y^1)]}{\frac{1}{2p_2'(x_2^1) - c_2''(x_2^1)} + \frac{\beta}{2p_2'(x_2^0) - c_2''(x_2^0)}}$$

Assuming  $c_2''$  is constant and demand is linear, we find Equation (4.21).

#### Determination of $t^{0m}$

Plugging the value of s into (4.36), we have:

$$t^{0} = \beta \left[ \frac{\frac{1}{2p'_{2}(x_{2}^{1}) - c''_{2}(x_{2}^{1})} [p'_{2}(x_{2}^{1})x_{2}^{1} + \psi(y^{0})B_{y^{1}}] - \frac{1}{2p'_{2}(x_{2}^{0}) - c''_{2}(x_{2}^{0})} [p'_{2}(x_{2}^{0})x_{2}^{0} + B_{y^{0}} + \beta\psi'(y^{0})B(y^{1})]}{\frac{1}{2p'_{2}(x_{2}^{1}) - c''_{2}(x_{2}^{1})} + \frac{\beta}{2p'_{2}(x_{2}^{0}) - c''_{2}(x_{2}^{0})}} + B_{y^{0}} + \beta\psi'(y^{0})B(y^{1}) + D'(x_{1}^{0})} \right]$$

Simplifying:

$$t^{0} = \frac{\frac{B_{y^{0}} + \beta\psi'(y^{0})B(y^{1}) + \beta(p'_{2}(x_{2}^{1})x_{2}^{1} + \psi(y^{0})B_{y^{1}})}{2p'_{2}(x_{2}^{1}) - c''_{2}(x_{2}^{1})} - \frac{\beta(p'_{2}(x_{2}^{0})x_{2}^{0})}{2p'_{2}(x_{2}^{0}) - c''_{2}(x_{2}^{0})}}{\frac{1}{2p'_{2}(x_{2}^{1}) - c''_{2}(x_{2}^{1})} + \frac{\beta}{2p'_{2}(x_{2}^{0}) - c''_{2}(x_{2}^{0})}}$$

When we look at the case where c'' is constant, we obtain Equation (4.22).

#### Determination of $t^{1m}$

Plugging  $s^m$  into (4.37), we find:

$$t^{1} = -\left\{ \frac{\frac{1}{2p_{2}'(x_{2}^{1}) - c_{2}''(x_{2}^{1})} [p_{2}'(x_{2}^{1})x_{2}^{1} + \psi(y^{0})B_{y^{1}}] - \frac{1}{2p_{2}'(x_{2}^{0}) - c_{2}''(x_{2}^{0})} [p_{2}'(x_{2}^{0})x_{2}^{0} + B_{y^{0}} + \beta\psi'(y^{0})B(y^{1})]}{\frac{1}{2p_{2}'(x_{2}^{1}) - c_{2}''(x_{2}^{1})} + \frac{\beta}{2p_{2}'(x_{2}^{0}) - c_{2}''(x_{2}^{0})}} + \psi(y^{0})B_{y^{1}} + D'(x_{1}^{1}) \right\}$$

Simplifying:

$$t^{1} = D'(x_{1}^{1}) + \frac{\frac{p_{2}'(x_{2}^{1})x_{2}^{1} + B_{y^{0}} + \beta\psi(y^{0})B_{y^{1}} + \beta\psi'(y^{0})B(y^{1})}{2p_{2}'(x_{2}^{0}) - c_{2}''(x_{2}^{0})} - \frac{p_{2}'(x_{2}^{1})x_{2}^{1}}{2p_{2}'(x_{2}^{1}) - c_{2}''(x_{2}^{1})}}{\frac{1}{2p_{2}'(x_{2}^{1}) - c_{2}''(x_{2}^{1})} + \frac{\beta}{2p_{2}'(x_{2}^{0}) - c_{2}''(x_{2}^{0})}}$$

After rearranging and taking the case where c'' is constant, we obtain Equation (4.23).

#### Determination of calculated quantities

As with  $y^{1m}(x_1^{1m}, x_2^{1m})$  and  $y^{0m}(x_2^{0m}, x_2^{0m})$ , quantities  $(x_1^{1m}, x_2^{1m}, x_2^{0m}, x_2^{0m})$  are obtained solving the following system:

$$p_1 - c_1'(x_1^1) - D'(x_1^1) + \frac{\beta - 1}{1 + \beta} \psi(y^0) B_{y^1} = 0$$

$$\frac{\beta}{1+\beta}p_2'(x_2^1)x_2^1 + p_2(x_2^1) - c_2'(x_2^1) - \left(\frac{\psi(y^0)B_{y^1} - p_2'(x_2^0)x_2^0 - B_{y^0} - \beta\psi'(y^0)B(y^1)}{1+\beta}\right) = 0$$

$$p_1 - c_1'(x_1^0) - D'(x_1^0) - B_{y^0} - \beta \psi'(y^0) B(y^1) = 0$$

$$\frac{p_2'(x_2^0)x_2^0}{1+\beta} + p_2(x_2^0) - c_2'(x_2^0) + \frac{\beta}{1+\beta} \Big( p_2'(x_2^1)x_2^1 + \psi(y^0)B_{y^1} - B_{y^0} - \beta\psi'(y^0)B(y^1) \Big) = 0$$
# Chapter 5\_

# On the efficiency of the mitigation hierarchy

This article focuses on the avoid, reduce, compensate (ARC) sequence that accompanies the no net loss policy. It studies the behavior of a developer in the face of this policy. Under perfect information, it appears that this policy is a policy of environmental standards, whose objectives are difficult to transpose into a microeconomic decision model. Moreover, we show that the demand for compensation does not depend on its price. We then assume that the regulator does not share the same information as the developer on the environmental damage of the project. In this case, the developer strategically uses this asymmetric information. Using the backward induction reasoning, he simultaneously defines his demand for offsets and the level of environmental damage reduction based on the offset price. In the end, the project choice is made by also taking into account the price of the offset. This article shows that the mitigation hierarchy is ineffective under asymmetric information, making the safeguarding of biodiversity inefficient.<sup>1</sup>

**Keywords**: biodiversity conservation  $\cdot$  avoid reduce compensate sequence  $\cdot$  mitigation banking  $\cdot$  mitigation hierarchy

<sup>&</sup>lt;sup>1</sup>This chapter is joint work with Sonia Schwartz.

#### 5.1 Introduction

The collapse of biodiversity is a well-documented phenomenon, which is likely to worsen with climate change (Dasgupta, 2021; Díaz et al., 2019; Ruckelshaus et al., 2020). A leading cause of the decline in biodiversity is the loss of various habitats due to land use change (Lewis et al., 2011; Bamière et al., 2013). According to Dasgupta (2021), an estimated 20% of species could become extinct in the next several decades, perhaps twice as many by the end of the century.

One concept that aims to halt biodiversity loss is mitigation banking, the idea behind which started in the context of declining wetland area in the United States, which was often the result of land use shifts to urban development projects and agriculture (Burgin, 2010; Dahl, 1990). Wetlands act as habitat for many species and also provide many environmental services, including water filtration and flood management. The Ramsar Convention in 1971 established an international treaty for the conservation and sustainable use of wetlands, especially as habitat for birds. In order to address the loss of wetland area, the US Clean Water Act of 1972 introduced a permitting program that requires following a mitigation hierarchy to obtain a permit for a development project. The mitigation hierarchy lists the steps to be taken by major development projects to achieve the goal of no net loss of wetland area: avoid, reduce, and compensate (ARC). Consider a plan for a development project that will damage at least part of a wetland. First, the developer must avoid as much damage as possible, for example by relocating the project or reducing its size. Second, if some damage remains, it should be reduced as much as possible, using pollution abatement technologies or other methods. And finally, if there is any remaining damage, there are two options to compensate for said damage: either re-establish a similar wetland, on site or at another location, or purchase credits from a mitigation bank. A mitigation bank buys credits from restoration projects and sells them to developers who need to offset their residual impacts that could not be avoided or reduced.

In France, the mitigation hierarchy was introduced by the founding law for the Protection of Nature of 1976. The effectiveness of the mitigation hierarchy is measured via impact studies, which are required when obtaining a permit for development projects of a certain nature or size that are likely to affect protected species or habitats (Bigard et al., 2018; Levrel et al., 2018). It applies to projects, plans and programmes subject to environmental assessment as well as to projects subject to various administrative authorization procedures under the Environmental

Code, such as environmental authorization, derogations for species protection or Natura 2000 impact assessment. The ARC sequence is widely practiced in European Union environmental policy and EU Directives, such as the Habitats Directive, have been a major driver in the reinforcement of the ARC sequence in France (Quétier et al., 2014). France's Law 2010-788 of July 12, 2010 led to important reforms concerning the mitigation of development impacts on biodiversity, including reforms on the requirements for impact assessments and enforcement capabilities (Quétier et al., 2014). Governmental guidance from 2012 states that compensatory actions should last as long as impacts, but there is little guidance about design, duration, or frequency of monitoring efforts (Quétier et al., 2014). The 2016 Biodiversity Law resulted in compensation becoming mandatory for residual impacts (Levrel et al., 2018) and introduced the use of natural compensation sites to anticipate future compensation demands (Aubry et al., 2021).

The idea of natural compensation sites first materialized in France with the pilot project on the Cossure site in the Bouches-du-Rhône department. This site was created in 2009 by the organization CDC Biodiversité, with the support of the Ministry of Ecology and Sustainable Development (MEDD) (Dutoit et al., 2018). The idea behind natural compensation sites is to create a supply of compensation credits by restoring larger connected areas, in order to avoid the time lag between damaging habitats and restoring compensation areas.

The economic literature has examined the efficacy of restoration policies to protect biodiversity. From an ecological point of view, the results of various monitoring studies show that the goal of restoring herbaceous vegetation has been successful so far, and that its maintenance should be upheld thanks to the re-establishment of pastoral practices in the area (Dutoit et al., 2018). However, the final ecological assessment of the restoration actions can only be carried out with a longer time span, as varying weather conditions can lead to different invasive species thriving, as happened in 2014 when a particularly rainy summer allowed an invasive species to proliferate (Dutoit et al., 2018). Campbell et al. (2002) compare natural and created wetlands in the state of Pennsylvania, looking at variables related to soil and plant quality and found that even the oldest created wetlands had few similarities with their natural counterparts. Tillman et al. (2022) looked at wetland mitigation banks that have aged past the required 5-year management and monitoring periods and found that plant communities in wetland banks have greater conservation value than the lowest quality, degraded natural wetlands, but were not close to the same value as high-quality, reference natural wetlands. Reiss et al. (2009) studied wetland

mitigation banks in Florida and found that while most banks were deemed successful in terms of permit criteria, the permit criteria were not explicitly tied to ecological criteria, and so the functional performance provided by the wetland banks remains unclear. While the natural compensation site has demonstrated the possibility to restore some parts of a natural habitat, it has also demonstrated the limitations of compensation actions to fully restore lost ecosystems (Dutoit et al., 2018).

Another part of the literature questions the feasibility of the compensation step. There is an issue of finding compensation areas that equate to the damaged areas, which is seen as one of the main challenges of mitigation banking (de Muelenaere, 2011). Often, criteria may be simplified in order to allow for more participation in compensation markets. There is also a potential issue of displacing wetlands or other natural environments from more urban to more rural areas due to differing land prices, making it more cost-effective for restoration projects to occur in more rural areas because of the lower land prices.

While the compensation step of the sequence has arguably received the most attention in the literature, many studies regarding the mitigation hierarchy also highlight that the first step, avoidance, is the most important but is "more often ignored than implemented" (Clare et al., 2011). Avoidance is the most certain and effective way to limit impacts on biodiversity, as it does not engender the same problems as compensation, such as restoration time lags, limitations to what can be offset, and negative social implications from taking away biodiversity in one area and improving it in another (Phalan et al., 2018). A few papers describe different reasons for which the avoidance step is not properly implemented. Clare et al. (2011) identify five key factors that lead decision-makers to fail to prioritize wetland impact avoidance and reduction above compensation in the US and Canada, namely a lack of consensus on what constitutes avoidance, a failure of land-use planning approaches to identify high-priority wetlands in advance of development, an economic undervaluation of wetlands, a "techno-arrogance" associated with wetland creation and restoration that results in wetland loss, and finally inadequate enforcement of compensation requirements. Similarly, Phalan et al. (2018) identify five challenges for effective impact avoidance: political will, legislation quality and its implementation in practice, process, capacity (informational and transaction costs), and technical knowledge. Finally, Levrel et al. (2018) identify five drifts that undermine the additionality of the ARC sequence in France, which relate to a diversion of resources from existing conservation actions toward compensation measures and the pursuit of rents and cost minimization by different stakeholders.

Bigard et al. (2018) sought to evaluate how the execution of the ARC sequence in France aligned with the definitions and national guidance for each step. They analyzed 42 impact studies for projects between 2006 and 2016 in the territory of the Montpellier metropolis and contiguous municipalities and found that in 60% of the cases, the qualifications of the ARC measures given in the impact study did not correspond to the national reference definitions. For example, the so-called avoidance measures in the impact studies were actually reduction measures according to the national reference definitions. They also found that this confusion had negative consequences on the ecological effectiveness of the ARC hierarchy. As Stevenson and Weber (2020) note, there is a temptation to skip to steps lower in the hierarchy that are easier or cheaper. The aim of this paper is to define the operational contents of the sequence and to identify the cases where the sequence is inoperative.

While the economic literature has focused on the efficiency of the sequence in protecting biodiversity or on the realization of the different stages of the sequence, the essence of the policy is unclear. It appears a confusion between not net loss policies based on the ARC sequence and a market-based biodiversity conservation policy. Mitigation banking can be considered a type of PES scheme, as both instruments involve providing payment for the restoration, preservation, and/or management of biodiversity and ecosystems (Bureau, 2010; Combe, 2020). However, a PES is a market mechanism. The not-net-loss policy is an environmental standard, accompanied by the implementation of three successive standards: avoid, reduce and compensate. However, this possibility of compensation can lead to the development of an offset supply. The simultaneity of norms, prices and the supply of offsets can lead to an amalgam of the ARC sequence with a market instrument. One of the objectives of this article is to clarify this point.

The economic literature has identified various factors contributing to the failure to meet the no net loss objective have been widely discussed. However, to our knowledge, there has not yet been an attempt to model the developer's behavior in the face of various incentives to comply or not with the different steps of the mitigation hierarchy. The objective of this paper is not to analyze the efficiency of the ARC sequence in protecting biodiversity as this question is better suited to biological or ecological analysis. The idea of this article is to analyze how a developer behaves when faced with the ARC sequence.

To answer this question, we will mobilize two informational contexts: perfect information and asymmetric information. Under perfect information, it is assumed that the regulator has all the information about the developer's projects, both on the level of possible damage and the possibilities of damage reduction. Second, we assume that this information is only held by the developer. Under both assumptions, we seek to characterize the behavior of a developer in the face of the ARC sequence and the no net loss policy. We first highlight the difficulties of transposing the sequence into an economic decision model with perfect information. Moreover we show that the demand for compensation does not depend on its price. Under asymmetric information, the developer uses information strategically to achieve the no net loss objective by circumventing the ARC sequence. We show that, using the backward induction reasoning, the developer first defines his demand for offsets based on their price. In the end, the project chosen is the one that is most profitable given the compensation expense. Under asymmetric information, the price of compensation is therefore a key variable, unlike in the situation of perfect information. The no net loss policy accompanied by the ARC sequence is a policy of environmental standards and not an economic instrument. In the real world, asymmetric information seems the most likely hypothesis. In this case, the ARC sequence is inoperative, with the consequence that it is ineffective in protecting biodiversity. The generalization of the ARC sequence and the creation of a public agency for the preservation of biodiversity may allow the problem of asymmetric information to be overcome, thus making the sequence more operative.

The article is structured as follows. Section 5.2 outlines the assumptions of our model. Then, Section 5.3 integrates the ARC sequence into a decision model under perfect information and Section 5.4 under asymmetric information. Finally, discussion and conclusion are presented in Section 5.5.

#### 5.2 Assumptions

We assume a developer wants to invest in a project  $v_i$ , i = 0, 1. The net economic benefit without taking into account the environmental damages is given by  $B(v_i)$ . Each project causes environmental damages in the amount of  $D(v_i)$ . We assume that the project  $v_1$  generates fewer benefits but less environmental damage than the project  $v_0$ . So we have:  $B(v_1) < B(v_0)$  and  $D(v_1) < D(v_0)$ . The regulator imposes the no net loss principle and the ARC sequence in order to help the developer achieve this environmental policy goal. Thus, each developer must sequentially avoid, reduce and compensate for its environmental losses.

Damage reduction is achieved by choosing a damage level  $D^{\min}(v_i)$  lower than the

initial damage, such as:  $r = D(v_i) - D^{\min}(v_i)$ . Damage reduction imposes an additional cost on the company given by C(r) with C'(r) > 0 and C''(r) > 0. The remaining environmental damage must be compensated. Offset banks offer offset credits. The offset credit market is assumed to be atomistic so that the price of the credits, p, is a competitive price.

#### 5.3 Perfect and symmetric information

We first assume perfect and symmetric information. In this case, the regulator and the developer share the same information about the environmental damages and the different costs of reducing the damages. The regulator can monitor whether the developer follows the ARC procedure. Under this assumption, the developer can only respect, step by step, the sequence. We follow the French definitions from the MEDD for each step of the sequence.

The avoidance stage For the avoidance stage, the definition of the MEDD emphasizes that the design of a project "must first of all seek to avoid impacts on the environment, including the fundamental choices related to the project (the nature of the project, location, even opportunity)". We interpret this definition as the fact that the developer must avoid the environmental impacts of the project as much as possible. In our framework, he chooses the project  $v_1$ , which causes the least environmental damage  $(D(v_1) < D(v_0))$ .

The reduction stage The MEDD gives the following definition for the reduction stage: "These impacts must then be sufficiently reduced (...) at a reasonable cost, to constitute only the smallest possible residual negative impacts." This step lacks operational content. Indeed, what does sufficiently reduced mean? What is a reasonable cost? If the level of damage reduction has to be the lowest level achievable by the developer, it is possible that the cost of reduction is such that the developer's profit is negative. Here, we clarify the desired requirement at this stage as keeping the level of damage reduction as high as possible, while remaining consistent with a non-negative profit. If the level of damage chosen is positive, this constraint must include the necessary cost of compensation, which is included in step 3. We note  $D^{\min}(v_1)$  this level of damage. So  $r = D(v_1) - D^{\min}(v_1)$ , with  $D^{\min}(v_1) < D(v_1)$ . So  $D^{\min}(v_1)$  has to solve  $B(v_1) - C(D(v_1) - D^{\min}(v_1)) = 0$ . If there is no damage level satisfying this condition, the developer chooses  $D^{\min}(v_1)$  such as:  $B(v_1) - C(D(v_1) - D^{\min}(v_1)) - pD^{\min}(v_1) = 0$ . Despite the developer's efforts to minimize damage, there may be no possibility of complete damage reduction compatible with the non-negative profit constraint. In this case, a positive level of damage is implied.

The compensation stage The compensation stage only appears if the damage level is positive, i.e.  $D^{\min}(v_1) > 0$ . The not net loss principle imposes a necessary damage compensation. This compensation is only permitted by the regulator if the ARC sequence has been previously scrupulously followed by the developer. The demand for offsets is therefore determined in a residual manner. This observation allows us to make this proposition:

**Proposition 1.** Under perfect information, the offset demand does not depend on the offset price.

The compensation cost is given by  $pD^{\min}(v_1)$  if  $D^{\min}(v_1) > 0$  and 0 if  $D^{\min}(v_1) = 0$ .

In the end, by applying the rules of the ARC sequence under perfect information, the developer chooses the project  $v_1$ , opts for an environmental reduction level compatible with a non negative profit and compensates for the residual damage. The sequence is respected at each level, which allows for the best protection of biodiversity. The price of compensation does not influence the developer's behavior. The ARC sequence coupled with the no net loss policy is a succession of environmental standards.

#### 5.4 Asymmetric Information

We now assume that only the developer has information about the different projects he wants to carry out, including the environmental damage of each project,  $D(v_i)$ , and the possibilities of damage reduction given by the function C(r). The information between the regulator and the developer is therefore asymmetric. The assumption of asymmetric information leads to two differences compared to the situation of perfect information. On the one hand, the regulator cannot know whether the developer has respected the ARC sequence. On the other hand, the developer – as an homo economicus – will not consider the sequence separately, that is, in a myopic way. The developer can be expected to behave strategically while respecting the no net loss requirement. He will adopt backward induction reasoning in order to choose the level of reduction and compensation. In the end, the rational developer will choose the project that offers the greatest profit. Assuming that the developer chooses the project  $v_i$ , he decides the amount of reduction and thus the level of compensation by minimizing the total environmental conformity cost:

$$\operatorname{Min} TC_{D\tilde{v}_i} = C(Dv_i - D\tilde{v}_i) + pD\tilde{v}_i$$
$$-C'(Dv_i - D\tilde{v}_i) + p = 0$$
(5.1)

According to Equation (5.1), the developer chooses the level of damage reduction  $[Dv_i - D\tilde{v}_i]$  such that the marginal cost of environmental damage reduction is equal to the offset price. If  $D\tilde{v}_i$  is equal to zero, there is no damage and no need for compensation. If  $D\tilde{v}_i > 0$ , the developer has to buy offset credits in order to compensate for the environmental damage at a cost  $pD\tilde{v}_i$ .

Therefore, what will explain the compensation level is the shape and limits of the damage reduction curves and the price of the offset credits. Compensation occurs if  $C'(Dv_i)$  has a finite limit in  $D\tilde{v}_i = 0$  such that the offset price is lower than this limit or if  $C'(Dv_i)$  satisfies Inada's condition in  $D\tilde{v}_i = 0$ . However, there is never compensation if  $C'(Dv_i)$  has a finite limit in  $D\tilde{v}_1 = 0$  such that the offset price is lower than this higher than this limit.

**Proposition 2.** Under asymmetric information, the demand for compensation is based on the offset price.

Thus, the level of compensation is no longer residual as required by law (step 2 of the ARC sequence), but results from an economic calculation. Homo economicus uses the asymmetric information to maximize its profit. Finally, the profit obtained with the project  $v_i$  is written as:

$$\Pi(v_i) = B(v_i) - C(D(v_i) - D(\tilde{v}_i)) - pD(\tilde{v}_i)$$

In the end, the developer chooses the project that gives the higher level of profit. He will choose the project  $v_1$  if:

$$\Pi(v_1) > \Pi(v_0)$$

Which can also be written as:

$$B(v_1) - C(D(v_1) - D(\tilde{v}_1)) - pD(\tilde{v}_1) > B(v_0) - C(D(v_0) - D(\tilde{v}_0)) - pD(\tilde{v}_0)$$
(5.2)

As  $C'(Dv_0 - D\tilde{v}_0) = p = C'(Dv_1 - D\tilde{v}_1)$ , we have:  $Dv_0 - D\tilde{v}_0 = Dv_1 - D\tilde{v}_1$ , and  $C(D(v_1) - D(\tilde{v}_1)) = C(D(v_0) - D(\tilde{v}_0))$ . Replacing in (5.2), we obtain:

$$B(v_1) - B(v_0) > p[D(\tilde{v}_1) - D(\tilde{v}_0)]$$

A project  $v_1$  is chosen if the difference in benefit between projects  $v_1$  and  $v_0$  is greater than the difference in expenditure between the projects on compensation.

**Proposition 3.** The project choice only depends on the project benefit and on the offset price.

Thus, the offset price determines the compensation level and thus the level of the environmental damage reduction. The offset price is also the determining factor in the choice of project. At no point does the avoidance stage play a role in the behavior of homo economicus. The reduce and compensate stages are not sequential but simultaneous. In the end, the ARC sequence is inoperative under asymmetric information. The developer makes decisions based on the compensation price. In this case, the not net loss policy can be considered as a market-based instrument.

#### 5.5 Discussion and conclusion

The ARC sequence appeared in the United States in 1972, then in 1976 in French law. In France, the definitions of the Avoidance, Reduction, and Compensation measures were introduced in 2012, followed by the not net loss of biodiversity objective in its Biodiversity Law of 2016. The study of this sequence in the economic literature is most often done by taking into account its impact on biodiversity conservation (Brown and Land, 1999; Quétier et al., 2014; Calvet et al., 2019). Some studies, such as Bigard et al. (2018), highlight the difficulties of translating the concepts of the ARC sequence into practice. According to the authors, the qualifications given in the impact studies often do not correspond to the national reference definitions. In particular, they note a confusion between avoidance and reduction measures.

Following Bigard et al. (2018), our work sought to analyze the operational dimension of the ARC sequence, but mobilizing another angle of study. We sought to incorporate the ARC sequence into an economic decision model. The objective of this paper is to investigate the rationality of the ARC sequence. To do so, we assumed the hypotheses of perfect and asymmetric information.

Using the perfect information hypothesis, we first highlighted the difficulties of transposing the imperatives of the sequence into an economic decision model. The ARC sequence corresponds to the implementation of three standards: avoid, reduce, and compensate. We selected the project that caused the least environmental damage, at the avoidance stage, which implies the existence of several projects or project modalities. The reduction stage raises the question of the meaning of "sufficiently reduced" (...) at "a reasonable cost". This point is difficult to transpose into an economic model. Here we have chosen the highest level of damage reduction that can be accounted for with a non-negative profit. We have shown that when the sequence is respected, the demand for compensation does not depend on its price. Under symmetric information, the ARC sequence is an environmental standard.

The environmental economics literature notes the importance of information in the implementation of environmental standards as a means of pollution control. However, the sequence goes even further in requiring information than pollution standards because it involves sequential implementation. So, we then lift this assumption of perfect information by assuming that the regulator does not have all the information about the projects. In this case, we showed the information is used strategically by the developer and that the ARC sequence is inoperative. The developer will not behave in a myopic way, considering the different steps independently of each other. Rather, as an homo economicus, he will reason backwards in order to choose the most profitable project, under constraint of the no net loss policy. We show that the developer will simultaneously decide on the amount of the damage reduction and the compensation based on the offset price and that, in the end, the choice of project is established in particular on the offset price. In other words, establishing this sequence in the law involving the no net loss policy is inefficient when the information is asymmetric. Therefore, under asymmetric information, the no net loss policy is a market-based instrument.

However, our article has not taken into account the existence of another operator, the auditors. In fact, the ARC sequence is carried out through impact studies. The developer is responsible for the study, and can call on the services of consulting firms to draw up this document. The company pays for these services. In the face of the operational vagueness of the ARC sequence, one can imagine that the company can draft the impact study by evading certain information or minimizing certain impacts, which justifies our hypothesis of asymmetric information. One way to overcome this information problem would be to create a specialized public agency whose mission would be to conduct all these impact studies. This agency would benefit from the experience gained from one study to another and would be better able to apply the same interpretation of the ARC sequence to each project. The generalization of the sequence in France to all projects regardless of their location and size would allow for these economies of scale. In our article, we have considered several possible projects with different levels of damage. In reality, developers consider a single project, hence the difficulties in interpreting the avoidance stage. One could imagine that the developer would be obliged to communicate to this agency his different projects so that the ARC sequence is applied upstream of the decisions. In the end, the creation of this agency would tend to remove the asymmetry of information concerning the environmental damage of the project, thus allowing the ARC sequence to be implemented. The sequence would be a standards approach, not a market-based approach. The objective of no net loss is thus achieved, but not at the lowest cost.

In fact, our work highlights the different objectives of a biodiversity conservation policy, either strictly protecting biodiversity or implementing the least cost criterion. The ARC sequence coupled with the not net loss objective is a sequence of norms aimed at preserving biodiversity. The offset mechanism, by putting a price on the offset, can be seen as an economic incentive to compensate. If each developer reduces his damage by equating the marginal cost of reducing the damage to the price of the offset, biodiversity will be saved at lower costs. This result is not possible by applying the ARC sequence, but perhaps this is the price to pay for taking into account a very particular and difficult to measure good, biodiversity.

In this article, we have assumed an exogenous price for offsets. When the supply of offsets is not well developed, this assumption can be challenged. Future research should take into account the sequence when the price of the offset is established on an over-the-counter market.

# Chapter 6

# Conclusion

Biodiversity collapse is a well-documented problem that affects human well-being through the decline in provision of numerous ecosystem services. Terrestrial biodiversity loss is mainly due to land use changes, in particular conversion to and intensification of agricultural lands. This thesis, consisting of four chapters, has focused on the theoretical efficiency of one market-based policy instrument, payments for environmental services (PES), which aims to address the problem of biodiversity decline by financially incentivizing its provision. The first chapter reviewed the literature on PES while the other three chapters fill in gaps in the existing literature by examining specific aspects of imperfect competition and PES design and efficiency are impacted. Two chapters analyzed price-based PES and one chapter analyzed a quantity-based PES.

### 6.1 Main results

In the first article, we looked at a scenario where an environmental tax and a PES scheme are used to address pollution and biodiversity conservation, respectively. Our model looked at how several identical farmers choose to allocate their land between conventional agriculture production, organic agriculture production, and leaving grass buffer strips. We assumed that conventional production causes environmental damages, organic production has a neutral impact on the environment, and the grass buffer strips provide environmental benefits. We added an additional market distortion in the form of an oligopoly in organic agriculture production. We found that the second-best tax on conventional agriculture production is higher than the marginal damage from pollution, and the second-best PES for biodiversity is lower than the marginal benefit. In order to account for the fact that funds must be raised to finance a public PES, we then introduced the marginal social cost of public funds (MCF). The PES decreases with the MCF, whereas the Pigouvian tax increases with the MCF, provided that demand for the conventional agriculture good is inelastic

and environmental policies provide buffer strips efficiently. The first article highlights a contributory component of the environmental incentive tax. We also identify cases where the PES is ineffective in promoting biodiversity.

This first article was extended by considering other assumptions. First, we have challenged the assumption of a neutral impact of organic farming on biodiversity by assuming that fallow buffer strips produce more biodiversity than organic farming. In this case, we use two PES. The level of organic farming would be subject to two effects: a negative effect that favors buffer strips and a positive effect that favors biodiversity from organic farming. The first effect would therefore outweigh the second and the mechanisms highlighted in the first article would remain relevant. Second, under our assumptions, we have modified the environmental policy tools by considering two PES schemes, one on uncultivated land and the other on organic agriculture but no environmental tax. We found that the PES for organic agriculture takes the market power into account, and is higher than the marginal benefit of organic production, whereas the PES for uncultivated land is equal to the marginal benefit of biodiversity and no longer adjusts to incorporate the market power. Finally, we have challenged the assumption that there are no negative externalities of conventional agricultural production on the level of organic production. In this case, we found that the farmer will internalize this negative impact himself and the PES and environmental tax levels do not differ from those in the main scenario of the article.

Budgets for PES programs are often limited, and regulators may decide to condition payments on additional area of a biodiversity-friendly land use. We denote such a policy as additionality-based PES. The second paper looks at whether additionalitybased PES achieve environmental efficiency. We used a model with two time periods and we considered the representative farmer's behavior. The regulator sets a PES that only pays for the additional environmental benefits that result from the PES. We show that this PES based on additionality distorts the farmer's behavior in the initial period. The farmer increases his production levels in order to obtain more payment in the final period. The second-best PES has to correct this distortion while taking into account the environmental benefits and damages. In the end, the second-best PES is equal to the discounted difference of the marginal environmental benefit in each period. The second-best environmental taxes in each period are no longer equal to the marginal damage. They are adjusted to correct the distortions induced by the PES. We then introduced market power in the organic market. If the market power reduces the organic production quantity in the final period, it limits the distortion in the initial period. Depending on the size of these effects, the second-best additionalitybased PES either increases or decreases compared to the scenario without market power. The taxes are adjusted to take into account the indirect effects of the market power on the level of conventional production. Finally, the second study has shown that the additionality condition of the PES does not achieve environmental efficiency, even under perfect competition. It also provides a better understanding of the interactions between different types of environmental policies.

Turning to a quantity-based PES, we looked at mitigation banking, and in particular, the mitigation hierarchy or avoid, reduce, compensate (ARC) sequence in the third paper. Some studies, such as Bigard et al. (2018), highlight the difficulties of translating the concepts of the ARC sequence into practice. According to the authors, the qualifications given in the impact studies often do not correspond to the national reference definitions. In particular, they note a confusion between avoidance and reduction measures. Following Bigard et al. (2018), our work sought to analyze the operational dimension of the ARC sequence, but mobilizing another angle of study. We sought to incorporate the ARC sequence into an economic decision model, with the objective of investigating its rationality. To do so, we assumed the hypotheses of perfect and asymmetric information.

Using the perfect information hypothesis, we first highlighted the difficulties of transposing the imperatives of the sequence into an economic decision model. The ARC sequence corresponds to the implementation of three standards: avoid, reduce, and compensate. We selected the project that caused the least environmental damage, at the avoidance stage, which implies the existence of several projects or project modalities. The reduction stage raises the question of the meaning of "sufficiently reduced" (...) at "a reasonable cost". This point is difficult to transpose into an economic model. Here we have chosen the highest level of damage reduction that can be accounted for with a non-negative profit. We have shown that when the sequence is respected, the demand for compensation does not depend on its price. Under symmetric information, the ARC sequence is an environmental standard.

The environmental economics literature notes the importance of information in the implementation of environmental standards as a means of pollution control. However, the sequence goes even further in requiring information than pollution standards because it involves sequential implementation. So, we then lift this assumption of perfect information by assuming that the regulator does not have all the information about the projects. In this case, we showed the information is used strategically by the developer and that the ARC sequence is inoperative. The developer will not behave in a myopic way, considering the different steps independently of each other. Instead, as an homo economicus, he will reason backwards in order to choose the most profitable project, under the constraint of the no net loss policy. We show that the developer will simultaneously decide on the amount of the damage reduction and the compensation based on the offset price and that, in the end, the choice of project is based on the offset price. In other words, establishing this sequence in the law involving the no net loss policy is inefficient when the information is asymmetric. Therefore, under asymmetric information, the no net loss policy is a market-based instrument.

In fact, our work highlights the different objectives of a biodiversity conservation policy, either strictly protecting biodiversity or implementing the least cost criterion. The ARC sequence coupled with the not net loss objective is a sequence of norms aimed at preserving biodiversity. The offset mechanism, by putting a price on the offset, can be seen as an economic incentive to compensate. If each developer reduces his damage by equating the marginal cost of reducing the damage to the price of the offset, biodiversity will be saved at lower costs. This result is not possible by applying the ARC sequence, but perhaps this is the price to pay for taking into account a very particular and difficult to measure good, biodiversity.

#### 6.2 Limitations and future research

Our theoretical analyses of PES policies, like all analyses, have limitations. While we explored some alternative assumptions in the first study, it could also be extended to consider differentiated demands for organic agriculture that occur for some level of market power. Another extension would be to look at the case where farmers are heterogeneous and analyze the definition of PES when the conventional good production has negative externalities on the organic good production and these externalities cannot be directly internalized. Additionally, the first study employs a relatively simple model, where the only way to increase biodiversity benefits is by leaving grass strips. This management action is assumed to only incur opportunity costs of not producing. Indeed, existing programs remunerate several different land management actions, some of which incur other costs, such as planting cover crops. Future research could explore how the farmer chooses which management action(s)

to undertake and how this impacts the level of environmental service provision.

Regarding the second study, it turns out that basing the PES on the additional environmental benefits obtained by the payment is not easy to characterize in a simple perfect information setting with two time periods. However, the second study could be extended by considering an infinite time horizon, in order to see how the results hold up. This would mean modeling strategic behavior in an optimal control model. In the same vein as Barnett (1980), we defined the second-best environmental policies under perfect information, which suggests that the regulator knows the firm's production costs. By mobilizing agency theory, these works could be extended under asymmetric information, which would allow for taking moral hazard into account. Finally, the model could be enriched by introducing the marginal social cost of public funds.

The third study examined the rationality of the ARC sequence. However, our study has not taken into account the existence of another operator, the auditors. In fact, the ARC sequence is carried out through impact studies. The developer is responsible for the study, and can employ consulting firms to draw up this document, which entails another cost for the developer. In the face of the operational vagueness of the ARC sequence, one can imagine that the consulting firm can draft the impact study by evading certain information or minimizing certain impacts, which justifies our hypothesis of asymmetric information. One way to overcome this information problem would be to create a specialized public agency whose mission would be to conduct all of these impact studies. This agency would benefit from the experience gained from one study to another and would be better able to apply the same interpretation of the ARC sequence to each project. The generalization of the sequence in France to all projects regardless of their location and size would allow for these economies of scale.

In our study, we have considered several possible projects with different levels of damage. In reality, developers consider a single project, hence the difficulties in interpreting the avoidance stage. One could imagine that the developer would be obliged to communicate to the specialized public agency his different projects so that the ARC sequence is applied upstream of the decisions. In the end, the creation of the specialized public agency would tend to remove the asymmetry of information concerning the environmental damage of the project, thus allowing the ARC sequence to be implemented. The sequence would be a standards approach, not a market-based approach. The objective of no net loss is thus achieved, but not at the lowest cost.

Finally, we have assumed an exogenous price for offsets. When the supply of offsets is not well developed, this assumption can be challenged. This work can be extended by taking into account the sequence when the price of the offset is established on an over-the-counter market.

#### | Chapter

# Résumé extensif en français: Paiements pour services environnementaux et concurrence imparfaite

Ce résumé extensif de la thèse est structuré comme suit. La section 7.1 définit les notions de la biodiversité et les services environnementaux. La section 7.2 introduit le concept de paiements pour les services environnementaux. Les motivations de la thèse sont exposées dans la section 7.3 et la section 7.4 présente les principaux résultats. Enfin, la section 7.5 discute des pistes de recherche futures.

# 7.1 De la biodiversité aux services environnementaux

La Convention sur la Diversité Biologique définit la biodiversité comme "la variabilité des organismes vivants de toute origine y compris, entre autres, les écosystèmes terrestres, marins et autres écosystèmes aquatiques et les complexes écologiques dont ils font partie ; cela comprend la diversité au sein des espèces et entre espèces ainsi que celle des écosystèmes" (Convention on Biological Diversity, 2010). La diversité de la vie et des écosystèmes permet diversifier les services environnementaux. Dasgupta (2021) les compare aux portefeuilles financiers : si la diversité des investissements permet d'atténuer les risques, la biodiversité permet à la nature d'être plus productive, résiliente et adaptable.

Le déclin de la biodiversité est un phénomène bien documenté, qui risque de s'aggraver avec le changement climatique (Díaz et al., 2019; Ruckelshaus et al., 2020; Dasgupta, 2021). L'une des principales causes de ce déclin est la perte de divers habitats due au changement d'affectation des sols (Lewis et al., 2011; Bamière et al., 2013). En effet, une grande partie de la perte de biodiversité terrestre provient de la transformation des sols en terres agricoles. L'intensification de l'usage des terres et l'augmentation de la taille des exploitations agricoles ont transformé et fragmenté les habitats naturels, entrainant le déclin de nombreuses espèces. Selon Dasgupta (2021), on estime que 20 % des espèces pourraient disparaître au cours des prochaines décennies, et peut-être le double d'ici la fin du siècle.

L'expression "services de la nature" est apparue pour la première fois en 1977 dans la littérature académique dans un article de Walter Westman (Westman, 1977). Elle a été suivie du terme "services écosystémiques", évoqué en 1981 par Costanza et al. (2017). Toutefois, ce n'est qu'à la fin des années 1990, lorsqu'un article paru dans Nature a estimé que la valeur totale de tous les services écosystémiques de la biosphère se situait entre 16 et 54 billions de dollars américains, que la notion de services écosystémiques a gagné en popularité en tant que sujet de recherche (Costanza et al., 1997).

L'évaluation des écosystèmes pour le millénaire définit les services écosystémiques comme les avantages que les agents retirent des écosystèmes. Il s'agit notamment des *services d'approvisionnement* tels que la nourriture, l'eau, le bois et les fibres ; des *services de régulation* qui affectent le climat, les inondations, les maladies, les déchets et la qualité de l'eau ; des *services culturels* qui fournissent des avantages récréatifs, esthétiques et spirituels ; et des *services de soutien ou services de support* qui permettent la production d'autres services tels que la formation des sols, la photosynthèse et le cycle des nutriments (Reid et al., 2005).

La littérature économique fait la distinction entre les services écosystémiques et les services environnementaux (SE). Le terme "services écosystémiques" fait référence aux bénéfices fournis par les écosystèmes tandis que le terme "services environnementaux" renvoie a la protection de ces écosystèmes par l'homme et a la notion d'externalités induites par les activités humaines. Par exemple, l'Organisation des Nations Unies pour l'alimentation et l'agriculture (FAO) propose une définition des SE en termes de services écosystémiques. Pour l'agriculture, les SE sont définis comme la sous-partie des services écosystémiques qui peuvent être qualifiés d'externalités, c'est-a-dire tous les services écosystémiques a l'exception des services d'approvisionnement (Lugo, 2007). La notion de SE peut être utilisée pour faire référence à la production de services par les agriculteurs pour protéger l'environnement. Nous pouvons citer plusieurs exemples. Les rotations longues de cultures améliorent les services écosystémiques tels que les services d'aunélioration de la qualité des sols. La diversité des activités productives d'une exploitation favorise les interactions bénéfiques entre les cultures et le bétail, et la gestion des éléments du paysage tels

que les bandes enherbées, les pentes, les haies ou les cours d'eau. Elle contribue au fonctionnement écologique des agroécosystèmes.

Toutes ces définitions permettent de justifier la rémunération de ces SE comme une internalisation des externalités. Cela laisse place à une intervention publique visant à encourager leur fourniture optimale, comme la mise en place de paiements pour services environnementaux (PSE). Les PSE sont un outil utilisé aujourd'hui pour la conservation et la restauration des écosystèmes et des services qu'ils fournissent (Dasgupta, 2021).

## 7.2 Paiements pour services environnementaux

L'une des définitions les plus largement citées du PSE est celle de Wunder (2005). Il définit le PSE comme une transaction volontaire dans laquelle un SE bien défini ou une utilisation des terres pouvant produire ce service est acheté par (au moins) un acheteur de SE à (au moins) un fournisseur de SE si et seulement si le fournisseur de SE garantit la fourniture du SE. Cette conditionnalité peut être compliqué à évaluer dans les mécanismes de PSE basés sur les résultats, car certains SE sont difficiles à mesurer. Dans la pratique, il est beaucoup plus courant de mettre en place des mécanismes de PSE basés sur des actions qui sont conditionnées à des pratiques spécifiques d'utilisation ou de gestion des terres.

La définition ci-dessus illustre le théorème de Coase, selon lequel une externalité peut être internalisée par une négociation privée. Dans ce cas, l'allocation optimale des SE peut être obtenue, indépendamment de l'allocation initiale des droits de propriétés, en supposant des coûts de transaction suffisamment faibles et des droits de propriété bien définis (Coase, 1960). Un exemple de PSE coasien est le PSE de Vittel, dans le nord-est de la France. L'entreprise, subissant la pollution des aquifères par les nitrates, a conclu un accord avec les agriculteurs locaux. Il s'agissait de les indemniser pour qu'ils réduisent leur utilisation d'engrais (Bingham, 2021).

Cette définition des PSE peut être élargie pour inclure certains types d'interventions gouvernementales qui reflètent une subvention pigouvienne (Pigou, 1920; Sattler and Matzdorf, 2013). Ce type de PSE est beaucoup plus courant dans la pratique qu'un PSE coasien. Par exemple, les programmes agro-environnementaux européens sont financés par des fonds publics et le gouvernement agit en tant qu'intermédiaire entre les acheteurs de SE (le public) et les vendeurs de SE (les agriculteurs qui reçoivent les PSE). Les systèmes de PSE coasiens et pigouviens suivent tous les deux le principe du beneficiaire-payeur. Une autre définition fournie par Muradian et al. (2010) décrit les PSE comme un transfert de ressources entre acteurs, visant à créer des incitations pour aligner les décisions individuelles et/ou collectives d'utilisation des terres sur l'intérêt social en matière de gestion des ressources naturelles. Cette définition est plus souple que celle de Wunder (2005) et reflète mieux la pratique actuelle des PSE. Ainsi, les paiements ne sont pas nécessairement monétaires, mais peuvent constituer des transferts en nature.

De nombreuses formes de PSE ont été mises en œuvre tant dans les pays en développement que dans les pays développés. Aux États-Unis, le Conservation Reserve Program est en place depuis 1985 (Hellerstein, 2017). Le Costa Rica est l'un des premiers pays à avoir adopté, en 1997, un programme national de PSE (Pagiola, 2008). En Chine, le programme de conversion des terres en pente et le programme de conservation des forêts naturelles ont investi plus de 50 milliards USD entre 2000 et 2009 (Salzman et al., 2018). Le programme REDD+, conceptualisé en 2007, met en lien les pays développés et les pays en développement. Il vise à réduire les émissions de carbone dues à la déforestation et à la dégradation des forêts. Plus précisément, il incite les pays en développement à préserver leurs forêts grâce à des paiements financés par les pays développés (Chiroleu-Assouline et al., 2018). Les politiques de protection de la biodiversité se concentrent souvent sur les terres agricoles, comme les Mesures Agro-Environnementales et Climatiques (MAEC) en Europe, que les États membres sont tenus d'appliquer depuis 1992. Les MAEC sont des PSE qui rémunèrent les agriculteurs pour les actions volontaires qu'ils entreprennent en vue de préserver et d'améliorer l'environnement. Dans l'Union européenne, la préservation de la biodiversité passe essentiellement par la mise en œuvre des PSE appliqués dans le cadre de la politique agricole commune (PAC) (Herzon et al., 2018). Les pratiques adoptées comprennent la réduction des engrais et/ou des pesticides, l'établissement de bandes enherbées notamment près des rivières et le recours à la rotation de cultures. Plus récemment, dans le cadre du Plan National pour la Biodiversité 2018 adopté en France, les agences de l'eau peuvent mettre en place leurs propres programmes de PSE. Elles ont reçu 150 millions d'euros du budget national français, avec pour objectif de maintenir ou de créer de bonnes pratiques écologiques, telles que la réduction des pesticides ou la plantation de couvert végétale. Ce nouveau programme vise des PSE basés sur les résultats annuels plutôt que sur les actions. Les actions de maintien ou de création de bonnes pratiques sont toutes les deux récompensées, mais la création de bonnes pratiques permet d'obtenir des paiements plus élevés (jusqu'à 676 euros/ha/an, contre 66 euros/ha/an pour le maintien des

bonnes pratiques).

Il est difficile de classer ces différents mécanismes de PSE. Comme le soulignent Sattler et al. (2013), les PSE reposent sur une multitude d'approches qui diffèrent en termes de SE visés, de mécanismes de formation des prix, d'origines et de niveaux de paiement, de caractéristiques des acheteurs et des vendeurs ou de règles régissant le contrat entre les parties concernées. Selon Wunder (2005), les principaux SE visés par les PSE sont la séquestration et le stockage du carbone, la protection de la biodiversité, la protection des bassins versants et la beauté des paysages.

## 7.3 Motivation de la thèse

Malgré l'abondante littérature sur les PSE, de nombreuses questions de recherche restent inexplorées. Tout d'abord, il est nécessaire d'analyser la manière dont les PSE interagissent avec d'autres politiques environnementales telles que les taxes pigouviennes, et de définir dans ce contexte leur niveau optimal. Selon Bryan and Crossman (2013), l'interaction de multiples incitations financières visant la fourniture de services par les agro-écosystèmes peut réduire leur efficacité. Les mesures agroenvironnementales doivent donc tenir compte du fait que les politiques sont généralement des combinaisons de mesures et que les incitations financières pour différents services écosystémiques interagissent (Huber et al., 2017). Lankoski and Ollikainen (2003) fournissent un cadre théorique intéressant visant à définir le niveau optimal d'un PSE et d'une taxe pigouvienne dans le secteur agricole.

Une autre question peu développée dans la littérature est d'investiguer dans quelle mesure un pouvoir de marche peut modifier le niveau optimal des PSE. Ce point a largement été traite en ce qui concerne les taxes pigouviennes. En présence de pouvoir de marche, il est montre que la taxe optimale de second rang doit être inférieure au dommage marginal (Barnett, 1980; Ebert, 1991). Toutefois, aucune étude ne définit le niveau optimal des PSE en présence de concurrence imparfaite. Le pouvoir de marche conduit à un niveau de production sous-optimal, car les entreprises restreignent la production pour augmenter leurs profits. Étant donne que les taxes et les PSE influencent les niveaux de production, les PSE devraient également tenir compte des pouvoirs de marche afin de ne pas fausser davantage la production en l'éloignant du niveau socialement optimal.

En outre, lorsque les PSE sont financés par de l'argent public, il est nécessaire de lever des fonds par le biais de l'impôt, ce qui peut entraîner des distorsions dans l'économie. L'augmentation des impôts contributifs peut modifier l'allocation des ressources en influençant les décisions en matière de consommation, de travail ou d'investissement. Il semble donc important de prendre en compte le coût social des fonds publics dans notre analyse. Il s'agit d'une mesure de la perte de bien-être subie par la société du fait de la mobilisation de recettes supplémentaires pour financer les dépenses publiques. Par exemple, Browning (1976) estime le Coût Marginal Social des Fonds Publiques (MCF) de l'impôt sur le revenu du travail aux États-Unis entre 1,09 à 1,16 \$ par dollar de recettes fiscales collectées. Selon Beaud (2008), ce coût est égal à 1,2 pour la France. Ainsi, lorsque l'État collecte un euro de taxe, cette dernière coûte finalement 1,2 euro à la société. Cet aspect ne doit donc pas être négligé dans la décision de mettre en place un PSE.

Un autre facteur important conditionnant l'efficacité économique des PSE est leur caractère additionnel ou non, c'est-a-dire s'ils impliquent la fourniture d'un SE qui n'aurait pas eu lieu en l'absence de tout paiement. Les premiers PSE n'exigeaient pas l'additionnalite. Il est vrai que le contrôle de l'additionnalite peut s'avérer très coûteux. Par exemple, dans le cas du programme national du Costa Rica, l'objectif est de reconnaître et de payer pour toute fourniture de SE indépendamment de son additionnalite. Ce n'est que plus récemment que l'évaluation de l'additionnalite des programmes de PSE est devenue une préoccupation. Ce caractère est toutefois essentiel pour qu'un mécanisme de PSE atteigne un objectif environnemental et économique, tout en maintenant la confiance des investisseurs (Bennett, 2010).

Enfin, les banques de compensation peuvent être considérées comme un PSE, car elles conduisent les agents à payer pour les actions de restauration, de préservation et/ou de gestion de la biodiversité et des écosystèmes (Bureau, 2010; Salzman et al., 2018). Dans ce cas, la persévération de la biodiversité repose sur un mécanisme quantité plutôt que basé sur les prix. En effet, les banques de compensation impliquent le recours à des permis négociables. Avant de pouvoir acheter des crédits auprès d'une banque de compensation, il faut, en théorie, suivre une hiérarchie des mesures d'atténuation dont l'objectif est l'absence de perte nette de biodiversité ou de services environnementaux. Lors de la conception d'un projet qui générera des dommages environnementaux, un aménageur doit d'abord éviter tout dommage possible en modifiant le projet. Ensuite, les dommages qui ne peuvent être évités doivent être réduits. Enfin, tout dommage résiduel après évitement et réduction doit être compense, soit par la restauration d'une autre zone naturelle équivalente, soit par l'achat de permis auprès d'une banque de compensation. Cependant, dans la réalité, la hiérarchie des mesures d'atténuation n'est pas correctement respectée, affectant ainsi l'efficacité de ces PSE.

## 7.4 Principaux résultats

L'objectif de cette thèse est d'analyser ces différentes questions, afin d'approfondir les connaissances sur les PSE. Elle se compose d'une revue de la littérature sur les PSE et de trois articles originaux dans lesquels sont introduits chacun de ces points inexplorés jusqu'à présent dans la littérature économique.

La revue de la littérature analyse l'efficacité des mécanismes de PSE. De nombreux articles traitent de l'asymétrie d'information qui existe entre les acheteurs et les vendeurs de services environnementaux. Cette asymétrie remet en question l'additionnalité des PSE. Afin d'y remédier, une partie de la littérature analyse les enchères de conservation. D'autres facteurs limitant l'efficacité des PSE sont également abordés, comme les coûts de transaction, la durée des contrats, et les motivations à la conservation. Le principe d'un bonus d'agglomération est considéré afin d'améliorer l'efficacité environnementale. Finalement, les études des banques de compensation montrent que la séquence des mesures d'atténuation des dommages est difficilement respectée, affectant l'efficacité de la politique de préservation de la biodiversité.

Dans le premier article, nous avons étudié un scénario dans lequel une taxe environnementale et un système de PSE sont utilisés pour lutter contre la pollution et préserver la biodiversité, respectivement. Ce modèle examine comment des agriculteurs choisissent de répartir leurs terres entre une production agricole conventionnelle, une production agricole biologique et la mise en place de bandes enherbées. Il est supposé que la production conventionnelle cause des dommages environnementaux, que la production biologique a un impact neutre sur l'environnement et que les bandes enherbées permettent de promouvoir la biodiversité. Une distorsion supplémentaire a été ajoutée en supposant que le marché agricole biologique est organisé en oligopole. Il est montre que la taxe de second rang sur la production agricole conventionnelle est supérieure au dommage marginal causé par la pollution, et que le PSE de second rang est inférieur au bénéfice marginal de la biodiversité.

Afin de tenir compte du fait que des fonds doivent être collectés pour financer un PSE public, le MCF a été ensuite été introduit. Le PSE de second rang diminue avec le MCF, tandis que la taxe environnementale de second rang augmente avec le MCF, a condition que la demande pour le bien agricole conventionnel soit inélastique et que les politiques environnementales fournissent des bandes enherbées de manière efficace. Le premier article met en évidence une composante contributive de la taxe incitative. Nous identifions également des cas ou le PSE est inefficace pour promouvoir la biodiversité.

Ce premier article a été prolongé par l'examen d'hypothèses alternatives. Tout d'abord, nous avons remis en question le postulat d'un impact neutre de l'agriculture biologique sur la biodiversité, et nous avons supposé que les bandes enherbées produisent plus de biodiversité que l'agriculture biologique. Dans ce cas, deux PSE doivent être mis en place. Le niveau d'agriculture biologique serait alors soumis à deux effets : un effet négatif induit par l'incitation à la production des bandes enherbées et un effet positif induit par la subvention à la production. Sous nos hypothèses, le premier effet l'emporterait sur le second et les mécanismes mis en évidence dans le premier article resteraient pertinents.

Deuxièmement, maintenant nos postulats de départ, nous avons modifié les outils de politique environnementale en envisageant deux systèmes de PSE - l'un portant sur les terres non cultivées et l'autre sur l'agriculture biologique - mais sans taxe environnementale. Dans ce cas, le PSE rémunérant l'agriculture biologique tient compte du pouvoir de marche et est supérieur au bénéfice marginal de la production biologique, tandis que le PSE sur les terres non cultivées est égal au bénéfice marginal de la biodiversité et ne s'ajuste plus pour intégrer le pouvoir de marche.

Enfin, nous avons modifié l'hypothèse selon laquelle il n'y a pas d'externalités négatives de la production agricole conventionnelle sur le niveau de la production biologique. Lorsque c'est le cas, l'agriculteur internalise lui-même cet impact négatif et les niveaux de PSE et de taxes environnementales ne diffèrent pas de ceux du scénario principal expose dans la thèse.

Les budgets des programmes des PSE sont souvent limités et les régulateurs peuvent décider de conditionner les paiements aux bénéfices environnementaux additionnels obtenus en termes de biodiversité. Nous qualifions cette politique de PSE additionnels. Le deuxième article examine si ces PSE additionnels permettent d'atteindre l'efficacité environnementale. Pour ce faire, nous avons utilisé un modèle a deux périodes, dans lequel un agriculteur affecte ses terres entre une production conventionnelle génératrice de dommages environnementaux, une agriculture biologique et des bandes enherbées favorisant la biodiversité. Une taxe environnementale est appliquée sur la production traditionnelle et un PSE sur les bandes enherbées, ne rémunérant que les bénéfices environnementaux supplémentaires résultant du PSE.

Nous montrons que ce PSE base sur l'additionnalite introduit une distorsion dans le comportement de l'agriculteur au cours de la période initiale. L'agriculteur augmente ses niveaux de production afin d'obtenir un paiement plus important au cours de la période finale. Le PSE de second rang doit corriger cette distorsion tout en prenant en compte les bénéfices et les dommages environnementaux. Au final, le PSE de second rang est égal a la différence actualisée des bénéfices environnementaux marginaux obtenus a chaque période. Les taxes environnementales de second rang pour chaque période ne sont plus égales au dommage marginal. Elles sont ajustées pour corriger les distorsions induites par le PSE.

Nous avons ensuite introduit une position dominante sur le marché de l'agriculture biologique. Si le pouvoir de marche réduit le niveau de production biologique dans la période finale, il limite la distorsion dans la période initiale. Selon l'ampleur de ces deux effets, le PSE additionnel de second rang augmente ou diminue par rapport au scénario sans pouvoir de marche. Les taxes sont ajustées pour tenir compte des effets indirects du pouvoir de marche sur le niveau de la production agricole conventionnelle.

Finalement, cette deuxième étude a montré que le caractère additionnel des PSE ne permet pas d'atteindre l'efficacité environnementale, même sous des hypothèses de concurrence parfaite. Elle permet également de mieux comprendre les interactions entre les différents types de politiques environnementales.

S'agissant d'un PSE basé sur la quantité, nous avons examiné les banques de compensation et, en particulier, la hiérarchie d'atténuation ou la séquence "éviter, réduire, compenser" (ERC) dans le troisième étude. Bigard et al. (2018) soulignent les difficultés à traduire les concepts de la séquence ERC dans la pratique. Selon ces auteurs, les qualifications données dans les études d'impact ne correspondent pas, la plupart du temps, aux définitions nationales de référence. Ils notent notamment une confusion entre les mesures d'évitement et de réduction. À la suite de Bigard et al. (2018), notre travail a analysé la dimension opérationnelle de la séquence ERC, mais en mobilisant un autre angle d'étude. Nous avons cherché à intégrer la séquence ERC dans un modèle de décision économique, avec l'objectif d'étudier sa rationalité. Pour ce faire, nous avons posé les hypothèses d'information parfaite et asymétrique.

Nous avons d'abord mis en évidence les difficultés de transposer les impératifs de la séquence dans un modèle de décision économique. La séquence ERC correspond à la mise en œuvre successive de trois normes : éviter, réduire et compenser. À l'étape "éviter", le projet le moins dommageable pour l'environnement doit être selectionné, ce qui implique l'existence de plusieurs projets ou modalités de projet. L'étape "réduire" pose la question de la signification de "suffisamment réduit" à "un coût raisonnable" que l'on trouve dans la définition de la séquence. Tel quel, ce point est donc difficilement modélisable. Nous avons défini cette étape comme le niveau le plus élevé de réduction des dommages compatible avec un bénéfice de l'aménageur non négatif. Nous montrons qu'en information parfaite, la séquence ERC est forcément respectée. Dans ce cas, la demande de compensation ne dépend pas de son prix.

La littérature en l'économie de l'environnement souligne l'importance de l'information dans la mise en œuvre des normes environnementales en tant que moyen de lutte contre la pollution. Cependant, la séquence va encore plus loin dans l'exigence d'information que les normes de pollution car elle implique une mise en œuvre séquentielle. Nous levons l'hypothèse d'information parfaite en supposant que le régulateur ne dispose pas de toutes les informations sur les projets. Dans ce cas, nous montrons que l'information est utilisée de manière stratégique par l'amenageur et que la séquence ERC est inopérante. L'amenageur ne se comportera pas de manière myope en considérant les différentes étapes indépendamment les unes des autres. Au contraire, en tant qu'homo-œconomicus, il effectura un raisonnement à rebours afin de choisir le projet le plus rentable, sous la contrainte de la politique d'absence de perte nette de biodiversité. Nous montrons que l'amenageur décidera simultanément du montant de la réduction des dommages et de la compensation en se basant sur le prix de la compensation et que, finalement, le choix du projet dépendra aussi de ce prix. En d'autres termes, l'introduction de la séquence ERC dans la loi impliquant la politique d'absence de perte nette est inefficace lorsque l'information est asymétrique.

Cette troisième étude met en évidence les différents objectifs d'une politique de conservation de la biodiversité, soit la protection stricte de la biodiversité, soit la mise en œuvre du critère du moindre coût. La séquence ERC couplée à l'objectif de "pas de perte nette" est une séquence de normes visant à préserver la biodiversité. Le mécanisme de compensation, en fixant un prix pour la compensation, peut être considéré comme une incitation économique à compenser. Si chaque aménageur réduit ses dommages en égalisant le coût marginal de la réduction des dommages au prix de la compensation, la biodiversité sera sauvegardée à moindre coût. Ce résultat n'est pas atteignable en appliquant la séquence ERC, mais c'est peut-être le prix à payer pour protéger un bien très particulier et difficilement mesurable, la biodiversité.

#### 7.5 Limites et pistes de recherche futures

Nos analyses théoriques des politiques de PSE présentent des limites, comme toutes études. Bien que nous ayons déjà exploré certaines hypothèses alternatives dans le modèle de la première étude, celui-ci pourrait également être étendu pour prendre en compte des demandes différenciées pour les biens agricoles. Une autre extension consisterait à examiner le cas où les agriculteurs sont hétérogènes. Les PSE auraient aussi pu être définis lorsque la production de biens agricoles conventionnels induit des externalités négatives sur la production de biens agricoles biologiques sans que ces dernières puissent être directement internalisées. En outre, cette première étude utilise un modèle relativement simple, dans lequel le seul moyen d'augmenter les bénéfices de biodiversité est de mettre en place des bandes enherbées. Cette mesure est supposée n'entraîner que des coûts d'opportunité liés a l'absence de production. Toutefois, les PSE existants rémunèrent d'autres movens de protéger la biodiversité, générant leur propre coût. On peut citer l'exemple de plantation de couverts végétaux. De futures recherches pourraient étudier la manière dont l'agriculteur choisit la ou les pratiques de conservation et l'impact de ces mesures sur le niveau de fourniture de services environnementaux.

En ce qui concerne la deuxième étude, la caracterisation des PSE basés sur les avantages environnementaux supplémentaires obtenus par le paiement s'avère complexe dans un cadre simple d'information parfaite avec deux périodes de temps. Cependant, elle pourrait être étendue en considérant un horizon temporel infini, afin d'analyser si les résultats sont maintenus sous cette nouvelle hypothèse. Il s'agirait alors de modéliser le comportement stratégique dans un modèle de contrôle optimal. Pars ailleurs, dans la même veine que Barnett (1980), nous avons défini les politiques environnementales de second rang en information parfaite, ce qui suggère que le régulateur connaît les coûts de production de l'entreprise. En mobilisant la théorie de l'agence, ces travaux pourraient être étendus lorsque l'information est asymétrique, ce qui permettrait de prendre en compte l'aléa moral. Enfin, le modèle pourrait être enrichi en introduisant le MCF.

La troisième étude a examiné la rationalité de la séquence ERC. Cependant, cette analyse n'a pas pris en compte l'existence d'un autre opérateur, les auditeurs. En effet, la séquence ERC est réalisée par le biais d'études d'impact. Ces dernières sont à la charge du promoteur, qui peut faire appel à des bureaux d'études pour rédiger ce document entraîneant ainsi un coût supplémentaire. Face au flou opérationnel de la séquence ERC, on peut imaginer que le bureau d'études mène l'étude d'impact en éludant certaines informations ou en minimisant certains impacts, ce qui justifie notre hypothèse d'asymétrie d'information. Une façon de surmonter ce problème d'information serait de créer une agence publique spécialisée dont la mission serait de réaliser l'ensemble de ces études d'impact. Cette agence bénéficierait de l'expérience acquise d'une étude à l'autre et serait plus à même d'appliquer la même interprétation de la séquence ERC à chaque projet. La généralisation de la séquence en France à tous les projets quels que soient leur localisation et leur taille permettrait ces économies d'échelle.

Dans notre étude, nous avons considéré plusieurs projets possibles avec différents niveaux de dommages. Dans la réalité, les aménageurs ne considèrent qu'un seul projet, d'où les difficultés d'interprétation de l'étape "éviter". On pourrait imaginer que l'amenageur soit obligé de communiquer à cette agence publique spécialisée ses différents projets afin que la séquence ERC soit appliquée en amont des décisions. Au final, la création de cette agence permetrait de rendre opérante la séquence ERC.

Enfin, nous avons supposé un prix exogène pour la pratique de la compensation. Lorsque l'offre de compensation n'est pas suffisamment développée, cette hypothèse peut être remise en cause. Ce travail peut être étendu en introduisant un prix de la compensation établi sur un marché de gré à gré.

# References

- Albers, H. J., Ando, A. W., and Batz, M. (2008). Patterns of multi-agent land conservation: crowding in/out, agglomeration, and policy. *Resource and Energy Economics*, 30(4):492–508.
- Albers, H. J., Lee, K., Sims, K., et al. (2018). Economics of habitat fragmentation: a review and critique of the literature. *International Review of Environmental* and Resource Economics, 11(2):97–144.
- Arguedas, C. and van Soest, D. P. (2011). Optimal conservation programs, asymmetric information and the role of fixed costs. *Environmental and Resource Economics*, 50(2):305.
- Arlidge, W. N., Bull, J. W., Addison, P. F., Burgass, M. J., Gianuca, D., Gorham, T. M., Jacob, C., Shumway, N., Sinclair, S. P., Watson, J. E., et al. (2018). A global mitigation hierarchy for nature conservation. *BioScience*, 68(5):336–347.
- Armsworth, P. R., Acs, S., Dallimer, M., Gaston, K. J., Hanley, N., and Wilson, P. (2012). The cost of policy simplification in conservation incentive programs. *Ecology letters*, 15(5):406–414.
- Arnold, M. A., Duke, J. M., and Messer, K. D. (2013). Adverse selection in reverse auctions for ecosystem services. *Land Economics*, 89(3):387–412.
- Aubry, S., Gaucherand, S., Spiegelberger, T., Padilla, B., Pioch, S., and Vaissière, A.-C. (2021). Place des sites naturels de compensation dans la séquence «éviterréduire-compenser» et panorama des problématiques associées. *Sciences Eaux Territoires*, 38(4):4–9.
- Bamière, L., David, M., and Vermont, B. (2013). Agri-environmental policies for biodiversity when the spatial pattern of the reserve matters. *Ecological Economics*, 85:97–104.
- Banerjee, S., Cason, T. N., de Vries, F. P., and Hanley, N. (2017). Transaction costs, communication and spatial coordination in payment for ecosystem services schemes. *Journal of Environmental Economics and Management*, 83:68–89.

- Banerjee, S., de Vries, F. P., and Hanley, N. (2011). The agglomeration bonus in the presence of private transaction costs. In 13th BIOECON conference, pages 11–13.
- Barnett, A. H. (1980). The pigouvian tax rule under monopoly. *The American Economic Review*, 70(5):1037–1041.
- Batáry, P., Dicks, L. V., Kleijn, D., and Sutherland, W. J. (2015). The role of agri-environment schemes in conservation and environmental management. *Conservation Biology*, 29(4):1006–1016.
- Beaud, M. (2008). Le coût social marginal des fonds publics en france. Annales d'Economie et de Statistique, pages 215–232.
- Beesley, K. and Ramsey, D. (2009). Agricultural land preservation, pages 65–69.
- Bell, A., Parkhurst, G., Droppelmann, K., and Benton, T. G. (2016). Scaling up pro-environmental agricultural practice using agglomeration payments: Proof of concept from an agent-based model. *Ecological Economics*, 126:32–41.
- Ben Arfa, N., Rodriguez, C., and Daniel, K. (2009). Dynamiques spatiales de la production agricole en france. *Revue dEconomie Regionale Urbaine*, (4):807–834.
- Bennett, K. (2010). Additionality: The next step for ecosystem service markets. Duke Environmental Law & Policy F., 20:417.
- Bigard, C., Regnery, B., Pioch, S., and Thompson, J. D. (2018). De la théorie à la pratique de la séquence éviter-réduire-compenser (erc): éviter ou légitimer la perte de biodiversité? Développement durable et territoires. Économie, géographie, politique, droit, sociologie, 11(2).
- Bingham, L. R. (2021). Vittel as a model case in pes discourse: Review and critical perspective. *Ecosystem Services*, 48:101247.
- Bjørkhaug, H. and Blekesaune, A. (2013). Development of organic farming in norway: A statistical analysis of neighbourhood effects. *Geoforum*, 45:201–210.
- Börner, J., Baylis, K., Corbera, E., Ezzine-de Blas, D., Honey-Rosés, J., Persson, U. M., and Wunder, S. (2017). The effectiveness of payments for environmental services. World development, 96:359–374.
- Brown, P. and Land, C. (1999). The effect of wetland mitigation banking on the achievement of no-net-loss. *Environmental management*, 23(3):333–345.

- Browning, E. K. (1976). The marginal cost of public funds. *Journal of Political Economy*, 84(2):283–298.
- Bryan, B. A. and Crossman, N. D. (2013). Impact of multiple interacting financial incentives on land use change and the supply of ecosystem services. *Ecosystem Services*, 4:60–72.
- Buchanan, J. M. (1969). External diseconomies, corrective taxes, and market structure. *The American Economic Review*, 59(1):174–177.
- Bull, J. W. and Strange, N. (2018). The global extent of biodiversity offset implementation under no net loss policies. *Nature Sustainability*, 1(12):790–798.
- Bureau, D. (2010). Les «PSE»: des rémunérations pour les services environnementaux. *Conseil Economique pour le Developpement Durable*, pages 1–8.
- Bureau, D. and Mougeot, M. (2005). Politique environnementale en économie ouverte. *Revue d'économie politique*, (4):441–450.
- Burgin, S. (2010). 'mitigation banks' for wetland conservation: a major success or an unmitigated disaster? Wetlands Ecology and Management, 18(1):49–55.
- Burton, R. J., Kuczera, C., and Schwarz, G. (2008). Exploring farmers' cultural resistance to voluntary agri-environmental schemes. *Sociologia ruralis*, 48(1):16–37.
- Calvet, C., Le Coent, P., Napoleone, C., and Quétier, F. (2019). Challenges of achieving biodiversity offset outcomes through agri-environmental schemes: Evidence from an empirical study in southern france. *Ecological economics*, 163:113–125.
- Campbell, D. A., Cole, C. A., and Brooks, R. P. (2002). A comparison of created and natural wetlands in pennsylvania, usa. Wetlands Ecology and Management, 10(1):41–49.
- Cason, T. N. and Gangadharan, L. (2004). Auction design for voluntary conservation programs. American Journal of Agricultural Economics, 86(5):1211–1217.
- Chabé-Ferret, S. and Subervie, J. (2013). How much green for the buck? estimating additional and windfall effects of french agro-environmental schemes by didmatching. Journal of Environmental Economics and Management, 65(1):12–27.
- Chiroleu-Assouline, M., Poudou, J.-C., and Roussel, S. (2018). Designing redd+ contracts to resolve additionality issues. *Resource and Energy Economics*, 51:1–17.

- Claassen, R., Cattaneo, A., and Johansson, R. (2008). Cost-effective design of agri-environmental payment programs: Us experience in theory and practice. *Ecological economics*, 65(4):737–752.
- Clare, S., Krogman, N., Foote, L., and Lemphers, N. (2011). Where is the avoidance in the implementation of wetland law and policy? Wetlands Ecology and Management, 19(2):165–182.
- Clarke, E. H. (1971). Multipart pricing of public goods. Public choice, pages 17–33.
- Coase, R. H. (1960). The problem of social cost. In *Classic Papers in Natural Resource Economics*, pages 87–137. Springer.
- Combe, M. (2020). Instruments Économiques et Protection de la Biodiversité: Analyse juridique des mécanismes de compensation écologique et de paiements pour services environnementaux. PhD thesis, l'Université Jean Moulin Lyon 3.
- Convention on Biological Diversity (2010). Global biodiversity outlook 3. In Montréal, Canada: Secretariat of the Convention on Biological Diversity. (http://gbo3. cbd. int/) Phil. Trans. R. Soc. B, volume 9.
- Costanza, R., d'Arge, R., De Groot, R., Farber, S., Grasso, M., Hannon, B., Limburg, K., Naeem, S., O'neill, R. V., Paruelo, J., et al. (1997). The value of the world's ecosystem services and natural capital. *nature*, 387(6630):253–260.
- Costanza, R., De Groot, R., Braat, L., Kubiszewski, I., Fioramonti, L., Sutton, P., Farber, S., and Grasso, M. (2017). Twenty years of ecosystem services: how far have we come and how far do we still need to go? *Ecosystem services*, 28:1–16.
- CPES (2020). Protection of the resource: Eau de paris launches its own agricultural aid scheme. Last accessed 17 March 2023.
- Dahl, T. E. (1990). Wetlands losses in the United States, 1780's to 1980's. US Department of the Interior, Fish and Wildlife Service.
- Dahlby, B. (2008). The marginal cost of public funds: Theory and applications. MIT press.
- Damiens, F. L., Backstrom, A., and Gordon, A. (2021). Governing for "no net loss" of biodiversity over the long term: challenges and pathways forward. One Earth, 4(1):60–74.
- Dasgupta, P. (2021). The Economics of Biodiversity: the Dasgupta Review. HM Treasury.

- Dayer, A. A., Lutter, S. H., Sesser, K. A., Hickey, C. M., and Gardali, T. (2018). Private landowner conservation behavior following participation in voluntary incentive programs: Recommendations to facilitate behavioral persistence. *Conservation Letters*, 11(2):e12394.
- de Muelenaere, L. (2011). Les permis échangeables, une alternative crédible aux politiques de conservation de la biodiversité.
- De Snoo, G. R., Herzon, I., Staats, H., Burton, R. J., Schindler, S., van Dijk, J., Lokhorst, A. M., Bullock, J. M., Lobley, M., Wrbka, T., et al. (2013). Toward effective nature conservation on farmland: making farmers matter. *Conservation Letters*, 6(1):66–72.
- Deci, E. L., Koestner, R., and Ryan, R. M. (1999). A meta-analytic review of experiments examining the effects of extrinsic rewards on intrinsic motivation. *Psychological bulletin*, 125(6):627.
- Delacote, P., Robinson, E. J., and Roussel, S. (2016). Deforestation, leakage and avoided deforestation policies: A spatial analysis. *Resource and Energy Economics*, 45:192–210.
- Díaz, S., Settele, J., Brondízio, E., Ngo, H., Guèze, M., Agard, J., Arneth, A., Balvanera, P., Brauman, K., Butchart, S., et al. (2019). Summary for policymakers of the global assessment report on biodiversity and ecosystem services of the intergovernmental science-policy platform on biodiversity and ecosystem services. United Nations: Paris, France, pages 1–39.
- Drechsler, M. (2017). The impact of fairness on side payments and cost-effectiveness in agglomeration payments for biodiversity conservation. *Ecological Economics*, 141:127–135.
- Dutoit, T., Jaunatre, R., Alignan, J.-F., Bulot, A., Wolff, A., and Buisson, É. (2018). Regards-focus regards d'écologues sur le premier site naturel de compensation français. *Natures Sciences Sociétés*, 26(2):215–222.
- Duval, L., Binet, T., Dupraz, P., Leplay, S., Etrillard, C., Pech, M., Deniel, E., and Laustriat, M. (2016). Paiements pour services environnementaux et méthodes d'évaluation économique. Enseignements pour les mesures agro-environnementales de la politique agricole commune. PhD thesis, Ministère de l'Agriculture, de l'Agroalimentaire et de la Forêt.
- Ebert, U. (1991). Pigouvian tax and market structure: The case of oligopoly and

different abatement technologies. *FinanzArchiv/Public Finance Analysis*, pages 154–166.

- Ezzine-de Blas, D., Wunder, S., Ruiz-Pérez, M., and Moreno-Sanchez, R. d. P. (2016). Global patterns in the implementation of payments for environmental services. *PLoS One*, 11(3):1–16.
- FAO (2007). The state of food and agriculture, 2007: paying farmers for environmental services. rome. Food & Agricultural Org.
- Ferraro, P. J. (2008). Asymmetric information and contract design for payments for environmental services. *Ecological economics*, 65(4):810–821.
- Ferré, M., Engel, S., and Gsottbauer, E. (2018). Which agglomeration payment for a sustainable management of organic soils in switzerland? an experiment accounting for farmers' cost heterogeneity. *Ecological Economics*, 150:24–33.
- Fooks, J. R., Higgins, N., Messer, K. D., Duke, J. M., Hellerstein, D., and Lynch, L. (2016). Conserving spatially explicit benefits in ecosystem service markets: Experimental tests of network bonuses and spatial targeting. *American Journal* of Agricultural Economics, 98(2):468–488.
- Freemark, K. E. and Kirk, D. A. (2001). Birds on organic and conventional farms in ontario: partitioning effects of habitat and practices on species composition and abundance. *Biological Conservation*, 101(3):337–350.
- Fuller, R., Norton, L., Feber, R., Johnson, P., Chamberlain, D. E., Joys, A. C., Mathews, F., Stuart, R., Townsend, M., Manley, W., et al. (2005). Benefits of organic farming to biodiversity vary among taxa. *Biology letters*, 1(4):431–434.
- Gerling, C., Sturm, A., and Wätzold, F. (2019). Ecological-economic modelling to compare the impact of organic and conventional farming on endangered grassland bird and butterfly species. *Agricultural Systems*, 173:424–434.
- Glebe, T. W. (2013). Conservation auctions: should information about environmental benefits be made public? American Journal of Agricultural Economics, 95(3):590– 605.
- Gómez-Baggethun, E. and Ruiz-Pérez, M. (2011). Economic valuation and the commodification of ecosystem services. *Progress in physical geography*, 35(5):613– 628.
- Grout, C. A. (2009). Incentives for spatially coordinated land conservation: a conditional agglomeration bonus. In *Western Economics Forum*, volume 8, page 21.
- Groves, T. (1973). Incentives in teams. *Econometrica: Journal of the Econometric Society*, pages 617–631.
- Hart, R. and Latacz-Lohmann, U. (2005). Combating moral hazard in agrienvironmental schemes: a multiple-agent approach. European Review of Agricultural Economics, 32(1):75–91.
- Hellerstein, D. M. (2017). The us conservation reserve program: The evolution of an enrollment mechanism. *Land Use Policy*, 63:601–610.
- Herzon, I., Birge, T., Allen, B., Povellato, A., Vanni, F., Hart, K., Radley, G., Tucker, G., Keenleyside, C., Oppermann, R., et al. (2018). Time to look for evidence: Results-based approach to biodiversity conservation on farmland in Europe. Land use policy, 71:347–354.
- Hiron, M., Berg, Å., Eggers, S., Josefsson, J., and Pärt, T. (2013). Bird diversity relates to agri-environment schemes at local and landscape level in intensive farmland. Agriculture, ecosystems & environment, 176:9–16.
- Hole, D. G., Perkins, A., Wilson, J., Alexander, I., Grice, P., and Evans, A. D. (2005). Does organic farming benefit biodiversity? *Biological conservation*, 122(1):113–130.
- Howlett, M. and Rayner, J. (2013). Patching vs packaging in policy formulation: Assessing policy portfolio design. *Politics and Governance*, 1(2):170–182.
- Huber, R., Rebecca, S., François, M., Hanna, B. S., Dirk, S., and Robert, F. (2017). Interaction effects of targeted agri-environmental payments on non-marketed goods and services under climate change in a mountain region. *Land Use Policy*, 66:49–60.
- Jones, K. W., Mayer, A., Von Thaden, J., Berry, Z. C., López-Ramírez, S., Salcone, J., Manson, R. H., and Asbjornsen, H. (2020). Measuring the net benefits of payments for hydrological services programs in mexico. *Ecological Economics*, 175:106666.
- Kaczan, D., Pfaff, A., Rodriguez, L., and Shapiro-Garza, E. (2017). Increasing the impact of collective incentives in payments for ecosystem services. *Journal of Environmental Economics and Management*, 86:48–67.

- Kawasaki, K., Fujie, T., Koito, K., Inoue, N., and Sasaki, H. (2012). Conservation auctions and compliance: theory and evidence from laboratory experiments. *Environmental and Resource Economics*, 52(2):157–179.
- Kemigisha, E., Babweteera, F., Mugisha, J., and Angelsen, A. (2023). Payment for environmental services to reduce deforestation: Do the positive effects last? *Ecological Economics*, 209:107840.
- Kiesecker, J. M., Copeland, H., Pocewicz, A., and McKenney, B. (2010). Development by design: blending landscape-level planning with the mitigation hierarchy. *Frontiers in Ecology and the Environment*, 8(5):261–266.
- Kits, G. J., Adamowicz, W. L., and Boxall, P. C. (2014). Do conservation auctions crowd out voluntary environmentally friendly activities? *Ecological Economics*, 105:118–123.
- Kleijn, D. and Sutherland, W. J. (2003). How effective are european agri-environment schemes in conserving and promoting biodiversity? *Journal of applied ecology*, 40(6):947–969.
- Kosoy, N. and Corbera, E. (2010). Payments for ecosystem services as commodity fetishism. *Ecological economics*, 69(6):1228–1236.
- Kozich, A. T. and Halvorsen, K. E. (2012). Compliance with wetland mitigation standards in the upper peninsula of michigan, usa. *Environmental Management*, 50:97–105.
- Krämer, J. E. and Wätzold, F. (2018). The agglomeration bonus in practice an exploratory assessment of the swiss network bonus. *Journal for nature conservation*, 43:126–135.
- Krautkraemer, A. and Schwartz, S. (2022). Payment for environmental services and pollution tax under imperfect competition. *Working paper LEO-UCA*.
- Kuhfuss, L., Préget, R., Thoyer, S., and Hanley, N. (2016a). Nudging farmers to enrol land into agri-environmental schemes: the role of a collective bonus. *European Review of Agricultural Economics*, 43(4):609–636.
- Kuhfuss, L., Préget, R., Thoyer, S., Hanley, N., Le Coent, P., and Désolé, M. (2016b). Nudges, social norms, and permanence in agri-environmental schemes. *Land economics*, 92(4):641–655.
- Lampach, N., Nguyen-Van, P., and To-The, N. (2020). Robustness analysis of

organic technology adoption: evidence from northern vietnamese tea production. European Review of Agricultural Economics, 47(2):529–557.

- Lankoski, J. and Ollikainen, M. (2003). Agri-environmental externalities: A framework for designing targeted policies. *European Review of Agricultural Economics*, 30(1):51–75.
- Latacz-Lohmann, U. and Van der Hamsvoort, C. (1997). Auctioning conservation contracts: a theoretical analysis and an application. American Journal of Agricultural Economics, 79(2):407–418.
- Latacz-Lohmann, U. and Van der Hamsvoort, C. P. (1998). Auctions as a means of creating a market for public goods from agriculture. *Journal of Agricultural Economics*, 49(3):334–345.
- Le Coent, P., Préget, R., and Thoyer, S. (2017). Compensating environmental losses versus creating environmental gains: implications for biodiversity offsets. *Ecological Economics*, 142:120–129.
- Levrel, H., Guillet, F., Lombard-Latune, J., Delforge, P., and Frascaria-Lacoste, N. (2018). Application de la séquence éviter-réduire-compenser en france: le principe d'additionnalité mis à mal par 5 dérives. VertigO: la revue électronique en sciences de l'environnement, 18(2).
- Lewis, D. J., Plantinga, A. J., Nelson, E., and Polasky, S. (2011). The efficiency of voluntary incentive policies for preventing biodiversity loss. *Resource and Energy Economics*, 33(1):192–211.
- Lewis, D. J. and Polasky, S. (2018). An auction mechanism for the optimal provision of ecosystem services under climate change. *Journal of Environmental Economics* and Management, 92:20–34.
- Liu, Z., Xu, J., Yang, X., Tu, Q., Hanley, N., and Kontoleon, A. (2019). Performance of agglomeration bonuses in conservation auctions: Lessons from a framed field experiment. *Environmental and Resource Economics*, pages 1–27.
- Luck, G. W., Chan, K. M., Eser, U., Gómez-Baggethun, E., Matzdorf, B., Norton, B., and Potschin, M. B. (2012). Ethical considerations in on-ground applications of the ecosystem services concept. *BioScience*, 62(12):1020–1029.
- Lugo, E. (2007). Ecosystem services, the millennium ecosystem assessment, and the conceptual difference between benefits provided by ecosystems and benefits provided by people. J. Land Use & Envtl. L., 23:243.

- Lundberg, L., Persson, U. M., Alpizar, F., and Lindgren, K. (2018). Context matters: exploring the cost-effectiveness of fixed payments and procurement auctions for pes. *Ecological Economics*, 146:347–358.
- Marja, R., Herzon, I., Viik, E., Elts, J., Mänd, M., Tscharntke, T., and Batáry, P. (2014). Environmentally friendly management as an intermediate strategy between organic and conventional agriculture to support biodiversity. *Biological Conservation*, 178:146–154.
- Martin, S. and Brumbaugh, R. (2013). Defining service areas for wetland mitigation: an overview. *Wetlands Newsletter*, 35(2):9.
- Mason, C. F. and Plantinga, A. J. (2013). The additionality problem with offsets: Optimal contracts for carbon sequestration in forests. *Journal of Environmental Economics and Management*, 66(1):1–14.
- Mezzatesta, M., Newburn, D. A., and Woodward, R. T. (2013). Additionality and the adoption of farm conservation practices. *Land Economics*, 89(4):722–742.
- Mirrlees, J. A. (1971). An exploration in the theory of optimum income taxation. The review of economic studies, 38(2):175–208.
- Mitsch, W. J. and Wilson, R. F. (1996). Improving the success of wetland creation and restoration with know-how, time, and self-design. *Ecological applications*, 6(1):77–83.
- Mohebalian, P. M. and Aguilar, F. X. (2016). Additionality and design of forest conservation programs: Insights from ecuador's socio bosque program. *Forest Policy and Economics*, 71:103–114.
- Moilanen, A., Van Teeffelen, A. J., Ben-Haim, Y., and Ferrier, S. (2009). How much compensation is enough? a framework for incorporating uncertainty and time discounting when calculating offset ratios for impacted habitat. *Restoration Ecology*, 17(4):470–478.
- Moreno-Mateos, D., Power, M. E., Comín, F. A., and Yockteng, R. (2012). Structural and functional loss in restored wetland ecosystems. *PLoS biology*, 10(1):e1001247.
- Moxey, A., White, B., and Ozanne, A. (1999). Efficient contract design for agrienvironment policy. *Journal of agricultural economics*, 50(2):187–202.
- MTES (2019). Biodiversity action plan. Last accessed 17 March 2023.
- MTES (2021). Plan biodiversité. Last accessed 17 March 2023.

- Muradian, R., Arsel, M., Pellegrini, L., Adaman, F., Aguilar, B., Agarwal, B., Corbera, E., Ezzine de Blas, D., Farley, J., Froger, G., et al. (2013). Payments for ecosystem services and the fatal attraction of win-win solutions. *Conservation letters*, 6(4):274–279.
- Muradian, R., Corbera, E., Pascual, U., Kosoy, N., and May, P. H. (2010). Reconciling theory and practice: An alternative conceptual framework for understanding payments for environmental services. *Ecological Economics*, 69(6):1202–1208.
- Needham, K., de Vries, F. P., Armsworth, P. R., and Hanley, N. (2019). Designing markets for biodiversity offsets: Lessons from tradable pollution permits. *Journal* of Applied Ecology, 56(6):1429–1435.
- Nguyen-Van, P., Stenger, A., and Veron, E. (2021). Spatial factors influencing the territorial gaps of organic farming in france.
- Ola, O., Menapace, L., Benjamin, E., and Lang, H. (2019). Determinants of the environmental conservation and poverty alleviation objectives of payments for ecosystem services (pes) programs. *Ecosystem services*, 35:52–66.
- Oregon Watershed Enhancement Board (2018). Oregon crep annual report. https://www.oregon.gov/oweb/Documents/CREP-Annual-Report-2018.pdf. Accessed September 12, 2023.
- Pagiola, S. (2008). Payments for environmental services in costa rica. *Ecological economics*, 65(4):712–724.
- Parkhurst, G. M. and Shogren, J. F. (2007). Spatial incentives to coordinate contiguous habitat. *Ecological economics*, 64(2):344–355.
- Parkhurst, G. M. and Shogren, J. F. (2008). Smart subsidies for conservation. American Journal of Agricultural Economics, 90(5):1192–1200.
- Parkhurst, G. M., Shogren, J. F., Bastian, C., Kivi, P., Donner, J., and Smith, R. B. (2002). Agglomeration bonus: an incentive mechanism to reunite fragmented habitat for biodiversity conservation. *Ecological economics*, 41(2):305–328.
- Pates, N. J. and Hendricks, N. P. (2020). Additionality from payments for environmental services with technology diffusion. *American Journal of Agricultural Economics*, 102(1):281–299.
- Pattanayak, S. K., Wunder, S., and Ferraro, P. J. (2010). Show me the money:

do payments supply environmental services in developing countries? Review of environmental economics and policy, 4(2):254-274.

- Pe'er, G., Dicks, L. V., Visconti, P., Arlettaz, R., Báldi, A., Benton, T. G., Collins, S., Dieterich, M., Gregory, R. D., Hartig, F., Henle, K., Hobson, P. R., Kleijn, D., Neumann, R. K., Robijns, T., Schmidt, J., Shwartz, A., Sutherland, W. J., Turbé, A., Wulf, F., and Scott, A. V. (2014). EU agricultural reform fails on biodiversity. *Science*, 344(6188):1090–1092.
- Phalan, B., Hayes, G., Brooks, S., Marsh, D., Howard, P., Costelloe, B., Vira, B., Kowalska, A., and Whitaker, S. (2018). Avoiding impacts on biodiversity through strengthening the first stage of the mitigation hierarchy. *Oryx*, 52(2):316–324.
- Pigou, A. (1920). The Economics of Welfare. Macmillan, London.
- Piha, M., Tiainen, J., Holopainen, J., and Vepsäläinen, V. (2007). Effects of land-use and landscape characteristics on avian diversity and abundance in a boreal agricultural landscape with organic and conventional farms. *Biological Conservation*, 140(1-2):50–61.
- Polasky, S., Lewis, D. J., Plantinga, A. J., and Nelson, E. (2014). Implementing the optimal provision of ecosystem services. *Proceedings of the National Academy of Sciences*, 111(17):6248–6253.
- Princé, K. and Jiguet, F. (2013). Ecological effectiveness of french grassland agrienvironment schemes for farmland bird communities. *Journal of environmental* management, 121:110–116.
- Pullin, A. S. and Knight, T. M. (2009). Doing more good than harm-building an evidence-base for conservation and environmental management. *Biological* conservation, 142(5):931–934.
- Purtauf, T., Roschewitz, I., Dauber, J., Thies, C., Tscharntke, T., and Wolters, V. (2005). Landscape context of organic and conventional farms: influences on carabid beetle diversity. Agriculture, Ecosystems & Environment, 108(2):165–174.
- Quétier, F., Regnery, B., and Levrel, H. (2014). No net loss of biodiversity or paper offsets? a critical review of the french no net loss policy. *Environmental Science* & Policy, 38:120–131.
- Rahmann, G. (2011). Biodiversity and organic farming: what do we know? vTI Agriculture and Forstery Research, 3:189–208.

- Reid, W., Mooney, H., Cropper, A., Capistrano, D., Carpenter, S., Chopra, K., Dasgupta, P., Dietz, T., Duraiappah, A., Hassan, R., Kasperson, R., Leemans, R., May, R., McMichael, A., Pingali, P., Samper, C., Scholes, R., Watson, R., Zakri, A., Shidong, Z., Ash, N., Bennett, E., Kumar, P., Lee, M., Raudsepp-Hearne, C., Simons, H., Thonell, J., and Zurek, M. (2005). *Ecosystems and human well-being* Synthesis: A Report of the Millennium Ecosystem Assessment. Island Press.
- Reiss, K. C., Hernandez, E., and Brown, M. T. (2009). Evaluation of permit success in wetland mitigation banking: a florida case study. *Wetlands*, 29(3):907–918.
- Rode, J., Gómez-Baggethun, E., and Krause, T. (2015). Motivation crowding by economic incentives in conservation policy: A review of the empirical evidence. *Ecological Economics*, 117:270–282.
- Rodriguez-Sickert, C., Guzmán, R. A., and Cárdenas, J. C. (2008). Institutions influence preferences: Evidence from a common pool resource experiment. *Journal of Economic Behavior & Organization*, 67(1):215–227.
- Ruckelshaus, M. H., Jackson, S. T., Mooney, H. A., Jacobs, K. L., Kassam, K.-A. S., Arroyo, M. T., Báldi, A., Bartuska, A. M., Boyd, J., Joppa, L. N., et al. (2020). The ipbes global assessment: Pathways to action. *Trends in Ecology & Evolution*, 35(5):407–414.
- Salzman, J., Bennett, G., Carroll, N., Goldstein, A., and Jenkins, M. (2018). The global status and trends of payments for ecosystem services. *Nature Sustainability*, 1(3):136–144.
- Santos, E. C., Storm, H., and Rasch, S. (2021). The cost-effectiveness of conservation auctions in the presence of asset specificity: An agent-based model. *Land Use Policy*, 102:104907.
- Sattler, C. and Matzdorf, B. (2013). PES in a nutshell: From definitions and origins to PES in practice—approaches, design process and innovative aspects. *Ecosystem* services, 6:2–11.
- Sattler, C., Trampnau, S., Schomers, S., Meyer, C., and Matzdorf, B. (2013). Multi-classification of payments for ecosystem services: How do classification characteristics relate to overall PES success? *Ecosystem Services*, 6:31–45.
- Schilizzi, S. (2017). An overview of laboratory research on conservation auctions. Land Use Policy, 63:572–583.

- Schilizzi, S. and Latacz-Lohmann, U. (2007). Assessing the performance of conservation auctions: an experimental study. *Land Economics*, 83(4):497–515.
- Schirpke, U., Marino, D., Marucci, A., and Palmieri, M. (2018). Positive effects of payments for ecosystem services on biodiversity and socio-economic development: Examples from natura 2000 sites in italy. *Ecosystem Services*, 34:96–105.
- Schmidtner, E., Lippert, C., Engler, B., Häring, A. M., Aurbacher, J., and Dabbert, S. (2012). Spatial distribution of organic farming in germany: does neighbourhood matter? *European Review of Agricultural Economics*, 39(4):661–683.
- Sexton, R. J. (2013). Market power, misconceptions, and modern agricultural markets. American journal of agricultural economics, 95(2):209–219.
- Shortle, J. S. and Abler, D. G. (2001). *Environmental policies for agricultural pollution control.* CABI.
- Sills, E., Arriagada, R., Ferraro, P., Pattanayak, S., Carrasco, L., Ortiz, E., Cordero, S., Caldwell, K., and Andam, K. (2008). Private provision of public goods: Evaluating payments for ecosystem services in costa rica. *Working Paper*.
- Simpson, K., Hanley, N., Armsworth, P., de Vries, F., and Dallimer, M. (2021). Incentivising biodiversity net gain with an offset market. *Q Open*, 1(1):qoab004.
- Smith, R. B. (1995). The conservation reserve program as a least-cost land retirement mechanism. American Journal of Agricultural Economics, 77(1):93–105.
- Smith, R. B. and Tomasi, T. D. (1995). Transaction costs and agricultural nonpointsource water pollution control policies. *Journal of Agricultural and Resource Economics*, pages 277–290.
- Stevenson, M. and Weber, C. (2020). Mitigation hierarchies first things first: Avoid, reduce... and only after that compensate. WWF Discussion paper.
- Suding, K. N. (2011). Toward an era of restoration in ecology: successes, failures, and opportunities ahead. Annual review of ecology, evolution, and systematics, 42:465–487.
- Suich, H., Howe, C., and Mace, G. (2015). Ecosystem services and poverty alleviation: a review of the empirical links. *Ecosystem Services*, 12:137–147.
- Tillman, S. C., Spyreas, G., Olnas, A., and Matthews, J. W. (2022). Plant communities in wetland mitigation banks surpass the quality of those in the most

degraded, naturally occurring wetlands, but fall short of high-quality wetlands. Ecological Engineering, 176:106526.

- Tscharntke, T., Grass, I., Wanger, T. C., Westphal, C., and Batáry, P. (2021). Beyond organic farming-harnessing biodiversity-friendly landscapes. *Trends in Ecology & Evolution*, 36(10):919–930.
- Turner, R. E., Redmond, A. M., and Zedler, J. B. (2001). Count it by acre or function—mitigation adds up to net loss of wetlands. *National Wetlands Newsletter*, 23(6):5–6.
- Uthes, S., Matzdorf, B., Müller, K., and Kaechele, H. (2010). Spatial targeting of agri-environmental measures: cost-effectiveness and distributional consequences. *Environmental management*, 46:494–509.
- Vatn, A. (2010). An institutional analysis of payments for environmental services. *Ecological economics*, 69(6):1245–1252.
- Vickrey, W. (1961). Counterspeculation, auctions, and competitive sealed tenders. The Journal of finance, 16(1):8–37.
- Wätzold, F. and Drechsler, M. (2014). Agglomeration payment, agglomeration bonus or homogeneous payment? *Resource and Energy Economics*, 37:85–101.
- Wätzold, F., Drechsler, M., Johst, K., Mewes, M., and Sturm, A. (2016). A novel, spatiotemporally explicit ecological-economic modeling procedure for the design of cost-effective agri-environment schemes to conserve biodiversity. *American Journal of Agricultural Economics*, 98(2):489–512.
- Weibull, A.-C., Östman, Ö., and Granqvist, Å. (2003). Species richness in agroecosystems: the effect of landscape, habitat and farm management. *Biodiversity & Conservation*, 12:1335–1355.
- Weissgerber, M., Roturier, S., Julliard, R., and Guillet, F. (2019). Biodiversity offsetting: Certainty of the net loss but uncertainty of the net gain. *Biological conservation*, 237:200–208.
- Westman, W. E. (1977). How much are nature's services worth? measuring the social benefits of ecosystem functioning is both controversial and illuminating. *Science*, 197(4307):960–964.
- Wilcove, D. S., McLellan, C. H., and Dobson, A. P. (1986). Habitat fragmentation

in the temperate zone in conservation biology, the science of scarcity and diversity, michael soule, ed.

- Williamson, O. E. (1975). Markets and hierarchies: analysis and antitrust implications: a study in the economics of internal organization. University of Illinois at Urbana-Champaign's Academy for Entrepreneurial Leadership Historical Research Reference in Entrepreneurship.
- Wollni, M. and Andersson, C. (2014). Spatial patterns of organic agriculture adoption: Evidence from honduras. *Ecological Economics*, 97:120–128.
- Wu, J. and Babcock, B. A. (1995). Optimal design of a voluntary green payment program under asymmetric information. *Journal of Agricultural and Resource Economics*, pages 316–327.
- Wu, J. and Babcock, B. A. (1996). Contract design for the purchase of environmental goods from agriculture. *American journal of agricultural economics*, 78(4):935–945.
- Wunder, S. (2005). Payments for environmental services: Some nuts and bolts.
- Wünscher, T. and Wunder, S. (2017). Conservation tenders in low-income countries: Opportunities and challenges. *Land use policy*, 63:672–678.